A GRAMMAR OF THE DIALECT

OF

FARNWORTH AND DISTRICT

(Greater Manchester County, formerly Lancashire)

bу

Graham SHORROCKS, B.A., Cert. Ed., M.A.

Submitted for the degree of Ph.D., Department of English Language, University of Sheffield.

September, 1980

Published by University
Microfilms International,
30-32 Mortimer Street,
London, WlN 7RA, reference
order number 81-70,023.

Available in microform and xerographic form.



IMAGING SERVICES NORTH

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

BEST COPY AVAILABLE.

VARIABLE PRINT QUALITY



IMAGING SERVICES NORTH

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

TEXT CUT OFF IN THE ORIGINAL

- (k) [ŋ] is an allophone of /n/ before /k, g/, and not a phoneme of the dialect - cf. RP /ŋ/.¹ The dialect therefore has the medial and final clusters /nk, ng/, etc.
- (1) Dialect /nt/ corresponds to RP /nd/:
 /ont/ 'hand'; /'ει(j)e(r)nt/ 'ironed'; /'θε:z(e)nt/
 'thousand'; /blaint/ 'blind'; /'fri:tnt/ 'frightened';
 /bι'(j)aint/ 'behind'; /'apnt/ 'happened'; etc.

5.5.18.8. Modification:

Word-medial gemination of /n/ tends not to occur. With reference to subsection 7:

Section (a) \longrightarrow /n/ before /s(e)n/

Section (b) \longrightarrow /n + g/ before /t(e)n/

Section (c) retains /tn/, or -> /ten/

Section (d) \longrightarrow /n/ before /t(e)n/

Section (e) -> with /n/

Section (f) retains /in/, or -> /ing/

Section (g) retains /in/, or -> /ing/

Section (h) retains /n/, or -> /ng/

Section (i) \longrightarrow /z/

Section (j) these words are not used in modified speech

Section (1) -> /nd/

The conversion of /n/ to /ng/ on nouns and present participles is not usually carried out in a completely consistent manner. However, as a general rule, the more modified one's speech, the greater the number of occurrences of /ng/ in words

^{1.} Cf. Gimson (1974: 197ff).

having /ŋ/ in RP vis-à-vis dialect /n/. [ŋ] remains basically non-phonemic in modified speech, for it still only occurs before /k, g/, e.g. ['singing] 'singing'. However, that some speakers on some occasions might say ['singing] is not open to dispute, and careful consideration would have to be given in a detailed treatment of modified speech to the question: At what point does [ŋ] become a phoneme?

Final [ŋg] is sometimes devoiced in the speech of some women. This seems to be a mark of considerable modification: ['Oɪŋk'ɪŋg, 'Oɪŋk'ɪŋk'] 'thinking'.

5.5.19.

5.5.19.1.

and finally within the word. In all positions it has a darker or more velar quality than RP /1/ and has a somewhat velar quality even initially before front vowels. The vowel quality or resonance of /1/ varies in accordance with the phonetic context, but is most typically in the area of [o], [o], [y]. This phoneme has a very marked influence on neighbouring vowels and diphthongs, especially when they precede it. The effects of /1/ may be discerned from a reading of the account of the vowel and diphthong phonemes of the dialect. In a general sense, it may be said that the markedly velar post-vocalic [½] is often preceded by a diphthong, 1 and that the quality of [½] can

^{1.} Historically speaking, it might be said that [1] has caused the diphthongisation of original long vowels such as /i:/.

result in a dipthong's having a second element of the same type as that of the vocalic resonance of [1].

Lodge, describing the speech of Stockport - and also that of parts of Manchester, which is quite close to Farnworth and district - even transcribes initial /1/ as [½], e.g. [½½½] 'little'. Although I have not seen fit to transcribe initial /1/ as [½] - I have reserved [½] for final, pre-consonantal, and syllabic occurrences - it must be remembered that realisations of this phoneme tend to have a distinctly velar quality. I stress this point because it is a typical feature of the dialect, and because many people associate a dark /1/ more especially with certain southern dialects. 3

[ö], [y]: /l/ is occasionally articulated without contact being made by the tip and the front of the blade of the tongue, e.g. [miy'k] 'milk'. However, this is not a regular, typical feature of the dialect.4

[w]: /1/ is occasionally realised as [w]: ['qüpöw andz] 'couple of hands'; ['k'3unse' 3uzlz] 'council houses'.

Again, this is not a regular feature of the dialect.

/l/ is rare after /o:/ (sometimes /oə/), and frequently does not occur (i.e. comparatively speaking) after /3 4 ; it does not occur to a lesser extent after / 4 :/, / 6 :/ and in certain final, unstressed positions. These questions

L. Cf. Lodge (1966: 26).

Lodge (1973: 86, footnote 6).
 Note that Hargreaves (1904: 5) gave an /1/for Adlington which he felt to be more reminiscent of German /1/ than English. German /1/ is, of course, light, clear or front in all positions.
 Cf. Gimson (1974: 203, section 3).

of distribution are copiously documented in subsection 7 below.

5.5.19.2. <u>Initial /1/</u>:

[1], somewhat velar, but less so than [1]: [lü2k] 'luck'; [lq:k2] 'like'; [lav] 'live'; [lap'] 'wrap'; etc.

Before a back vowel, [1] is very close in quality to [2].

5.5.19.3. In Initial Clusters:

/l/ occurs in the initial clusters /pl, tl, kl, bl, dl, gl, fl, sl, spl/, and /skl/ if /skle'ro:sis/
'sclerosis' is counted. /lj/ only occurs when
unaccented, e.g. /'miljen/ 'million', and is in
free variation with /lie/. The combinations /tl,
dl/ are rather distinctive in comparison with many
other varieties of English. Examples: /ple:/ 'play';
/tlyez/ 'clothes'; /kle:m/ 'claim'; /bly:/ 'blue';
/dlad/ 'glad'; /glom/ 'glum'; /fly:/ 'influenza';
/sli:p/ 'sleep'; /splsis/ 'splash'; etc.

/tl/ and /kl/ are in free variation, as also are /dl/ and /gl/. $^{\rm l}$

Variants:

- [1], lenis, less velar: in the clusters /bl, dl, gl, spl, skl/.
- []], devoiced, less velar: in the clusters /pl, tl, kl/. The fricative quality of []] is particularly strong in the combination /tl/.
- [1], slightly devoiced, less velar: in the clusters /sl, fl/, and unaccented /pl/ in /ple'ty:n/ 'platoon'.

^{1.} Cf. sections 5.5.3.3, 5.5.5.3, 5.5.4.3. and 5.5.6.3. respectively.

5.5.19.4. Medial /1/:

Medial /l/ occurs intervocallically, e.g. /'moli/'Molly', and may be geminated, e.g. /'lollin/'lolling'.

/l/ occurs in medial clusters. Examples:
/'tʃimbli, 'tʃimli/ 'chimney'; /'simblin/ 'Simnel';
/'fɛtlin, 'fɛklin/ 'repairing'; etc.

/e/ or /o/ may sometimes occur between /l/ and a
preceding consonant: /'gambelin/ 'gambling';
/'fckelin/ 'feckling, fettling = repairing';
/'angelin/ 'angling'; etc.

This does not happen in the case of /tl, dl/ where the consonants are laterally released.

The allophones follow the general distribution pattern given in subsection 3 above. [1] follows accented /p, t, k/; [1] follows fortis fricatives and weakly accented /p, t, k/; [1] occurs intervocalically, and after lenis consonants; and [1] must be added to the list, occurring preconsonantally in medial clusters, e.g. in /'syldom/'Oldham'. Again, it must be remembered that [1] is not a clear, front variant, but that it is merely less velar than [1].

Medially, in final clusters, and word-finally in context, /1/ before $/\theta$ / or $/\delta$ / has a dental tongue-contact. Medially, and word-finally in context, /1/ has a post-alveolar contact when followed by /r/.

5.5.19.5. Final /1/:

[1]. [f2] 'fill'; [wal] 'wool'; etc.

Final /el, ol/ occur to some extent in free variation with final consonant clusters containing syllabic /l/, but

not in the case of the final clusters /tl, dl/ where the plosives are laterally released. The quality of an unstressed vowel before [½] is usually around [ω], e.g. ['p'li'p'ω½] 'people'.

Final [½] in context may be less velarised before a word beginning with /j/ or a vowel, i.e. [1].

Final /1/ may be realised as a vowel, or as [w]. Final /1/ may be geminated before a vowel, hesitation or /1/ in context.

5.5.19.6. In Final Clusters:

/1/ is present in a large number of final clusters: /lp, lt, lk, lb, ld, 2 lts, ld3, lm, ln, lf, lv, l0, ls, lz, ls, lpt, lps, lts, lkt, lks, lbz, ldz, ltst, ld3d, lmd, lmz, lnz, lfs, lf0, lvd, lvz, l0s, lst/. Examples: /alp/ 'help'; /fiəlt/ 'field'; 'sulk'; /balb/ 'bulb'; /fild/ 'filled';2 'mulch'; /bold3/ 'bulge'; /film/ 'film'; /koln/ 'Colne'; /wilf/ 'Wilf'; /solv/ 'solve'; /al0/ 'health'; /pols/ 'pulse'; /milz/ 'mills'; /wels/ 'Welsh'; /tselpt/ 'spoke' (perj.); /tselps/ 'speaks' (perj.); /bclts/ 'belts'; /milkt/ 'milked'; /solks/ 'sulks'; /bolbz/ 'bulbs'; /fieldz/ 'fields'; /bsltst/ 'belched'; /boldsd/ 'bulged'; /filmd/ 'filmed'; /filmz/ 'films'; /milnz/ 'Milne's'; /wilfs/ 'Wilf's'; /twelf0/ 'twelfth'; /Selvd/ 'shelved'; /wolvz/ 'wolves'; /twεlθs/ 'twelfths'; 'wolst/ 'waltzed': etc.

^{1.} Cf. subsection 1 above.

^{2. /}ld/ is of a somewhat restricted distribution due to the use of /lt/: /filt/ 'filled'; /kilt/ 'killed', etc. Cf. further section 5.5.3.6.

[½] is the allophone which occurs in final clusters. When /l/ follows a consonant, syllabic [½] often occurs. If [½] is preceded by /t, d/ they are laterally released. Examples:

['bot2+] 'bottle'; ['ad2+] 'addle = earn'; ['gin+] 'ginnel = passage, entry'; etc.

Note also: ['dobt] 'Daubhill'.

However, since $[\frac{1}{2}]$ is $[\alpha]$ -qualitied, syllabic /1/ is difficult to distinguish from $(\alpha + 1)$.

5.5.19.7. Comparative Distribution:

- (a) The dialect has the initial clusters /tl, dl/.
- (b) [1] is more velar than in RP and the distinction between the allophones [1] and [1] is therefore less clear.
- (c) The dialect has /l/ in the following words: /'leri/
 'hurry'; /lap/ 'wrap'; /'tsimbli, 'tsimli/ 'chimney';
 /'simblin/ 'Simnel'.
- (d) Compared with RP, ME and NS, the dialect does not have /l/ in a large number of words. This is especially the case after /o:/ (to a lesser extent /oə/) and /ɔʊ/:

 (i) Medially. /'woəweks/ 'Wallwork's'; /'wo:ge:t/
 'Wallgate'; /'o:t0e(r)/ 'alter'; /'bɔʊtn/ 'Bolton';

 /'fo:t1/ 'faulty'; /fo:n/ 'fallen'; /stɔʊn/ 'stolen';

 /'ko:1n/ 'calling'; /'o:n1/ 'only'; /'ʃɔʊdðe(r)z/
 'shoulders'; /'ʃɔʊðe(r)t/ 'shouldered'; /'sɔʊdʒe(r)/
 'soldier'; /'ɔ:mo:st/ 'almost'; /ˌɔ:'rɛdı/ 'already';
 /ɔə'ri:t/ 'all right'; /ˌɔ:t'gɛðe(r)/ 'altogether';
 /ˈɔʊdem/ 'Oldham'; /'bɔʊsθe(r)/ 'bolster'; etc.

In the traditional dialect, word-medial /l/ is used optionally after /3Y/ without constituting a modification of speech: /'3Yldom/'Oldham'; /'53Ylðe(r)/'shoulder'; /'p3Ylθrl/'poultry'; /'b3Ylðe(r)/'boulder'; etc.

```
(ii) In final clusters, including inflected forms.

/'Orelfo:z/'Threlfall's'; /payd/'cut'; /'notez/
'Nuttall's'; /bo:z/'balls'; /'appoz/'apples';

/so:t/'salt'; /fo:t/'fault'; /fo:s/'false';

/bo:d/'bald'; /ayd/'old; hold'; /bayd/'bold';

/mayd/'mulled; mould'; /sayd/'sold'; /tayd/
'told'; /kayd/'cold'; /gayd/'gold'; /fayd/
'fold'(v); /fayt/'Fold'; /bayt/'bolt'; /me:t/
'moult'; etc.

Again, /l/ may be used optionally after /ay/:

/mayld/'mould'; /maylt/'moult'; and /me:lt/
'moult'.
```

(iii) Word-finally. /fs:/ 'ugly'; /pv:/ 'pull';
/sty:/ 'stool'; /sky:/ 'school'; /fy:/ 'fool';
/'blakpy:/ 'Blackpool'; /f3y/ 'ugly'; /p3y/ 'cut';
/'tsli.p3y/ 'telegraph-pole'; /o:/ 'all; hall';
/bo:/ 'ball'; /fo:/ 'fall'; /ko:/ 'call'; /sto:/
'stall'; /wo:/ 'wall'; /'appo/ 'apple'; /'sso/
'ash-hole'; /'jare/ 'ear-hole'; /'note/ 'Nuttall';
etc.

5.5.19.8. Modification:

The use of a less velar /1/ is a mark of considerable modification, and is perhaps more common amongst women.

With reference to the preceding section, /l/ is introduced into the words in subsection 7(d), along with appropriate vowel and diphthong changes where necessary, although exceptions may be made for /'o:ni/ 'only', /o:/ 'all', /o:'rait/ 'all right' and /f3Y/ 'ugly' (but not /fc:/); /l/ is lost in the words in subsection 7(c), although /lap/ 'wrap' and /'leri/ 'hurry' might be retained by some speakers; and a trace at least of the combinations /tl, dl/ is likely to be retained in modified speech in word-initial position.

/w/

5.5.20.

5.5.20.1.

/w/ is a labio-velar semi-vowel which occurs initially and medially within the word. /m/ is not a phoneme of the dialect, and the combination /hw/ does not occur, so that contrasts between /w/ and /m, hw/ are not possible. /w/ is pronounced with lip-rounding, but the exact degree of rounding depends upon the following vowel. In consonant clusters, consonants which precede /w/ are somewhat rounded.

5.5.20.2. <u>Initial /w/</u>:

[w]. Examples: /we:(r)ts/ 'ache'; /wom/ 'home';
/wik/ 'wick, lively, alive; week'; /'wikiz/ 'insects';
/wed/ 'wed, marry'; etc.

5.5.20.3. <u>In Initial Clusters</u>:

- (a) [w]: Devoiced after accented /t, k/, as in /twang/
 'accent'; /kwe:(r)/ 'queer'; /kwo:p/ 'Co-op'; etc.
- (b) [w]: Slightly devoiced after /sk, θ, s/ in /skwint/
 'squint'; /'skwozn/ 'squozzen = squeezed' (p.p.);
 /θwæ:(r)t/ 'thwart'; /swop/ 'swap'; etc.
- (c) [w]: In the cluster /dw/: /dwindl/ 'dwindle'.

Unless names of Welsh origin are included, the initial clusters containing /w/ are /tw, kw, dw, 6w, sw, skw/; /gw/ is only secured by names such as Gwen, which are not really a part of the dialect. There is no initial cluster /hw/, as /h/ is not a phoneme of the dialect.

5.5.20.4. Medial /w/:

(a) [w]: Occurs intervocalically: [e'wom] 'at home';
 [e'we:] 'away'. Note the geminated form in
 ['p'3ÿwwin] 'powing = cutting hair (polling)'.
 [w] occurs generally as a glide in vowel and diphthong groups, and within diphthongs. Such occurrences of [w] do not affect meaning, and the strength or prominence of the glide is variable. Examples:
 ['3wwe] 'our'; ['fɔ̃: Win] 'falling'; ['gȳ win] 'going'; ['dȳ: Win, 'dw̄ win] 'doing'; ['mȳ we] 'moor'; etc.

Cf. forms of <u>falling</u>, <u>going</u> and <u>doing</u> without [w] [fÿ:in], [gÿ:in], [dṛ:in], etc.

Such uses of [w] are common in modified speech, e.g. ['sky:wa'] 'school'.

/w/ is realised medially as [w] after a lenis
consonant: /'somwie(r)/ 'somewhere'; /'samwid3/
'sandwich'; etc.

(b) /p, t, k/ medially are not released before /w/ unless the cluster is all a part of the same syllable. In /'alkwol/ 'alcohol' and /'i:kwol/ 'equal', the syllable boundary precedes the /k/, and a somewhat devoiced allophone [w] may be readily discerned.

5.5.20.5. Final /w/:

/w/ does not occur finally, but [w] occurs intervocallically as a glide between words, e.g. [gr: Wet?] 'go at'.

Final [w] may also result from a very velar /l/: ['qupöw andz] 'couple of hands'; ['k'suns*"'suzzz] 'council houses'.

5.5.20.6. In Final Clusters:

None.

5.5.20.7. Comparative Distribution:

- (a) Vis-à-vis RP, ME, and often NS, a considerable number of words in the dialect do not contain /w/ in unstressed syllables. Typical are words which end in -ward(s) and -worth: /'pɛne(r)θ/ 'pennyworth'; /'e:pe(r)θ/ 'halfpennyworth'; /ɔəz, 'ɔəloz, 'ɔəlos/ 'always'; /'fæ:(r)neθ/ 'Farnworth'; /blak onz/ 'black ones', and very many similar examples; /'rɔðelz/ 'Rothwells'; /ˌlɔng'kɔ:zi/ 'Longcauseway'; /'daren/ 'Darwen'; /'darene(r)z/ 'Darweners'; /'fɔre(r)d/ 'forward(s)'; /'bake(r)t/ 'backward'; /'bake(r)ts/ 'backwards'; /'ɔ:ke(r)t/ 'awkward'; /'saɪde(r)ts/ 'sideways'; /tɔe(r)t/ 'towards'; /'ope(r)ts/ 'upwards'; /somot/ 'something', cf. somewhat; 1 /'bɛzɪk/ 'Beswick' (both as a place-name and a personal name).
- (b) /w/ is present in dialect: /kwo:p/ 'Co-op';
 /'alkwol/ 'alcohol'; /wom/ 'home'; /e'wom/ 'at
 home'; /we:(r)ts/ 'ache' cf., however, RP work.
- (c) There is no /m/ phoneme or /hw/ combination in the dialect. Dialect /w/ therefore corresponds to /m/ in all words containing /m/ in some varieties of RP.

5.5.20.8. Modification:

Phonetically, none. With reference to the preceding section:

- (a) /w/ may be introduced into these words.
- (b) /kɔ:p/ is a modified form of /kwɔ:p/; /'alkwɔl/ remains as such, unless /h/ is used; /w/ is

^{1.} Cf. Sieß (1929: 122).

lost in /wom/ and /e'wom/, which become /o:m/ and /et o:m/. /we:(r)ts/ is not used.

(c) /w/ is retained.

5.5.21.

/r/

5.5.21.1.

Note that the symbol [J] was approved in 1973 as an IPA symbol which has the value of a "retroflex frictionless continuant".

/r/ is a post-alveolar frictionless continuant, which is subject to variation in its distribution. Two specifications may be given: one for a speaker of traditional dialect who uses a final and pre-consonantal /r/, and one for a speaker of traditional dialect who does not use such an /r/. The first type of speaker is doubtless more typical of the dialect viewed historically, and the second reflects the current direction of modification. However, speakers of traditional vernacular with little trace of final and pre-consonantal /r/ are nowadays not uncommon. Furthermore, the two specifications represent idealised positions. What happens in reality is that many speakers use final and pre-consonantal /r/ here and there, on one occasion but not on another, i.e. after a rather unsystematic and unpredictable fashion.

^{1.} Journal of the International Phonetic Association, vol. 3, no. 2 (Dec. 1973), p. 61. Note further Wakelin (1972a: 99):
"In Lancashire and the counties bordering it to the south and east non-initial <u>r</u> is of an [J] type, but is perhaps slightly more retroflex than this transcription would suggest."

It is probably fair to say that in the outlying, more countrified districts around the area, final and pre-consonantal /r/ is more extensively used than in the urban areas with which I am concerned in this study. The decline in such uses of /r/ in the urban environment brings with it problems of transcription. For instance, a mere trace of post-vocalic /r/ in a word-final, unstressed syllable - perhaps in "sentence"-final position too, or obscured by noise - can be extremely difficult to detect. Equally, a faint trace of /r/-colouring can be very difficult to distinguish from the roundness and tension associated with /e:/ or /e/, or from schwa-type diphthongisation of a long There are also problems of phonemicisation. The now very uncertain status of final and pre-consonantal /r/ and its total absence in the speech of some speakers render phonemicisations of such words as sure, pork, cord and here as /Y:r, o:r, o:r, i:r/ over-idealised and unrepresentative. The solution adopted here is to phonemicise as $/y \ni (r)$, $o \ni (r)$, $\mathfrak{p}(r)$, $\mathfrak{t}\mathfrak{p}(r)$, which is exact when r is not pronounced, and phonetically quite realistic when it is. 2 By the same token, one may phonemicise /e(r)/ after a diphthong, e.g. /'late(r)/ 'liar'.

With speakers who use final and pre-consonantal /r/, /r/ occurs initially, medially and finally within the word. With those who do not, /r/ occurs initially, medially when

^{1.} The <u>SED</u> and <u>ALE</u> transcriptions for Harwood, on the outskirts of the Greater Bolton area, offer evidence in support of this view. See Orton and Halliday (1962-3) and the <u>Atlas</u> <u>Linguarum Europae</u> Harwood responses.

^{2.} For further details of sections 5.4.5.1, 5.4.6.1, 5.4.7.1. and 5.4.8.1.

intervocalic or post-consonantal, and finally only as a linking \underline{r} or in free variation with /t/, when the next word begins with a vowel.

In stressed syllables /r/ does not occur in the environment /e/ - voicelss consonant. Examples:

> /tsets/ 'church'; /teps/ 'turpentine'; /sket/ 'skirt'; /wek/ 'work'; /we0, wo0/ 'worth'; /nes, nos/ 'nurse'; etc.

5.5.21.2. <u>Initial /r/:</u>

/reits/ 'reach'; /ryad/ 'road'; /roks/ 'rucks'; etc.

The allophone used initially is [x].

5.5.21.3. In Initial Clusters:

/r/ occurs in the initial clusters /pr, kr, br, gr, fr, fr, Or, Or, spr, skr, sOr/. Words containing /tr, dr, str/ in RP have /0r, or, s0r/ respectively in the dialect. Examples: /'pre:te/ 'potato'; /krod/ 'curdle'; /bring/ 'bring'; /greit/ 'great'; /'fri:tn/ 'frighten'; /Srimp/ 'shrimp'; /Ori:/ 'three; tree'; /oronk/ 'drank, drunk'; /spring/ 'Spring'; /skry:/ 'screw'; /s0rong/ 'strong'; etc. Note also: /bron/ 'burn'; /bræ:st/ 'burst';

/brid/ 'bird'.

The initial cluster /\(\sigma r / \) is sometimes /sr/, e.g. /[rimp, srimp/ 'shrimp'.

Allophones:

[4]: Devoiced after /p, k/ when accented.

^{1.} Cf. sections 5.3.6.1. and 5.2.8.1.

- [3]: Slightly devoiced after unaccented /p, k/ (prepare, creation); after /f, \$/, and /s/ when used; and after /sp, sk/.
- [J]: After /b, g/.
- [f] or [f]: Devoiced alveolar tap or roll after $/\theta/$.
- [x] or [x]: Somewhat devoiced tap or roll after /s0/.
- [r] or [r]: Alveolar tap or roll after /o/.

5.5.21.4. Medial /r/:

Speakers who use pre-consonantal /r/ evince a greater range of medial clusters than those who only use medial /r/ intervocalically and post-consonantally. /r/ occurs intervocalically, e.g. /'fored/ 'forward', etc. and thus also in free variation with /t/:1 /'gertn/ 'getting'; /'lertn/ 'letting'; etc. This latter use of /r/ is restricted to a number of present participles after /ɛ/ or /o/. I have noted: getting, setting, letting, cutting and putting. In addition to being used intervocalically by all speakers, /r/ occurs post-consonantally with all speakers, e.g. /'sɛke@rt/ 'secretary'. Again, the dialect has /s@r, @r, or/ corresponding to RP /str, tr, dr/ respectively.

The medial /r/ of speakers who use a pre-consonantal or post-vocalic /r/ is of a more retroflex character. It is often heard after /æ:/: /'pæ:rkin/ 'parkin'; /'kæ:rpit/ 'carpet'; etc. Since it is not used by some speakers, and is used inconsistently by others, I have placed brackets

Usually the voiced flap allophone [t].
 Cf. the parallel use of /r/ word-finally in subsection 5 below. Additionally /r/ may occur in /'bɛre(r)/ 'better', although /'bɛtθe(r)/ is more usual. /'bɛre(r)/ probably occurs more frequently in slightly modified speech.

around it in the general phonemicisation in order to indicate that it is optional. It should be noted that the tongue is often in the retroflex position for the duration of the preceding vowel, resulting in an /r/-coloured vowel, e.g. ['k'ä':jt'in] 'carting'. /r/, of course, occurs after other vowels and diphthongs too.

- [4]: More retroflex allophone. Occurs post-vocalically, and may be used intervocalically too by speakers who use final and pre-consonantal /r/. Other speakers use [4] intervocalically.
- [r] or [r]: Sometimes occur between vowels, e.g. [in0ə'restin] 'interesting'.

5.5.21.5. Final /r/:

[4]. Examples: /far/ 'far'; /foər/ 'four'; /iər/ 'here'; /'moŏer/ 'mother'; etc.

As in the case of medial pre-consonantal /r/, final /r/ is only used by some speakers, and even then not with complete regularity. In the general phonemicisation it is therefore placed in brackets. However, with all speakers, /r/ may alternate with /t/ - principally the voiced flap allophone [t] thereof - word-finally before words beginning with vowels.

Examples: /ger e:t/ 'get out'; /por on/
'put on'; /sor op/ 'shut up'; /nor 'o:fn/
'not often'; etc.

/r/ may be used after $/\epsilon$, $\mathfrak o$, $\mathfrak o$ / in free variation with /t/, but not in all words. The words involved, as in the parallel

^{1. [4]} is merely given as [4] in the presentation of the thesis generally. Its retroflex character may be taken for granted if it occurs post-vocalically.

medial usage described in subsection 4 above, all appear to be very common ones. The allophones used are $[\mathfrak{z}]$, and sometimes $[\mathfrak{z}]$ in the case of those speakers who use a more retroflex allophone.

['slo:0er &:s] 'slaughter house'. This observation also applies to speakers who do not use final /r/ other than before a vowel.

[J] and [J] also function as linking forms when a word ends historically in /r/ and the next word begins with a vowel, e.g. /ster it/ 'stir it'. /r/ also functions as a historically "intrusive" linking form after /æ:, e/ when the next word begins with a vowel: /næ:r a sez/ '"No", I says'; /'granmer en/ 'grandma and'; etc. However, /r/ may not intrude as a linking /r/ after /ɔ:/, as it sometimes does in RP.¹ The dialect uses vowel groups or diphthongs, or a [w]-glide: [k'5'*m, k'o:im, k'o'*m] 'call him'.

Final /r/ may be geminated before a hesitation.

5.5.21.6. In Final Clusters:

[4]. /r/ occurs pre-consonantally in final clusters, and in the general phonemicisation is placed between brackets, as some speakers do not use /r/ in this position, and others only do so inconsistently. The number of such final clusters is very great. In the pattern: final vowel + consonant + consonant, /r/ may be followed by /p, t, k, b, d, 2 g, t, d3,

^{1.} Cf. Gimson (1974: 209).

^{2.} Final /e(r)d/ is limited in distribution due to the use of /e(r)t/-cf. section 5.5.3.6, and the next footnote.

 \int , m, n, l, f, v, θ , s, z/.

Examples: /we:(r)p/ 'warp'; /fie(r)t/ 'afraid'; /wanore:(r)z/ 'Wanderers'; /we:(r)ts/ 'ache'; /je:(r)b/ 'herb'; etc.

with the /r/ pronounced in each case. Note especially /'onŏe(r)t/ 'hundred'. There are many further combinations still, however:

This list of combinations does not lay any claim to exhaustiveness. The attempt to sketch out at least some of the possibilities reminds one of the extent to which detailed phonological work is dependent upon an extensive corpus of dialect lexis <u>in phonemic form</u>. Further, in the case of /r/ there is the particular additional difficulty that it is not always pronounced.

5.5.21.7. Comparative Distribution:

- (a) Final and pre-consonantal /r/ may be used in the dialect. This is more retroflex in form. /r/ is present pre-consonantally in medial clusters, word-finally, and in final clusters, in addition to the positions which it occupies in RP.
- (b) The dialect has the combinations $/\theta r$, δr , $s\theta r/$ in initial and medial clusters. Dialect $/\theta r/$ corresponds to RP $/\theta r/$, and also to all occurrences of RP

^{1. /}m, n, l/ are often followed by /t/ traditionally:
 /we:(r)mt/ 'warmed'; /le:(r)nt/ 'learned'; /ke:(r)lt/
 'curled'; etc. However, /d/ is sometimes used in the
 same words. Cf. further section 5.5.3.6.

/tr/, and therefore has a much wider distribution than RP / θ r/. / δ r/ and / $s\theta$ r/ correspond to all occurrences of RP /dr/ and /str/ respectively. The question of treating /tr, dr/ as unit phonemes consequently does not arise in the dialect. /sr/ sometimes occurs in the dialect for /sr/.

- (c) The presence of a more retroflex allophone, the absence of /tr, dr/, and the presence of /sθr, θr, δr/, give a different range of allophones in the dialect, and a somewhat different allophonic distribution.
- (d) /r/ is used medially and finally in some words in free variation with /t/ in the dialect.
- (e) The dialect does not use linking /r/ after /o:/.
- (f) There is extensive diphthongisation in the dialect where historical /r/ has receded.
- (g) The plural of dialect /tsalt/ 'child' is formed in /r/: /'tslbe(r)/ 'children'.
- (h) /r/ occurs in /'pre:te/ 'potato'.
- (i) Compare the position of dialect /r/ in: /brid/ 'bird';
 /bran/ 'burn'; /krad/ 'curdle'; /bræ:st/ 'burst';
 /'anðe(r)t/ 'hundred'.

5.5.21.8. Modification:

The direction of modification is away from the use of pre-consonantal and final /r/. This affects both the distribution of /r/ and its phonetic realisations. Where /r/ remains after a vowel, the degree of retroflexion tends to be less. /r/ is most resilient after /æ:/.

The clusters /s0r, 0r, or/ give way in modified speech to /str, tr, dr/ respectively. The latter set contains a more fricative allophone of /r/, and the possibility of setting up /tr/ and /dr/ as unit phonemes might be considered.

/r/ is retained in modified speech in words where it is in free variation with /t/.

With reference to subsection 7 above,

- (g) -> /'t[ldren/
- (h) -> /pe'te:te/, etc.
- (i) /brid/ --> /be:(r)d/
 /bron/ --> /be:(r)n/
 /krod/ is retained
 /bræ:st/ --> /best, bost, be:(r)st/
 /'onðe(r)t/ --> /'ondrid/.

People who modify their speech to a considerable extent are very conscious of the use of medial and final /r/, where it corresponds to RP /t/. Children are urged to modify with the exhoration:

/not se mot very jet erinz en 'gorinz/
'Not so much of your "gerrins" and "gorrins"!'

Note that there is actually no form "gorrin", unless one includes 'got in', which is spoken with a different stress and intonation pattern.

5.5.22. /<u>j/</u>

5.5.22.1.

/j/ is an unrounded, palatal semi-vowel which occurs initially and medially within the word, although many occurrences of [j] in word-medial position, or as a glide between words, do not occasion distinctions in meaning.

[j] is a voiced, lenis allophone, but [ç] is a fortis allophone. /j/ occurs in a considerable number of initial

and medial clusters, although the number is less than in RP.

5.5.22.2. <u>Initial /j/</u>:

[j]. Examples: /jɛlz/ 'healds'; /jvə(r)/ 'hair'; /jæ:(r)bz/ 'herbs'; /'jare/ 'ear-hole'; /jɛd/ 'head'; /'jɛdðe(r), 'jɛθe(r)/ 'dragon-fly'; /jaɪ/ 'yes' (when contradicting); /'jɛzɪ/ 'easy'; /'jɛzɪə(r)/ 'easier'; /jæ:(r)/ 'hear'; etc.

5.5.22.3. In Initial Clusters:

- (a) [ç] and [j]: Gimson gives [ç] after accented /p, t, k, h/ before /u:, və/ for RP.¹ In the dialect there is no /h/ phoneme, and the combination /tj/ is unusual, for /tʃ/ is generally used. A fortis fricative may be heard, however, in the combination /kj/, e.g. [k'çÿ't'] 'cute', and also in the cluster /pj/ in a word such as pure, although in the case of /pj/ I have often transcribed [phj].
- (b) [j] occurs after /sp, st, sk, f/:2 /'stjy:we(r)t/
 'Stewart'; /spjy:/ 'spew'; /'skjy:we(r)/ 'skewer';
 /fjy:/ 'few'; etc.
- (c) [j] follows the lenis consonants /b, m, n, v/ and,
 in the case of certain exceptions, /d/. Examples:
 /'bjstn/ 'beaten'; /mjvəl/ 'mule'; /njv:t/ 'newt';
 /vjv:/ 'view'; etc.

/dj/ occurs in /djɛd/ 'dead'; /djɛθ/ 'death' and /djɛf/ 'deaf', although it is less usual than /dɪə/ or /dʒɛ/.

^{1.} Cf. Gimson (1974: 213f).

^{2.} The same allophone occurs after unaccented /p, k/, e.g. in /'dispjy:t/ 'dispute' in section 4(c) of this discussion. Unaccented /pj, kj/ are rare initially, but there is one case of each: curiosity and purification.

- (d) Isolated occurrences of /tj/, more usually /tʃ/:
 /'tjv:de(r)/ 'Tudor'; /tjv:n/ 'tune'.
 Isolated occurrences of /dj/, more usually /dʒ/:
 /djv:n/ 'June'; /'djv:ti/ 'duty'.
- (e) There is an occasional but seemingly very residual instance here and there of palatalisation of initial /k, g/, or of /kj, gj/ clusters in words now generally pronounced with /k, g/: [k^gä:nf?] 'can't'; [g^jɛt] 'get'; ['gjɛt?n] 'got' (p.p.); etc.¹
- (f) The very occasional omission of /j/ in clusters where it would be normal, e.g. /'my:zik/ 'music', could possibly be due to American influence, and I have not recorded any such forms from female informants.

5.5.22.4. Medial /j/:

- (a) In compounds: /ili:t'jɛdɪd/ 'light-headed, fair'; /'skɪnjɛd/ 'skinhead'; etc.
- (b) As a glide in groups of vowels and diphthongs, or within a diphthong. [j] is optional in such contexts, and has no effect on meaning. Examples: ['fɛījə.mən] 'fireman'; ['ɪ̞-jə, 'ɪ̞-jə, 'ɪ̞-jə, 'ɪ̞-jə] 'here'; etc.

 Note also the geminated forms in ['θ̞ɾa'ɪjjɪn] 'trying' and ['ɪ̞-jjə] 'here'.
- (c) In consonant clusters: /'fɛbjv:ɛri/ 'February';
 /'æ:(r)gjv:/ 'argue'; /'dɪspjv:t/ 'dispute'; etc.

 When unstressed, forms such as /'tʃampjen/ 'champion'
 and /'mɪljen/ 'million' alternate with forms
 containing /iə/.

/j/ is not present in the unstressed syllables in /'ambelens/ 'ambulance'; / rege'le: Senz/ 'regulations';

^{1.} Cf. Strang (1974: 82): "Removal of /j/ occurs between a velar stop and a front vowel - can, girl, were regularly /kjæn/, /gj3:1/ till well on in the 19c, and analogous forms occur in New England."

/'dspertiz/ 'deputies'; /'koleri/ 'colliery'; /'sɛle:loid/ 'celluloid'; /'vakom/ 'vacuum'; etc. /tj/ does not occur in either stressed or unstressed syllables generally speaking, but /dj/, which does not generally occur in stressed syllables, alternates with /d3/ in unstressed syllables in words such as gradual, individual, etc.

5.5.22.5. Final /j/:

/j/ does not occur as such, but [j] occurs as a glide between words. This glide may be geminated in the manner of a final consonant. Examples: [wijad] 'we had'; [bijin] 'be in'; [θ rī:jjin] 'three in'; [p'ī:jjönt?] 'pea on the'; ['ī'jjə] 'he er'; etc.

5.5.22.6. In Final Clusters:

None.

5.5.22.7. Comparative Distribution:

- (a) In the words in section 2 of this discussion, dialect initial /j/ corresponds to RP /h/, or no consonant at all.
- (b) The allophone [c] is more restricted in its distribution in the dialect.1
- (c) /j/ is not present in the dialect in a number of unstressed syllables.2
- The dialect does not have the clusters /tj/, /dj/, (d) except as stated above; 3 /lj/, except in some unstressed syllables; 4 /0j/; /sj/; /hj/ (/h/ not

Cf. subsection 3(a) above.
 Cf. subsection 4(c) above, for examples.
 See subsections 3(c), (d) and 4(c) above.
 Cf. section 4(c) above, and /'valjy:/ 'value'.

being a phoneme of the dialect); nor the unstressed combination /zj/. Initial /gj/ is restricted to forms as in (f).

In words listed by Gimson as having two pronunciations, namely /u:/ and /ju:/, 1 (after /1. 0. s, z/), the dialect has only pronunciations with /y:/.²

In words listed by Gimson as having pronunciations with /tj, dj, sj, zj/ alternating with /ts, d3, s, 3/5the dialect uses the latter set.4

- Occasional apparent "omissions" of /j/, as given (e) in subsection 3(f) above.
- (f) The use of [j] or /j/ in words such as [g]st] 'get' contrasts with standard usage.
- (g) Corresponding to RP /d/: /djsd/ 'dead': /djs0/ 'death'; /djɛf/ 'deaf'.
- Corresponding to RP /b/: /bjɛt/ 'beat'; /'bjɛtn/ (h) 'beaten'.

5.5.22.8. Modification:

The pattern of allophones remains basically the same. The phonetic quality of /j/ is not changed, and its use as a glide in vowel groups, diphthongs, and between words is retained in modified speech.

With reference to section 7 of this discussion:

(a) Initial /j/ is lost. /h/ may appear, where appropriate, for those speakers who use it, although it is likely to appear inconsistently, and possibly hypercorrectly.

See Gimson (1974: 214f).

In as far, that is, as the words in question can be elicited from dialect speakers, and - when not - in as far as we can predict their pronunciation in the event of their being used.

3. See Gimson (1974: 215).

4. Again, in as far as the words can be elicited from informants,

or their pronunciations predicted.

^{5.} Cf. section 5.5.23.

- (b) Use of [q] may be slightly increased in the event of /tj/'s being used. | /hj/ is also possible.
- (c) /j/ may be introduced on an optional basis.
- (d) Resistance to these clusters remains in modified speech. Some use of /tj/ and /dj/ is possible.
- (e) Neither more nor less likely than in the residual dialect.
- (f) [j] disappears.
- (g) /dj/ -> /d/.
- (h) $bj/ \rightarrow b/ (+ /i:/)$.

5.5.23. [h]

[h] is not a phoneme of the dialect.² Amongst traditional speakers it occurs very occasionally in word-initial position when extreme emphasis is required.

Examples: [ə 'fq: vəj in ðɛm dë:z | we la' k

ə 'honðət? p'ɛ:nd? tə 'd'e: | ə tÿ: w 'onðət#] 'A fiver

in them (those) days were (was) like a hundred pound(s)

today - or two hundred'. [ī: jæ:s dæ: ə hët əv ə

'blædi gud? Θçe:d] 'He used to do a hell of a bloody

good trade'. In these examples, the words pronounced

with [h] are massively stressed.

The normal equivalent of RP /h/ is therefore zero, but in some cases the dialect has /j/:

/jɛd/ 'head', and compounds thereof: /jæ:(r)bz/ 'herbs'; /jvə(r)/ 'hair'; /jɛlz/ 'healds';

Cf. item (d).
 So also Lodge (1966: 28) for Stockport. On the implications for dialect orthographies, see Shorrocks (1978d).

/jm:(r)/ 'hear'; etc. 1

[h] is sometimes used in modified speech, depending upon the degree of modification. Some speakers use it sporadically, others more extensively, but, even in the latter case, not with absolute consistency. Thus, in addition to the large-scale "omission" of /h/ where it occurs in RP on the part of modified speakers, we also encounter instances of hypercorrect usage, such as a hact of parliment [sic] and a hengine. Hypercorrect usage apart, the use of /h/ or [h] in modified speech is generally associated with the social situations in which a change of style is called for, i.e. in formal situations, at work, and so on. One modified speaker observed: "I always pronounce my aitches at work, but never at home."

Although I have referred to the use of [h] for purposes of emphasis and modification, and as a hypercorrect phenomenon, these do not constitute firm, discrete categorisations. An informant once told me how his "Dad got promoted [||| tə ðī 'họslə [J] |||]" 'to the ostler'. Here, the word ostler is emphasised; the informant is modifying his speech as well (note the full form of the definite article, and the lowered variant of /ɔ/); and the [h] is hypercorrect to boot. All three factors can, of course, be readily understood, due to this being a moment of great personal pride and importance for the informant.

Note: It is worth observing that [h] was largely absent from the speech of those junior schoolchildren whom I recorded.

^{1.} Cf. section 5.5.22.2.

6. MORPHOLOGY AND SYNTAX

We now move on to a discussion of the morphology and syntax of the dialect of Farnworth and district.

6.1 Methodology and Presentation:

An account of morphology and syntax is proposed:

- (a) as a necessary component of a grammar of a dialect; 1
- (b) as desirable within the range of dialect studies, as observed by McIntosh:2
- (c) as a part of the purpose of the study a personal hypothesis is advanced, to the effect that differences between dialects at the grammatical (especially syntactic) level are underestimated:
- (d) as relevant within linguistics as a whole, where analysis of spoken colloquial language is still in its infancy.4 It is suggested that a formal approach is at the moment premature, and that an approach which is corpus-based is more appropriate to the distillation of fundamental syntactic categories, and is a prerequisite to a formal analysis. 5

Some matters relating to grammar have been raised earlier in terms of the problems of interpreting in transcription, the

Cf. subsection 1(a) of section 1.1.1.7, and especially section 1.1.1.8.

Cf. section 1.1.1.8; McIntosh (1961: 104ff). Cf. section 1.1.1.8.

Cf. subsection 2 of section 1.1.1.7.

Cf. section 1.1.1.5; subsections 5, 6 and 9 of section 1.1.1.7 Cf. section 3.9.

grammatical exceptions to dialect $/\theta/$ and $/\delta/$ in correspondence to RP /t/ and /d/, and glottalisation and the realisations of the definite article, the preposition to, and the second person singular interrogative forms of modal and auxiliary verbs.²

A corpus of tape recorded speech is not so adequate for a grammar as it is for a phonology. Many grammatical features will probably not be present in a given corpus and others are likely to be insufficiently clear. Occurrence in a corpus is not enough to justify accounting an item a form or construction The item might perhaps be an error, a of the dialect. reformulation, or an idiosyncrasy. Recurrence is a safer But even then, the fact that a speaker uses a construction several times does not prove that it is part of the grammar of the dialect - or even that it is that speaker's regular choice of construction. It seems preferable, then, that a form should be used at least twice, and by speakers who are:

- (a) not related to each other;
- (b) not being interviewed together;
- (c) not in the position of just having heard the form from the dialectologist, whether deliberately or by chance. These criteria are an attempt to avoid errors and idiosyncrasies, or forms restricted to a single family, or used in imitation of others in the interview situation, and on this reckoning it

^{1.} Cf. sections 5.5.11.4 and 5.5.12.4. 2. Cf. sections 3.9, 5.5.3.5-6, 5.5.4.5-6, 5.5 and 6.2.1.1. 3. Cf. McIntosh (1961: 104).

is necessary to visit two or more unrelated informants when using questionnaire techniques.

In order that a corpus should be of sufficient size to afford a basis for the outline of a grammar, tape recordings may be supplemented by other material. The observations of the dialectologist and the notes which he makes when living in the area are very valuable. However, these are no substitute for analysing a corpus completely, because the dialectologist may simply be collecting forms which he is predisposed to collect, no matter how great the number of chance examples which he assembles.2 This criticism would apply also to hypotheses which he formulates when reading dialect literature, and to the a prioristic nature of existing questionnaires. use of collecting slips sent in by others is in some ways a desirable corrective, in that it at least harnesses the powers The same might of observation and analysis of many others. be said of the use of studies on neighbouring areas. Both sources can yield many hypotheses which prove fruitful when checked out in the area. Ultimately, however, it would be desirable to carry out a thorough grammatical analysis of a very large corpus of recorded speech, attempting to account for all the features in that corpus. But this is probably a task more suited to a team than an individual. It presupposes both the human and material resources required to assemble an adequate corpus. Further, it presupposes the resources to

^{1.} Cf. section 1.2.1.6.
2. Cf. Ruoff (1973: 158). Contrast the procedure of Camproux (1960).
3. Cf. section 1.2.1.4 on the use of written sources.

analyse the corpus when assembled. Since there is no widely accepted grammar of English - especially of the colloquial spoken form - it might well be necessary to formulate an original grammar from scratch. Notions such as text. utterance and sentence would be crucial, whilst the analysis of sentence patterns might well presuppose a suprasegmental There would also be considerable problems of phonology. interpretation for the transcriber(s) and analyst(s), given the constant hesitations and readjustments to syntax which are encountered in the spoken situation.² Clearly, such an analysis is beyond this present study, in terms both of size and the grammatical tools required. The requirement that a corpus be analysed completely is therefore less fully met in the grammatical analysis than in the phonology. In the preparation of the grammar which follows, I used:

- (a) the corpus of tape recordings4
- (b) notes made whilst living in the area 5
- (c) the SED Questionnaire
- (d) collecting slips
- (e) other dialect studies⁸
- (f) dialect literature
- (g) my own questions 10

Cf. section 3.9.

Cf. Wackernagel-Jolles (1971: 90). 2.

^{3.} Cf. section 1.2.

Cf. section 1.2.1.6.

Cf. sections 1.2.1.5-6.

^{5.} 6. Orton and Dieth (1962). Cf. sections 1.2.1.1 and 1.2.1.6. See Appendix and sections 1.2.1.4 and 1.2.1.6. Cf. sections 1.2.1.4 and 1.2.1.6.

^{7.}

^{8.}

Cf. sections 1.2.1.4 and 1.2.1.6. 9•

Cf. section 1.2.1.6. 10.

Forms and constructions of potential interest in (d), (e) and (f) were treated as hypotheses. Special questions were formulated to elicit potential structures. and when living in the area I listened carefully for confirmation of such hypotheses. 2 Questions were also formulated to check out suggestive or problematic items in the corpus more fully.3

It follows from what has been said that the description of morphology and syntax presented here is partial. This must inevitably be the case, unless one accounts for all the features in an exceedingly large corpus. 4 which might well involve the formulation of a grammar from scratch. The partial nature of the grammar leads to the question: which features are described? The answer to this question lies in part in a comparative procedure. Features which are clearly distinctive vis-à-vis SE are described. That it is both valid and interesting to proceed on a comparative basis is stated by Camproux. 5 He recorded the "original", or distinctive syntactic "facts" of a dialect, in comparison with standard French, simply as he found them, and without any preconceived plan - a method which he justified by reference to Descartes and Tesnières. 6 Chaurand, commenting on the work of Remacle (on dialect syntax) with approval, drew attention to the latter's

Cf. section 1.2.1.4.

^{2.} Camproux (1960: 29) suggests the same procedure. Samarin (1967: 61) confirms the need to supplement careful sampling "by other techniques to get more data on linguistic elements which normally are of low frequency".

Even this would be inadequate in the view of those linguists who wish to describe all and only the sentences of a language. or the competence of the native speaker. 5. Camproux (1960: 26). See also section 1.1.1.8. 6. Cf. ibid., 27.

Cf. <u>ibid</u>., 27.

use of a comparative technique in relation to standard French. Similarly, there is a strongly comparative element in Wackernagel-Jolles' treatment of the syntax of spoken German. There are, then, more than adequate precedents for a comparative analysis, and from a practical point of view such a procedure is very expedient. However, it should not be forgotten that such an approach is selective: there are features in a corpus of dialect speech which are not brought out by these means, and there are, of course, features of the standard which are not to be found in the dialect; further, there are forms and constructions which may be common to the dialect and to the standard, or which have a different frequency of occurrence in the two varieties.

Nonetheless, a partial morphology and syntax will serve a useful purpose

- (a) as a part of a description of the dialect of Farnworth and district:
- (b) as a test of a hypothesis that English dialects vary significantly at the morphological and syntactic levels. The general disregard of dialectal syntax is summarised by Wakelin, who writes that "syntax is an unwieldy subject which dialectologists have fought shy of". This means that such generalisations as are encountered concerning dialect syntax cannot be said to be supported by very much evidence. Although little descriptive work has been carried out, two major positions

^{1.} See Chaurand (1972: 246).

^{2.} Cf. Wackernagel-Jolles (1971: 250).

^{3.} Wakelin (1972a: 125).

on syntax appear to have been common to English, French and German dialectologists alike. From the first position, calls are made for the analysis of dialects at the grammatical level, and suggestions concerning the likely difficulties of such analysis are sometimes made. From the second position, it is asserted that grammatical and especially syntactic differences between dialects are not all that great, or even that there are none of any significance.

The first of these views at least suggests that dialect grammar should be investigated, and some recent work shows that investigations of dialect syntax can and should be carried out:

Depuis la magistrale étude de Louis Remacle sur la "Syntaxe du parler wallon de la Gleize" (3 volumes, Liège-Paris, 1952-60), et, dans un autre domaine géographique, l'"Etude syntaxique des parlers gévaudanais" de Charles Camproux (Paris, 1958) il n'est plus possible d'avoir, à l'égard de cette partie de la grammaire, la même, indifférence, ou la même réaction d'effroi.

The eminent Germanist and linguist, Hugo Moser, has encouraged his students to supplement their dialect work with a study of syntax. Furthermore, considering the different emphasis of their work, and the scale of their studies, and allowing for the general disregard of larger syntactic patterns, there seems to be enough evidence in monographs of the school of Wright to suggest that further study of English dialects at the

3. Chaurand (1972: 241). 4. Cf. Ruoff (1973: 34).

^{1.} Cf. for instance McIntosh (1961: 104); the references cited in Ruoff (1973: 33f); and many others. On the need for grammatical analysis of English, see e.g. Strang (1974: 65f). On difficulties of grammatical work, see e.g. McDavid (1971: 128). On the need for a larger corpus than is often available, see e.g. Melchers (1972: 7, 14).

e.g. Melchers (1972: 7, 14).

2. See for instance Ruoff (1973: 35, 62), and the references there; Chaurand (1972: 241). The view is common amongst linguistic geographers: Gregg (1972: 111); Melchers (1972: 14, footnote 1); etc. Cf. also Strang (1974: 235): "The popular assumption that dialect-differences are primarily a matter of accent is erroneous".

grammatical level might well be worthwhile. There is certainly no reason to suppose - as was perhaps done in some instances - that the grammatical accounts in these studies were exhaustive.

The second of the two positions is important, as it may well be one of the reasons why dialect syntax has been neglected. Once such a notion gains acceptance, there is a danger of a vicious circle operating. Chaurand refers to it, with justification, as "cette illusion paralysante" ('this paralysing illusion'). 2 How might such an illusion come Apart from a dearth of descriptive work at the about? relevant linguistic levels, the answer may well lie in the level of generality of grammatical rules. We probably have a wide inbuilt tolerance of many types of grammatical variation, and it is by no means easy for us to formulate the rules which govern our spoken usage. Strang's comments on our perception of linguistic change are based upon this latter point. Noting that we are less perceptive of grammatical change than of phonological or lexical change, she writes:

Grammar deals with observed rules of a middle degree of generality. The rules of grammar are more numerous than those of phonology and less specific than those of lexis; consequently they are difficult to spot and formulate... The level of generality, then, tends to shield grammatical rules from observation; they constitute an area in which our frame of reference is particularly vague and ill-informed, so that departures from precedent are difficult to identify. Perhaps from the same feature derives the particular problem of noting gaps in usage — for example, that older people, or young ones, or

See, for instance, Wright, J. (1892), Reaney (1927), Hargreaves (1904), and Oxley (1940).
 Chaurand (1972: 241).

non-standard or superior speakers, never employ a construction which we ourselves accept as commonplace. \(^1\)

She notes further that what we are taught at school - debatable, prescriptive points usually, and not essential grammatical structures - conditions our view of language, and that there is an academic weakness at the descriptive grammatical level. She adds:

There is also a theoretical weakness. Linguistic theory gives us a frame of reference in terms of which we can compare and notice differences. Even amongst linguistic laymen such a frame of reference, however subliminal, prevails in phonology and lexis. Everyone has had the experience of reacting to speech with such reflections as 'That's not the vowel I use in but or castle' or 'That's not how I use the word disinterested (or 'That's not the right pronunciation to use...'). But except for the Shibboleths on which attention was focussed at school our grammatical frame of reference is very weak, and we are much less well equipped to compare, identify and distinguish, because we really do not know what constitutes 'the same' grammatical use or construction.

One further possible explanation of the view that dialects do not vary significantly at the syntactic level could lie in the influence of dialect literature. Much dialect literature fits dialectal pronunciations, morphological forms and lexical items into written, or more standardised syntax. This tendency could well have been misleading.

The present grammatical account has been described as comparative and partial. Camproux reported that where there were two constructions in the Gévaudanais dialect, one distinctive and one as in standard French, he generally did

^{1.} Strang (1974: 60).

^{2.} Ibid., 62.

not concern himself unduly with the latter, i.e. he kept rather strictly to a comparative method rather than a descriptive one. 1 Apart from the fact that some variation is a question of style, and that what informants say in answer to the learned investigator and his questions is not always what they would say amongst themselves, the purpose of a study must be kept in mind. Where it is the purpose of a study to decide whether there is much difference between dialects at the grammatical level, then it is the different, original or distinctive features which are of interest. is a necessary link between purpose and method. a hypothesis of the present study that English dialects vary significantly at the grammatical level, a comparative approach is in order. At the same time, the extensive use of modified speech in the urban environment cannot be overlooked entirely, and notes on the modification of grammatical features are included where possible.

Like Camproux, I have tried to record dialect grammar as I found it, and to analyse the "facts". However, the data which I collected were no doubt determined by my field-work, purpose and method, and further affected by the transcription and the analytical framework brought to bear upon them.

It seemed inadvisable to adopt a single grammar of English as a comparative base. There are different varieties of SE, and linguists are not agreed as to which type of grammar

^{1.} Cf. Camproux (1960: 29f).

offers the best analysis. No analysis is recognised as being anywhere near exhaustive, especially not in respect of spoken, colloquial SE. My definition of SE is therefore inevitably loose, and resort was sometimes had to the opinions of educated native speakers. Despite difficulties of discrimination and description, it is nonetheless necessary, as Strang writes, "to record some of the now widely current practices which depart from the norms described in such grammars as we have". A considerable amount of such description is in terms of "school grammar", or extensions of it. This grammar is not perfect but, as Ruoff observes, is better than its It has the advantage of being widely understood, and its categories usually mean the same thing to different Where it is not appropriate to the spoken language, people. it can be modified or extended.³ I have used traditional and readily comprehensible terminology wherever possible, but here and there have resorted to whichever terms would best enable me to account for the data.

In spoken language, ellipsis is common. The speaker suits his speech to the situation, and the understanding of the hearer. He corrects his speech, or makes it more precise, as he perceives the effect of it on the listener. Speech tends to be clear enough, given the linguistic context and the pragmatic (extralinguistic) situation. Syntax and grammaticality depend upon the linguistic context and the

^{1.} Strang (1974: 65).

^{2.} Cf. Ruoff (1973: 146).

^{3.} Cf. <u>ibid</u>., 146f.

^{4.} Cf. Wackernagel-Jolles (1971: 90).

pragmatic situation - hence the unsuitability of questionnaires in many instances. "Incomplete sentences", comparative "omissions", optional ellipses, corrections, supplements, hesitations, tag phrases and reformulations are all to be expected. Despite the difficulties of analysis thus caused. I share Ruoff's view that whilst the totality of linguistic usage in its contextual and pragmatic surroundings cannot be grasped, linguistic form may nonetheless be meaningfully examined in abstraction, and that linguistic terminology suffices to this end. 1

I have taken seriously Strang's recommendation that "there is room for far more investigation of total patterns rather than bits of sentences", 2 and have tried to describe such matters as the way that clauses are linked, or the distribution of dependent pronouns. However, the scope of this study will not allow any attempt at a definition of It is repeated that both the analysis and the sentence. corpus are restricted.

In order to substantiate the claim that English dialects vary significantly at the grammatical level, a considerable weight of evidence must be adduced. Therefore in the discussion of the definite article, and other grammatical features which occur frequently in the corpus, considerable importance is attached to the inclusion of sufficient illustrative material.

See Ruoff (1973: 45). Strang (1974: 66).

^{2.}

The arrangement of the grammatical features is as follows. Firstly, some basic constituents of the noun phrase are presented: the definite article, the indefinite article, adjectives, nouns, numerals and pronouns; these are followed by an account of the verbs of the dialect; thirdly, the adverbs, conjunctions and prepositions are discussed.

6.2. The Definite Article:

There are many occurrences of the definite article in the corpus, and all have been considered in the preparation of the account which follows.

6.2.1. Form:

A major distinction in the realisations of the definite article may be made in terms of whether the noun which the article accompanies begins with a consonant or a vowel.

6.2.1.1. <u>Before consonants:</u>

The definite article is realised primarily as glottal stricture, the exact nature of which depends upon the phonetic context. When the word preceding the noun ends in a plosive consonant, the consonant is glottalised, but not audibly released, whilst the initial consonant of the following noun may be released with considerable expiratory energy, e.g.

[tsɔp2 tsëps] 'chop the chips (= firewood)'

[p?] in this example is accompanied by a greater degree of glottal stricture than is the case when a word-final consonant is not audibly released before a word-initial consonant when no article is present. There are occasions in continuous speech when the greater degree of glottal stricture which represents the definite article is not present. This does not usually result in any ambiguity. Generally speaking, however, the degree of glottal stricture is greater when the definite article is present. Word-final /p/, /b/, /k/ and /g/ are simply glottalised, regardless of the quality of the word-initial consonant of the noun, but final /t/ and /d/ may be realised as [t] and [d], or may assimilate to a following bilabial or velar consonant, more particularly to the former. It is therefore often the case that the final glottalised plosive and the initial plosive of the noun are homorganic, and this results in a geminate with strong glottal stricture, e.g.

[i j ip2 bo:] 'he hit the ball'

When the glottal stricture associated with the definite article operates on a final voiced plosive in the preceding word, then there is total or partial devoicing of that plosive; if a voiced consonant follows, then the devoicing is usually only partial. Examples:

[JE:nt? k'amp'] 'round the camp'
[ə'lɨŋg? sa ld] 'along the side'
[ma ind? 'babhi] 'mind the baby'

In contexts where the definite article is transcribed as [Ω] or [Ω], it is important to note that if the following consonant is not homorganic, then alveolar tongue contact is

not always made, and the symbol represents a movement towards the position indicated rather than a definitely alveolar articulation. Only rarely have I been tempted to transcribe a pure glottal stop [2]. Admittedly, the quality of the glottalisation is at times rather indeterminate, and in some cases where the articulation seemed neither bilabial nor velar, nor sufficiently open to warrant [2], I have used [£2]. Hesitation forms described in subsection 3 of this discussion possibly suggest that /t/ is the phoneme which the speaker "has in mind".

After words ending in /m/, /n/, /r/, /1/ or a vowel, the definite article takes the form of a glottalised plosive, which is bilabial after /m/, and alveolar after /n/, /r/, /1/, although the alveolar form may assimilate to a following bilabial or velar consonant, especially the former. After a vowel, the articulation of the glottalised consonant may be of a type $\{\Omega\}$, although $[\Omega]$ is usual before /p/, /m/, and $[\Omega]$ is likely before /k/, /g/. Examples:

['I: vnt? jy: 'na' tid] 'even the United' (Manchester United football team)

[f; ump? m: tz] [it? 't'ak?;]

[dr got? ts:ts]

'from the mills'

'in the tackle'
'they go to the church'

[giz imp? 't'woo p';:s] 'gives him the tother (other) piece'
In the sequence in the mill, with a nasal actually pronounced
in the word in, 2 the assimilation of the definite article to
/p/ results in the assimilation of the preceding /n/ to /m/,

^{1.} Contrast Carney (1969: 9).

^{2.} In is often /1/ before a word beginning with a consonant.

thus:

[*mp? mvt] 'in the mill'

After f, v, θ , δ , s, z, s, t, and d, glottal stricture is often less noticeable, and is sometimes absent altogether, so that if the fricative or affricate is fortis, there may be no perceptible realisation of the definite article, e.g.

[ə'kiɔ:s Jo'ud] 'across [the] road'

If the fricative or affricate is lenis, however, devoicing is readily perceptible under the influence of the glottal stricture associated with the definite article.

Before /s/, the definite article can result in a [ts] affricate, e.g.

[2tsim ez]: sed?] 'the same as he said'
/3/, /w/ and /j/ do not occur word-finally.

It has been observed that the glottal stricture associated with the definite article is typically stronger than that associated with the customary glottal suspension of final stops before words with an initial consonant, but it is not always so, as there would rarely be any resultant ambiguity. There is a joke which was certainly very popular when I was younger and which was used by one child to another. It plays upon the possibility of ambiguity due to the presence or absence of the definite article, given an appropriate context:

FIRST CHILD (aggrieved): [\tau^2 \tau^2 \tau^

^{1.} This joke is produced from memory. It is confirmed by others, but the pronunciation is my own, as the joke does not occur in the corpus.

^{2.} In some Lancashire towns, the <u>wakes</u> refers to the annual local holiday. In Farnworth and district, however, the chief meaning of the word is the annual fair.

In some phrases, the analyst may have little more than intuition to go on in deciding whether or not a definite article is present. This is because glottal stricture operates:

- (a) on a word-final plosive, when the next word begins with a consonant;
 - (b) to indicate the definite article;
 - (c) to indicate the preposition to;
 - (d) to indicate to + the.

Thus, in:

[ä went? bäk? fət? get? bäk? 'nombe fo'ə]
'I went back for to get back (to (the)) number four.'
[bäk? 'nombe] could theoretically be read as:

- (a) 'back number'
- (b) 'back the number'
- (c) 'back to number'
- (d) 'back to the number'.

Context, of course, is helpful, and the informant is in fact saying that he went back in order to get back to the number four shaft at the pit. Yet it would still be impossible to prove the presence of the, unless the informant were questioned, and then gave /tet/([t'et?]) as an alternative, which may also be used for 'to the'. Even then, it must remain questionable whether this can really indicate the presence of the definite article in the original construction, and one is thrown back upon analogous constructions which are semantically more clear, e.g.

[ä wënt? səp?]

which definitely means 'I went to the shop', and not 'I went

to shop' or 'I went shop', and upon the native's intuition that speakers of the dialect use definite articles in such contexts. The fact that they use 'full' forms (and therefore perceptible forms) of the definite article in such constructions when they modify their speech offers a further clue, as SE uses the definite article in a narrower range of contexts than the dialect. 1

6.2.1.2. Before vowels:

When the noun begins with a vowel, the definite article is realised as glottal stricture followed by $[\theta]$, the latter being attached to the noun. The glottal stricture follows the patterns described in the previous subsection, but an alveolar plosive cannot of course assimilate to a bilabial or velar plosive, as $[\theta]$ always follows. Examples:

[s:\frac{1}{2} θειt] 'all the eight'

[bif? θε:v] 'by the half'

[komp? 'θang lin 's [:zn]' come the angling season'

[sii'd? θ3 vd] 'seed (= saw) the old'

(contrast [f2θ] in see the old)

There are no exceptions to the use of $/\theta$ / before a vowel, although see the next subsection on hesitation forms. At the beginning of an utterance, or after a long pause, the element of glottal stricture is sometimes less marked, and sometimes absent altogether. At other times, however, it is clearly present, e.g.

[# f2 Ge' v on i2t'] 'the half on (of) it'

^{1.} Cf. the account of the distribution of the definite article which follows, especially subsection 10.

6.2.1.3. <u>Hesitations</u>:

It is interesting to note that the definite article is released as a /t/ before a hesitation:

[fgu'mp2 t'3:] 'from the er...'

This fact perhaps suggests that the phoneme which the speaker "has in mind" is /t/. If the hesitation takes the form simply of a pause, then the glottal stricture is held across the pause, unless the pause is of very considerable duration.

6.2.1.4. Full forms:

/ðe/ before a consonant, /ði/ before a vowel. A relatively "full" form of the definite article occurs in exclamations, e.g.:

/wot on ck/ 'what the heck!' (= a mild form of what the hell!)²

Full forms of the definite article may also be found in modified speech, but such usage is generally erratic and inconsistent. Glottalised forms often persist in modified speech. A full or modified form is sometimes encountered in broader speech when an informant is giving a careful and exact explanation. For instance, one informant who was carefully spelling out his reasons for working at a particular pit, said:

/it we(r) dgost e 'likl bit myə(r) 'moni | en ðɛn mi 'broðe(r) wekt et ðe se:m pit/

'It were (was) just a little bit more money, and then (= moreover) my brother worked at the same pit.'

^{1.} Suggested in Shorrocks (1978d: 14).

^{2.} Similarly Hargreaves (1904: 79).

6.2.2. <u>Distribution</u>:

Semantically based schemes of classification can be problematic in most fields of study. It is hoped that the following arrangement will, however, at least provide a guide to the distribution of the definite article, despite its probable subjectivity. The classification is comparative vis-à-vis SE, and is therefore only partial.

6.2.2.1. Presence of the definite article:

In the following subsections, the use of the definite article is well substantiated, and in many cases is compulsory.

1. With articles of clothing:

```
/ŏɛn wi jy:s gy: e'bɛ:t wit pit klogz ɔn/
'Then we used to go about with the pit clogs on.'

/wit klogz ɔn/
'with the clogs on'

/ki:p δι'sɛl int tlyəz/
'keep thyself in the clothes'

/ŏɛ dʒos wɛnt dɛ:n it 'takl/
'thou just went down in the tackle'(= clothes)

/ont klogz wer e god θing bi'kɔ:z δι wer æ:(r)d/
'and the clogs were a good thing because they were hard'
(i.e. clogs in general)
```

2. With relatives (and friends):

```
/fer e:r ent t[ailt/
'for her and the child' (= my wife and our child)
/so: mi: en æ:(r) bilt 'broðe(r)/
'so me and our Bill, the brother'
```

```
/ðe waifs 'moðe(r)/
 'the wife's mother' (= my wife's. Full form of article
                       due to mocking a more precise speaker.)
/e 'kopl et me:ts did/
 'a couple of the mates did' (= my mates)
/t ladz/
 'the lads' (= my/our brothers)
/i: siz tet waif/
'he says to the wife' (= his wife)
/'spesli dem ez we(r) 'bringint kidz op/
'especially them as (= those who) were bringing the kids
(children) up' (no particular children are meant)
Similarly:
/t 'nefjy:/ 'the nephew'; /t ni:s/ 'the niece'; /t waif/
'the wife'; /t son/ 'the son'; /t 'gran, d3 ytθe(r)/ 'the
granddaughter'; /t θ3 vd t[ap/ 'the old chap' (father);
/t 83Yd 'fele/ 'the old fellow' (father); etc.
      3.
           With parts of the body:
/i:d e 'liftid oi e'kro:s [t] ryed wit fy:t/
'he'd have lifted thee across (the) road with the foot' (his foot)
/it we(r) 'mæ:(r)veles ε: δε gi:t jy:st ty: it | wit legz 'o:pm/
'it were marvellous how thou geet (got) used to it - with the legs open' (working with your legs open in a low coal seam)
/jy:s bu e'mo:lin ot ni:z/
'used to be a-mauling on the knees' (working on your knees
to your discomfort)
/je jr:s ger e palt et said et 'blemin 'iero:l/
'you used to get a pelt at the side of the blooming earhole'
/smo:l wiel wot or jr:z wit 0ond/
'small wheel what (which) they use with the hand'
```

```
/bit te:1/
'by the tail' (= by its tail)
/a gots di:z le(r) 'makrel | pots e sto:n int 'beli so: di
flo:t 'opraits/
'I guts these here mackerel, puts a stone in the belly so
they float uprights'
/'py::in em &:t wit n&k/
'pulling them out with (by) the neck'
         With the seasons:
                               'in the spring'
/int spring/
/it 'some(r)/
                               'in the summer'
/int '0o:tem/
                               'in the autumn'
/it 'win\theta e(r)/
                               'in the winter'
         With illnesses: 1
      5.
/ðin 'getnt 'diðe(r)z/
'they han getten (have got) the dithers (tremours)'
/av 'getnt 'jedwæ:(r)ts/
'I've getten (got) the headache'
/as 'getnt 'ty:0we:(r)ts/
'Has thou got the toothache?'
/wit momps/
'with the mumps'
/i:z kopt 'me:zlz/
'he's copped (caught) the measles'
etc.
      6. With the concept "night":
/æ:r in spekter ez wer in ot naits/
'our inspector as were (who was) in on the nights'
```

^{1.} Cf. Strang (1974: 137).

```
/au wekt it naits/
'I worked in the nights'
/www gor em o:l in bit nait/
'we got them all in by the night' (by nightfall)
          With the concept "next door":
/ont 'neksde ty: it we(r)t || igri:n'gro:se(r)z/
'and the next door to it were (was) the greengrocer's'
/walt 'smiði wer 'o:nit 'nɛks'de/
'well the smithy were only the next door'
/i we(r) 'gy:in it neks'dyer 'æ:f0e(r)/
'he were going in the next door after'
      8.
          With tools or implements:
/'gerint ko:l wit piks/
'getting the coal with the picks'
/it we(r) laik 'india, robe(r) wen de 'stæ:(r) tid on it wit naif/
'it were like india-rubber when thou started on it with the
knife'
/'nierli o: got bit pik/
'nearly all got by the pick'
      9.
          With occupations:
/i:z it 'spinin/
'he's in the spinning'
/a went it silk/
'I went in the silk'
/it 'weivin/
'in the weaving'
/went de:nt pit/
'went down the pit'
/i:z it mil/
'he's in the mill'
etc.
```

The meaning of these expressions may not be known to the stranger. Whilst, say, a spinning or weaving room at a particular mill is sometimes meant, such phrases often simply refer to the general occupation of a person. Thus, in the last two examples, no particular occasion, pit or mill, is involved; they mean "[he] became a miner" and "he works in a mill" respectively.

With names, customs and institutions: /'i:vnt .jy:'naitid/ 'even the United' (Manchester United) /ont skan/ 'and the Scan' (a supermarket) /'anðe(r)t 'dæ:(r)li o:1/ 'under the Darley Hall' /e'dgent 'krompm 'lodgiz/ 'again (next to, by) the Crompton Lodges' /wot www koelt | 'ke:(r)zlw mo:s || mt 'fæ:ðe kom fromt || 'ke:(r)zli mos/ 'what we called the Kearsley Moss - my father came from the Kearsley Moss' /e'gent 'beri ryad/ 'again (by) the Bury Road' /a went de:nt 'komin 'nombe(r) foə(r)/ 'I went down the Common number four' (a pit) /on went tet | 'njy:t3Yn/ 'and went to the Newtown' (a pit) /je no: we:(r)t 'wɛslı sky: we(r)/ 'you know where the Wesley School were?' (a Sunday School) /et 'botom et 'wo:ge:t/ 'at the bottom of the Wallgate' (in Wigan)

```
/bait 'boske(r) roid/
 'by the Bosco[bel] Road'
/wi jris to liv int | | Osrd at stail fort/
'we used to live in the old High Stile Fold' - similarly,
without the "old":
/we: livd int Oai stail fart/
'we lived in the High Stile Fold'
/je went ont 'fæ:(r)ne0 'mæ:(r)kit/
'you went on the Farnworth market'
/ðats et '03Ytwod/
'that's at the Outwood'
/je no:t 'beriz 'kemis/
'you know the Berry's chemists?'
/'wellet | me:n Oingz we(r)t | 'wo:kin de:/
'Well er ... the main things were the Walking Day'
/tet 'litl 'altn 'moə(r)t[eri/
'to the Little Hulton mortuary'
/ot se:(r)menz 'sondi/
'on the Sermons Sunday'
/int te | 'sivi s0ri:t/
'in the er... Civvy Street'
           With games and entertainments:
/o:ft 'teli vigen ot 'ne:[en:waid/
'off the television on the Nationwide' (television programme)
/st 'ti:'vi:/
'on the TV'
/we(r) 'watsint 'teli te'ni:t/
/We're watching the telly tonight'
/at 'gy:in tep 'bingo: 'lizi/
'Art thou going to the bingo, Lizzie?'
                               'to the football'
/tet 'fy:tbo:/
                               'to the cricket'
/tet 'krikit/
```

```
12. With institutions not preceded by a name:
/or got tsets/ and /or go tet tsets/
'they go to the church' (but no particular church is meant)
/ði men got sky:/
'they mun (must) go to the school' (attend school)
/tet 'me:(r)kit/
'to the market'
/am 'we:tin fe(r)t '01'lek0rik/
'I'm waiting for the electric' (i.e. for the man who reads
the meter, or for any other representative of the institution,
whether delivery man, inspector, or whatever)
Similarly:
/'we:tin fe(r)t pe:s/
'waiting for the gas'
       13. With measurements and numbers:
/'dobl it weit/
'double in the weight'
/it wield o:lt Ocit on je/
'it wheeled all the eight on (of) you'
/'getn o:t 0eit/
'getten (got) all the eight'
/wen p:t Ori: ed/
'when all the three had'2
/t 0e:v on it/
'the half on (of) it'
/it kot mai wek bit 0e:v/
'it cut my work by the half'
/et fas ting ev e 'moe(r)nin/
'at the first thing of a morning'
```

Cf. subsection 10.

^{2.} Note also all the under subsection 15.

```
/welt fost ov o:/
'Well the first of all'

/bit fost in/
'be the first in'

/δι komt fos θing in e 'moe(r)nin/
'they come the first thing in a morning'

/komz op tet 'sekem ple:s/
'comes up to the second place' (of a horse)
```

14. In general statements, where the referent is not specific or 'definite!

The definite article is often used in general statements, especially, it would appear, with well-known referents. In the following examples, it should be stressed that the use of the definite article is not to be accounted for by occurrence of the concept in the preceding text:

/wel je jr:st tek år 'tomi || je jr:st ko: it 'tomi ||
int red 'anke,t\i:f/
'Well you used to take thy tommy (meal) - you used to call
it tommy - in the red handkerchief'

tseind3d e lot | a Oink kidz e 'gerin ont top ne: |

oats Orr: t 'pe:rents | t 'wimin gi:t ont top | 'aftet

wee: (r) wen oi 'ste: (r) tid 'wekkin | ait | kidz e

'gerin ont top ne:/

*[It has] changed a lot. I think [the] kids (children) are getting on the top now - that's through the parents. The women geet (got) on the top after the war, when they started working. Aye (yes), the kids are getting on the top now.'

/'filint tobz wit ko:l/
'filling the tubs with the coal' (The generality of the statement is crucial: particular tubs and particular coal have not been mentioned.)

```
/bat der e'gent | 'man: went 'fe:(r)de(r)/
 'but there again, the money went further' (i.e. money went
further in the past)
A modified speaker explains: "it went in as the raw cotton,
and came out as the finished article - the cloth, you know."
/a 'o:los ri'member im 'selint peiz/
'I always remember him selling the peas' (of a pea-vendor)
/oats roid εit θroit 'moni e'wei/
'that's [the] road (way) how to throw the money away'
/'gi:in emp 'medisen/
'giving them the medicine' (Again, there is no previous
mention of medicine)
/'natselit | 'moniz kom mye(r) te 'pi:pol/
'Naturally, the money's come more to people'
/δε met to bin at pi: ti:z se | fas θing/
'thou might have been on the PTs say, first thing' (in the army)
/'sy:ner at kat fy:d dent dogz meit/
'[He would] sooner have the cat food than the dog's meat'
(could mean 'meat' or 'food'. The informant's dog prefers
cat food to dog food.)
       15. In miscellaneous phrases and constructions:
/it ple:s e/
'in the place of!
/gy: ot spri:/
'go on the spree'
/p:t 'broðe(r)z on om/
'all the brothers on (of) them'
/t waiz ent 'we:(r)foə(r)z/
'the whys and the wherefores'
                         'the sun's at the height'
/t sonz et 0 eit/
                         'down the South' (down South, in the
/darnt sarθ/
                                                 South)
```

6.2.2.2. Formal distribution:

The definite article tends to be repeated in a run of nouns or noun phrases. Examples:

```
/t waiz ent 'we:(r)foe(r)z/
'the whys and the wherefores'
/wi bayt 'te:bl ent t[ie(r)z δie(r)/
'we bought the table and the chairs there'
/wi tlie(r)t sno: ent θais/
'we cleared the snow and the ice'
```

6.2.2.3. Absence of the definite article:

1. As suggested above when discussing the form of the definite article, on occasion no realisation of the definite article can be heard. However, we are probably dealing with a zero realisation of the definite article in certain phonetic contexts in such cases, rather than with an absence of the definite article as a grammatical form. Examples:

```
/e'kro:s ryad/
'across [the] road'
/so 'ɛni.jay || 'mɔa(r)nin 'fɔlewin laik/
'so anyhow, [the] morning following like'
2. In some exclamations:
```

/i: | 'lafin ez we: jr:s dr:/
'Eh, laughing as (that) we used to do!'
/|| 'lafin öɛ jr:s dr:/
'Laughing thou used to do!'
/i: 0ingz | ez öer iz/
'Eh, things, as there is (that there are)!'

/i: lets on no: i:z wiel o:f | e:s i: az ont rest on it/

'He lets thee know he's well off - house he has and all the rest on (of) it!'

The last example shows that we are not just dealing with a lack of glottal stricture after a pause (cf. the first two examples), for there is no $/\theta/$ preceding house.

- 3. With hesitations and reformulations:

 /eret | θ3 vd 'gafe(r) sεz te mi: | 3 vd ιιπ'spεkte(r) sεz te mi:/

 'Er the old gaffer says to me old inspector says to me...'

 In cases such as this, there is no compulsion to omit the article. Indeed, it would be more usual to include it.
- 4. Some dialect uses of the demonstrative and possessive adjectives have the definite article as SE equivalents.

6.2.2.4. <u>Summary</u>:

The comparative absences of the definite article visà-vis SE are few in number, and rather peripheral in character. They are optional, and in some cases unusual. Clearly, they are vastly outweighed by instances - which vary from typical to compulsory - of the definite article in contexts where the standard does not use it. The overall impression, therefore, is that the definite article varies considerably in both form and distribution from its SE equivalent, and that the definite article enjoys a wider distribution in the dialect than in SE.

^{1.} Cf. section 6.4.3. and some of the examples in section 6.4.2.

6.3. The Indefinite Article:

6.3.1. Form:

When the indefinite article is used (see <u>Distribution</u> below), its form is as follows:

6.3.1.1. Before vowels:

/e/ or /en/. There is no rule that would enable one to predict either form in any context. Note that when /e/ is used before a vowel, a glottal stop may well be interposed between the two. Examples:

[ə²'ɛndʒə'n] 'a engine'
[ə²'ë':ɹɹö'əd] 'a air-road'
[laɪk ə 2'ɪt͡ʔtʃɪn ɹe'ət] 'like a hitching rail'
[ə at] 'a hat'
[en ë:s] 'an house'
[en 'ë:ɹgjement] 'an argument'
etc.

6.3.1.2. Before consonants:

	/e/.	Examples:
/e Ging/		'a thing'
/e won/		'a one'
/e siks/		'a six'
etc.		

6.3.2. Distribution:

6.3.2.1. <u>Presence</u>:

The indefinite article has been recorded in the following environments, but it is not compulsory in any of them.

1. Before numbers:

/al 'avnt gor e won/
'I haven't got a one'

/e θri: e foə(r) wik sin/
'a three or four week (weeks) since'

/oat we(r) | e siks æ:(r)z a did/
'that were (was) a six hours I did'

/oi gid 'ezze foə(r) bəb/
'they gave us a four bob (shillings)'

/e 'twɛlvmonθ/
'a twelve month'

etc.

In several cases there is considerable difficulty - which ultimately cannot really be resolved - in deciding whether or not an indefinite article has been used. Consider the following cases before numbers:

(a) ['qp?n ə'bɛ:tə|| t'u: "a:z]
'happen (= perhaps) about X.. two hours'
where X could be:

^{1.} This form could, of course, also be seen as a noun.

- (i) a hesitation
- (ii) an indefinite article
- (iii) a parasitic schwa
 - (iv) due to a reformulation of some sort.
- (b) [oat? wez? e won poetitle instents]
 'That was X one particular instance.'

where X could be:

- (i) a hesitation
- (ii) an indefinite article
- (iii) due to a reformulation, with one being substituted for \underline{a} .

Whilst there are enough examples of the indefinite article before a numeral in the corpus to confirm an optional rule, the problem of certain identification of the indefinite article raises such further issues as intonation patterns, hesitation forms, and the intuition of the native speaker if he is asked to comment or explain.

2. Before times:

/in e 'moə(r)nin/

'in a morning' (during the mornings)

/'sandi e 'mae(r)nin en 'æ:f0einy:n/

'Sunday a morning and afternoon' (on Sunday mornings and afternoons)

/e(r)z bin bad e'bov e 'twelvmone/

'her (she)'s been bad (= ill) above a twelve month'

See also the adverbial phrases of time which involve numbers in the preceding section.

3. In a phrase:

/no: sats e Oing ez/

'no such a thing as'

6.3.2.2. <u>Absence</u>:

In the dialect of Farnworth and district, the indefinite article is very often not used at all in comparison with SE - or at any rate it is not realised. There is no rule to predict any individual case, but relative absence of the indefinite article is a frequent phenomenon. Examples:

```
/ad 'aksident/
'had [an] accident'

/it we(r) 'lovli 'some(r)/
'it were (was) [a] lovely summer'

/it we(r) 'vɛri 'se(r)'praizin θing/
'it were [a] very surprising thing'

/ðe(r)z e:(r)b fer 'ɛvri θin/
'there's [a] herb for everything'

/at ber i: wer 'ɛi(j)e(r)n monge(r)/
'Aye (yes) but he were [an] ironmonger'

and very many others.
```

6.3.3. Modification:

In modified speech, /en/ occurs more frequently before a vowel, there is less use of the indefinite article before numbers, and fewer comparative absences. Otherwise, the form and distribution are essentially the same.

6.4. Adjectives:

```
Adjectives in -en occur in:

/'ayden talmz/

'olden times'

/e 'òronkn 'boge(r)/
'a drunken bugger = a drunkard'
```

Plenty may be used without a following of, in which case it functions in the manner of an adjective, e.g.

/'plenti kom'prest e:r ot fe:s/
'plenty compressed air on the face'

Important from a syntactic point of view is the pattern

NOUN + ADJECTIVE

in:

/it wer e tip en gy: d3ob 'prope(r) je no:/
'it were (was) a tip-and-go job proper, you know'
/// 'moe(r)nin 'folewin/
'[the] morning following'

In both cases the adjective might precede the noun without changing the meaning.

Constructions in which ADJECTIVE + NOUN is perhaps the more usual pattern may also, in some cases, be formed with $NOUN^{1}$ + NOUN:

```
/e wod 'botn/ 'a wood button'
/e wol 'stokin/ 'a wool stocking'
/e 'topms ri'te:(r)n/ 'a twopence return'
- cf. alternatives with wooden, woollen and twopenny.
```

^{1.} Whether the first element is a noun is not perfectly clear, and it might be necessary to draw a distinction between modifiers which have nominal form (wood) rather than clearly adjectival form (wooden).

However, a recent coinage with an adjectival rather than nominal first element is:

```
/e 'kole(r)d :tsl:'vigen/
'a coloured television = a colour television'
```

6.4.1. Comparison of Adjectives:

The comparative and superlative forms of adjectives are formed in -/e(r), ist/respectively, or periphrastically with /mye(r), moe(r)/'more' and /myest, mo:st/'most'. Note, however,

	Positive	Comparative	Superlative
'bad'	/bad/	/was(wes)/	/wast(west)/
'good'	/gad/	$/$ 'bɛtθe $(\mathbf{r})/$	/best/
'much, many'	/mots, mon/fæ:(r)/	//myə(r)(moə(r)) /'fə:(r)ðə(r)/	//myəst(mo:st)/ //'fe:(r)ŏıst/

<u>Little</u> may pattern with <u>less</u> - <u>least</u>, but may equally take the regular endings.

The dialect uses the superlative form of the adjective where SE has the comparative in such constructions as he is the bigger (of the two).

The conjunctions or prepositions used in the comparison of adjectives are:

```
/ne(r)/ nor (most common of the traditional forms)

/til/ till (traditional)

/tin/ tin (perhaps least common of the traditional forms)

/den/ than (common)
```

The conjunction or preposition may be followed by what. Examples:

```
/a @ink its 'jezi@(r) ne: | ŏen wot | wot it we(r)/
'I think it's easier now .. than what .. what it were.'
/'bige(r) tin ŏat/
'bigger tin (than) that'
/its 'faine(r) til slak/
'it's finer till (than) slack'
/i: we(r) 'jonge(r) ner e(r)/
'he were (was) younger nor (than) her'
```

In modified speech, only than is used. The possibility of hypercorrect periphrastic constructions is suggested by one lady's and I was getting more mad and more mad...

The use of double comparatives is a regular feature, e.g.

/koz je we(r) myə(r) 'fite(r)/ 'because you were more fitter'

Such constructions carry over into modified speech, and were at one time acceptable in the standard language.

6.4.2. Possessive Adjectives:

Stressed forms	Unstressed forms (where applicable)
/mai	mı
ðaı	ðı
lZ	
e:(r)	e(r)
it, its	
ε:(r), æ:(r)	· ·
joə(r)	je(r)
ðe:(r)	ðe(r)/

In the third person singular /it/ is the more traditional

^{1.} See Strang (1974: 138). Cf. also Peters (1968: 239).

```
form.
        It is often used of people or animals, e.g.
       /on it fi:t/
                            'on it feet' (of a horse)
However, it is not restricted to animate possessors, for
instance
       /its ad it de:/
       'it's had it day'
may be said of anything which is 'worn out, finished, old
fashioned'.
       The possessive adjective is for the most part distributed
as in SE, although it features predominantly before certain
concepts which are very familiar to the speaker, especially
entertainments, work, and meals.
                                  Examples:
/o: at at a mon got mt 'bingo:/
'oh aye aye I mun (must) go to my bingo.'
/mi 'ge:(r)dne(r)z 'kwsstjen taim/
'my 'Gardners' Question Time'' (radio programme)
/δε ad δι 'dine(r)/
'thou had thy dinner' (a statement of daily routine)
/ði ky:mp ðe(r) te:/
'they came to their tea' (= they came to tea at the
                            speaker's house)
/wi jyist at woik tai(r) wek/
'we used to have to walk to our work'
/ty: 1z wek/
'to his work'
etc.
Sometimes a possessive adjective may be used before some
specific item with which the speaker is familiar, e.g. a
miner said:
```

/bai 'kraimi av 'brokn mi jed o:f mi 'ome(r)/
'By Crimey I've broken my head off my hammer!'

The form <u>our</u> is used before the christian name of a relative:

and is equivalent not only to 'our brother, John' etc. in some kinds of English, but also to 'my brother, John' etc., just as <u>our lads</u> can mean 'my brothers'. Cf. also, perhaps, /giv it os/ 'give it <u>us</u> = me'. These possessive forms may be used in the presence of the person referred to. Thus, when two brothers were talking to me, one would tell me what /æ:(r) d3on ie(r)/ 'our John here' had done in days gone by.

In modified speech, the distribution of the possessive adjective is as in the dialect. Modified forms of the possessive adjective are /ıts, 'sy(w)e(r), jye(r)/ and /me/ as an alternative unstressed form, although /mi/ is usual. /me/ represents a higher degree of modification.

6.4.3. Demonstrative Adjectives:

Singular demonstrative adjective	Plural demonstrative adjective	Noun Adverb
/ðis/	/ði:z/	$(+Noun) (+ /\iota \ni (r)/) + Noun$
/ðat/	/ðem/	(+Noun) (+ /ðiə(r)/) + Noun
,	(/ðy:z/	+Noun)
/jon/	/jon/	+Noun
(/'jɔnðe(r)/	/'jonðe(r)/	+Noun)

You is not used extensively as a demonstrative adjective except when stressed, as explained below. This and these are optionally followed by the adverb here, and that and them 'those' by there. The adverb usually follows the demonstrative directly and therefore precedes the noun (example 1), but in some cases the adverb follows the noun (example 2). /òv:z/ 'those' is infrequent (example 5), and younder relatively so (example 8). Examples:

- 1. /oem oie(r) de:z/
 'them there days = in those days'
- 3. /et öis iə(r) 'bæ:(r)nziz mil/
 'at this here Barnes' mill'
- 4. /Oi:z tə(r) | rodz 'wentsiz/
 'these here.. Rudd's wenches (daughters)'
- 5. /i ör:z de:z/
 'in those days'
- 6. /ði:z lə(r) | 'bə[l]mɛn we(r) 'we:tın e ði:z

 'these here.. haulmen (hauliers) were waiting of (for) these
 lə(r) fe(r)t tɛk ðis lə(r) ko:l/
 here for to take this here coal...'
- 7. /i jon 'ste:bel/
 'in yon (that) stable'
- 8. [modified speech] /et 'jonde(r) 'botem End/
 'at yonder bottom end'

The demonstrative adjective is often followed by an adverb in narratives (examples 4 and 6). The person or thing qualified by the demonstrative will often have been mentioned already.

Examples:

/ois 3rd wolt web/
'this old Walt Webb'

/so: ði:z ty: set o:f te ðis fæ:(r)m/
'so these two set off to this farm'

However, there are other cases in which the noun is being mentioned for the first time, so that the demonstrative is roughly equivalent in meaning to the indefinite article in some varieties of English:

/i:d bin 'raitin to dis 'womon/
'he'd been writing to this (a) woman'
/dis 'ode(r) blo:k/
'this (an)other bloke'

Whilst admittedly often emphatic in questions and commands, the use of that and them sometimes approximates to the use of the definite article in SE, e.g.

/reits az ðat so:t/

'reach us that salt = pass me the salt'

Here, that certainly does not mean 'that as opposed to others'. The main stress is on salt, cf. the possible alternative 'that salt. Similarly, put that cup on that shelf was used in the context of one cup and one shelf, with no reason for emphasis on either. Also, some uses of that + a person's name are not particularly emphatic or perjorative (although other cases are): /ðat 'brai(j)en/ 'that Brian' can probably' be equivalent to SE 'Brian' pure and simple.

^{1.} It is difficult to be absolutely certain, as ultimately one cannot tell whether the speaker had any particular emphasis or connotation in mind.

As in SE, a demonstrative adjective cannot be followed by a possessive adjective. When possession is shown, the construction is of the type:

/ðɛm ðiə(r) soks e ðain/
'them (those) there socks of thine'

Two constructions which occur regularly, followed by a plural verb, are:

/oi:z lot/ 'these lot' (meaning varies with context)

/oi:z soa(r)t e θingz/ 'these sort of things'

In modified speech, /ox:z/ is not used, /oxə(r)/ ->
/oe:(r)/, and those may be introduced in free variation with
them. The general patterns described thus far, however, may
be carried over into modified speech, e.g.

/ðat 'misiz 'sywe(r)d ðe:(r)/
'that Mrs. Howard there' = SE "Mrs. Howard"

Yon is occasionally used as a straightforward demonstrative adjective with the force of that (+ there):

/jon blo:k/ 'yon bloke'

However, with a different stress pattern /'oss mon/ 'this mon' this man' and especially /'jon mon/ 'yon mon are used with the same force and specialised meaning as certain stressed forms of the pronouns yon, him and her. 'Yon mon may refer to a stranger, a visitor, a relative - to any male who is "obviously" intended from context. It refers to what I shall term the known referent. The speaker knows whom he has in mind, and assumes that his listeners understand

^{1.} Cf. sections 6.7.4 and 6.7.1.4. Cf. also the adverbs in section 6.9.4.

the reference. However, only those who are intimate with the speaker can know what he means on some occasions, for what is obvious from context varies from one person to Thus, one informant will use 'yon mon to refer to a brother (who is probably not present, cf. 'this mon), another to a son, another to a mutual acquaintance of himself and the listener, yet another to the landlord of the public house, and so on. Similarly, 'that place, like stressed /'ðiə(r)/ 'there', will refer to a place known to the speaker, and assumed by him to be known to the listener. Occasionally, even the listener who is closely related to the speaker fails to understand, or shows a degree of uncertainty, but generally speaking the device works well. A distinction must therefore be made between this 'mon = 'this man' ('a man', someone whose identity is not crucial or not even known) and 'this mon = the known referent, although one which is usually rather obvious, as 'this mon is always nearer to hand than 'yon mon. In distinguishing between two males, or two known referents, /t'toðe(r) mon/ 'the tother (other) man' may be used.

6.5 Nouns:

6.5.1 Plurals:

Nouns generally add /s/, /z/, / ι z/, depending upon the preceding phoneme, as in SE:

/tsips/ 'chips'
/ladz/ 'lads'
/'pi:siz/ 'pieces'
etc.

and /'ɛ:ziz/ 'houses' shows the same irregularity as SE houses. Dialect /ri:0s/ 'wreaths', however, contrasts with SE /rioz/.

The combination /sts/ is seldom pronounced as such, being rather [s:], or /siz/:

[p'o:s:] or /'po:siz/ 'posts'

Note also /ni:sts/ or /'ni:ziz/ 'nests'.

After a numeral, weights and measures are singular:

/siks int; e det/
'six inch of dirt'

This pattern is common, and has been recorded with the following weights and measures: <u>bob</u> (shilling), <u>foot</u>, <u>gallon</u>, <u>hundredweight</u>, <u>inch</u>, <u>mile</u>, <u>month</u>, <u>pound</u>, <u>quid</u> (pound), <u>shilling</u>, <u>stone</u>, <u>ton</u>, <u>week</u>, <u>yard</u> and <u>year</u>. Further, the same rule applies to <u>pair</u>:

/tr: pe:r e \fr:n/
'two pair of shoes'

and also to load and bundle.

The following plurals involve vowel mutation:

Singular		Plural
/fy:t/	'foot'	/fi:t/
/gy:s/	'goose'	/gi:s/
/lɛ:s/	'louse'	/leis/
/mɛ:s/	'mouse'	/mɛis/
/mon/	'man'	/men/
/'womon/	'woman'	/'wimin/

An irregular plural involving vowel mutation, a consonantal change, and an -er ending is:

/tsailt/ 'child' - /'tsiloe(r)/ 'children'
Plurals in /n/:

Singular		<u>Plural</u>
/i:/	'eye'	/i:n/
/SY:/	'shoe'	/Sr:n/

Oxen is something of a literary or biblical word to informants, as there are none in the area.

A double plural is found in /'galosiz/ 'pair of braces', and the singular form /tsai'ni:/ 'Chinaman' is worthy of note. /fo:k/, /fo:ks/ 'folk, people' are used in free variation. /ɛs/ 'ash' is both singular and plural. Some words such as /'bɔtomz/ 'sediment' and /'siðe(r)z/ 'scissors' are plural only.

A number of compounds (more than in SE) are formed with a NOUN ending in -er + ADVERB, e.g. /'noker op/ 'knocker-up (the man who used to come round with a pole to awaken people)'. The position of the plural morpheme varies in this type of formation: carryings-on is plural only, with the plural morpheme preceding the adverb; slip-ups must have the plural on up; but washer-up 'person who does the washing-

up' and <u>cleaner-up</u> may take the plural morpheme on either element.

In the phonology, reference was made to cases of "missing" final /s/, /z/, which were indeterminate in status, being possibly low-frequency grammatical phenomena, ellipses or errors. The following examples illustrate what could be a low-frequency grammatical phenomenon, namely, plurals without the usual /s/ morpheme:

```
/e lo:d e brik/ 'a load of brick'
/wi ði:z brik/ 'with these brick'
/ði:s det/ 'these dirt'
```

However, interpretation is difficult. In the last instance a singular demonstrative adjective might perhaps have been expected rather than a plural noun. Additionally, one informant insists that the plural of mouse is /e set e me:s/ or /e nest e me:s/ 'a set, nest of mouse', which are clear cases of a singular form where SE would use the plural. On the whole, however, there is insufficient evidence to indicate any clear patterns.

The use of the singular after weights and measures, and the variable position of the plural morpheme in compounds of the type NOUN + ADVERB are both features which occur in modified speech too.

6.5.2. Genitive:

The genitive follows the SE pattern, except that:

^{1.} See sections 5.5.13.6. and 5.5.14.6.

Names ending in /s/, /z/ add /ız/:
 /am 'gy:\int 'wodz\iz/
 'I'm going to Woods''
 /'bæ:(r)nz\iz m\il/
 'Barnes' mill'
 /in 'mo:\z\iz\iz\z de:\z/
 'in Moses' days'
 /'o\int k\int z/
 'Oaks''
 etc.
Similarly: /\its 'o\int e'\int fo\iks\iz/
 'it's other folks''

This ending is also used in modified speech.

- 2. There are cases of a "missing" possessive s morpheme, which might be errors, ellipses or low-frequency zero-genitives. There are precedents for zero-genitives in the history of the English language, and in English dialects. However, interpretation is by no means straightforward, e.g. /e ke: te:1/

 'a cow [or cow's] tail' might just as well be a NOUN + NOUN compound as a zero-genitive. On the whole there is insufficient evidence in the corpus to set up zero-genitives as definite forms of the dialect.
- 3. One or two constructions in the dialect might be

See sections 5.5.13.6 and 5.5.14.6.
 Cf. for instance Strang (1974: 197), Oxley (1940: 61).

rendered with a genitive or double genitive construction in other varieties of English:

/æ:(r) bilz son | iz waif/
'our Bill's son.. his wife...'

[i.e. not our Bill's son's wife or the wife of our Bill's son]

/E:(r) lad ter y:z 'koe(r)tin wit went; ðie(r)|||
'our lad here who's courting with the wench there..

be: liv de:n bat we: 'somwie(r)/
they live down that way somewhere'

(= the wench's family live down that way),

[i.e. not our lad's girlfriend's family, or the family of our lad's girlfriend]

These constructions should perhaps be seen as part of the wider tendency of the informants to state a focus of interest at the beginning of a clause, and then to proceed from there using pronouns and possessive adjectives. 1

^{1.} See the discussion of pronominal syntax, section 6.7.1.6.

6.6 Numerals:

```
CARDINAL
                                             ORDINAL .
         /won, (won)/
 1
                                       /fast/
 2
         /ty:/
                                       /'sɛk(e)n, 'sɛk(e)nt, 'sɛk(e)nd/
 3
         /θri:/
                                       /θe:(r)d/
 4
         /fvə(r)/(older), /foə(r)/
                                       /f v = (r)\theta/(older), /f o = (r)\theta/
                                       / f \iota f \theta / \frac{1}{}
        /faiv/
 5
 6
        /siks/
                                       /siks0/
                                       /'sevn(t)\theta/<sup>2</sup>
        /'sevn/
 7
                                       /ει tθ/
 8
        /ειt/
        /nain/
                                       /nain(t)\theta/
 9
        /tɛn/
                                       /t\epsilon n(t)\theta/
10
                                       /e'lɛvn(t)\theta,'lɛvn(t)\theta,\iota'lɛvn(t)\theta/
        /elevn, 'levn, i'levn/
11
12
        /twelv, (twalv)/
                                       /twelf0, twel0, twelt0; (twalf0, etc)/
13
        /'Oetti:n/
                                       /'0etti:n(t)0/
        /'foe(r)ti:n/
                                       /\text{'fo}(r)\text{ti:n}(t)\theta/
14
        /'fifti:n/
                                       /'fifti:n(t)0/
15
                                      /'siksti:n(t)0/
        /'sikstiin/
16
                                       /'sevnti:n(t)θ/
        /'sevnitiin/
17
        /'eitti:n/
                                      /'ɛɪtti:n(t)\theta/
18
                                      /'nainti:n(t)0/
        /'naintiin/
19
                                      /'twentjee, 'twentee/3
        /'twenti, 'tweni/
20
        / twent: 'won, twen: 'won/ / twent: 'fost, twen: 'fost/
21
        etc.
                                      /'Oettje0, ('Oe:(r)tje0)/
        /'θettι,('θe:(r)tι)/
30
        /'fotti.'foetti.'fo:ti,
                                       /'fattje0,'fa:tje0,'faattje0,
40
            'foə(r)tı/
                                         'fɔə(r)tjeθ/
        /'fifti/
                                      /'fiftje8/
50
60
        /'siksti/
                                      /'sikstjeθ/
        /'sevnti/
                                      /'sεvntjeθ/
70
```

When f/ is not pronounced, forms such as $[fi^2\theta s, fi\theta:s]$ 1. occur.

Bracketed /t/ = optional glottal stricture, i.e. [£?]. /je/ may be /ie/ in this and the following forms.

```
80 /'ειττ/ /'ειττjeθ/
90 /'naιntι/ /'naιntjeθ/
100 /'onŏe(r)t, 'onŏe(r)d/ /'onŏe(r)tθ, 'onŏe(r)dθ/
1,000 /'θε:z(e)nt, 'θε:z(e)nd/ /'θε:z(e)n(d/t)θ/
1,000,000 /'mιlιən/ /'mιlιənθ/
```

Stress: the stress given is that for counting in sequence. Cf. $/\delta e(r)$ we(r) $\theta e t$! ti:n on om/ ! there were thirteen on (of) them!.

Older people, when telling the time, sometimes use forms involving <u>five-and-twenty</u>:

```
/faiv en 'twenti te θri:/
'five-and-twenty to three'
```

When pronouncing dates, o and aught are used, as follows:

```
/'nainti:n o:t 'sevn/ '1907'
/o:t foe(r)/ '[19]04'
/'nainti:n o: tr:/ '1902'
/o: 'sevn/ '[19]07'
```

On the use of "a" before a number, see the <u>Indefinite</u>
Article. 1

The unit of weight, "pounds", may be omitted after a number in the following:

```
/at ad e:f e 'dozn e 'flæ:(r) left/
'I had half a dozen [pounds] of flour left'
/e skoer e pe'te:tez/
'A score of [pounds of] potatoes'
```

The use of score is now quite rare, but /'dozn/ 'dozen' and /e:f 'dozn/ 'half dozen' are still very common indeed.

Numbers in children's games: When children wish to 'bag' a turn in a game, they say:

^{1.} See section 6.3.2.1.

```
/'fest/ 'first', /'sckt/ 'second' and /'0e:(r)dt/ 'third'.
Note also /'lagi/ and /'lasi/ 'last'.
      Fractions:
     /p:v, p:f/
1/2
                             (Traditional)
      /e:v, e:f/
                             (Traditional, very common)
      /æ:v, æ:f/
                             (More modified)
                             (Modified /'e:pni/)
1/2d. /'p:pni/
1.1/2d.
     /, 0ri: '(j) o: pms, , 0ri: '(j) o: pns/
                             (Modified / Ori: (j)e:pms, Ori: (j)e:pns/)
      /'a:pe0/ or /'e:pe0/ 'halfpennyworth'
     /'kwæ:(r)tθe:(r)n/ (Traditional)
1/4
      /'kwo:te:(r)n/
                             (Unusual)
      /'kwæ:(r)t\theta e(r)/
                            (Traditional, common)
      /'kwat\theta e(r)/
      /'kwɛtθe(r)/
                             (Unusual)
      /'kwe:t\thetae(r)/
                             (Unusual)
```

3/4 / θ ri: kwæ: (r) $t\theta e(r)z$ /

6.7. Pronouns:

6.7.1. Personal Pronouns:

6.7.1.1. Subject case:

Singular 1. /ai, æ:, a

- 2. δε:, (δæ:), δε
- 3. i:, (t)
 e:(r), e(r) (short form may be stressed)
 t

Plural 1. we:, wi:, wi

2. jo:, (joə), je

3. de:, di/

The first person singular is often pronounced as a long backed vowel, e.g. [a:]. Although two informants are quite certain that hoo was formerly used for the third person singular feminine, at least by some people, I have never once heard this form used in natural conversation. All my informants, and all other persons observed in the area, used her, or the more modified she in subject case. Duncan classified Lancashire as a hoo-area, and referred to the SED Harwood response her as an "inexplicable exception".2 can only observe that the "exception" is an extremely thoroughgoing one, and that it occurs in an area of some considerable The Harwood response was certainly representative. size.

In an instance quoted by one informant from memory, the person cited by the informant as using hoo (/x:/) was one who had been born in Ringley village, which is close to Kearsley.
 See Duncan (1972: especially p. 189).

Sometimes more modified /(i:, (())/ 'she' and /jy:, (je)/ 'you' are heard mixed in with her, yo and thou forms in speech which is still fairly broad, but it would be fair to say that the former are relatively rare in the broadest speech.

In the second person, all informants distinguished singular from plural in as far as /oɛ:/ could only be singular. Generally speaking, however, /jo:/ could be either singular or plural. Yet one relatively young informant maintained a stricter singular plural-distinction, /os:/ being singular and /jo:/ plural only. However, the singular plural distinction is complicated by other factors. Firstly, modified you-forms have singular and plural reference, and unstressed you and yo are both /je/. Secondly, there is some evidence of a polite use of yo, where the reference may be either singular or plural. Wright distinguished the use of the second singular from that of the second plural and noted that thou was impertinent if used to a superior, 2 but more recently Wakelin reported that "according to SED, the use of this pronoun now implies no discrimination". 3 and Melchers wrote of Wright's distinction "it is difficult to distinguish such a pattern here".4 It must be doubtful, however, whether linguistic geographers have sufficient descriptive evidence on which to base such generalisations. Whilst the earlier distinction is no doubt fading, there are still sons in Farnworth and district who appear to use only the yo form when addressing

Cf. NHG <u>Sie</u>, NF <u>vous</u>; on earlier English cf. Strang (1974:139f) See Wright, J. (1905: section 404) Wakelin (1972a: 164, footnote 6 in respect of p. 113). Melchers (1972: 137).

their fathers.

Thou is traditionally the pronoun which can carry the meaning "one, anyone, you, people" - cf. the SE impersonal pronoun one and colloquial indefinite pronoun you.

With modal and auxiliary verbs, the second singular pronoun often coalesces with, or is incorporated into, the verb form in the interrogative. Examples:

```
/'katnt/ 'can thou not?'
/'atnt/ 'art thou not?'
/'wotnt/ 'would thou not?'
/wilt/ 'will thou?'
/'dos(e)nt/ 'does thou not?'
/dost/ 'does thou?'
```

The terms <u>subject case</u> and <u>object case</u> are not altogether satisfactory when describing pronoun forms in the dialect of Farnworth and district. As will be described below, object case forms occur in a number of instances where the pronoun has subject function.

6.7.1.2. Object case:

```
/oz/ is often used with the meaning 'me', e.g.
```

/lond it oz/ 'lend it us (me)'

The object case is used whenever the pronoun is in object function, and under certain circumstances in subject function too. Emphatic uses of the pronoun are in the object case. For instance, an informant who had had a number of jobs remarked:

/al | av gon re:nd mi:/
'Aye (yes) .. I've gone round, me.'

When there is a multiple subject, pronouns take the object case:

/mi: en me: 'æ:(r)gre:vz sɛz/
'me and May Hargreaves says (said)...'
/mi: en æ:(r) bil gi:t/
'me and our Bill geet (got)...'
/im en mi: we(r) ðie(r)/
'him and me were there'

The object case is used if a numeral follows the pronoun:

/oz Ori: went/
'us three went'

It is also used after the verb be:

/iv at we(r) di:/
'if I were thee'

in commands:

etc.

/ði: kom tə(r)/
'Thee come here!'

in exclamations:

/oi | di:/
'Hey, you!'

in the comparison of adjectives and adverbs:

/en e(r)z | Ori: jer 'syde(r) den di: inne(r)/
'And her's (she's).. three year (years) older than
thee, isn't her (she)?'

in relative constructions:

/im ez kom e'fyə(r) we(r) ðiə(r)/
'him as (he who) come (came) afore (before)
were (was) there.'

and corresponding to SE POSSESSIVE ADJECTIVE + VERBAL NOUN:

/its no: jy:s ði: 'gy:in de:n iz it/'it's no use thee going down, is it?'

All of these rules for the use of the object case are invariable. They also tend to be carried over into modified speech, although hypercorrect uses of the subject form such as 'scuse I, between you and I, etc. can sometimes be heard in speech which is considerably modified.

6.7.1.3. Order of pronoun objects:

When the direct and indirect objects are both pronouns, either one may precede the other:

/i: 'kodnt gr im rt/
'he couldn't give him it'
/a tan rt e(r) bak/
'I tan (took) it her back.'

6.7.1.4. The known referent!

Certain stressed uses of the third person singular pronouns (masculine and feminine) may refer to a known referent. A given informant will regularly use 'her to refer to his sister-in-law, his mother, or some other female relative. The reference will often not be comprehensible from the linguistic context, nor from the pragmatic situation - unless one is intimately acquainted with the informant. When one informant's sister asked him: 'Where's Graham?' [his grandson], he replied:

/i:z gon fe(r)t pik 'e:r op/
'He's gone for to pick 'her up.'

The reference here is to the grandson's wife, who had not previously been mentioned. Apart from being baffling to the outsider, such uses of stressed 'her can also seem rude. A speaker of a variety of NS observed that she felt it was "awful" that a certain dialect speaker should refer to his mother as 'her. Needless to say, no disrespect is intended by those who use stressed pronouns in this way.

6.7.1.5. Impersonal constructions and idioms containing "it":

The pronoun <u>it</u> features in a common impersonal construction /it kom/ 'it come (came)', which means 'it came about that, it happened that':

^{1.} See also the discussion of the demonstrative adjective and the demonstrative pronoun - sections 6.4.3 and 6.7.4. See further the adverbs you and youder, section 6.8.4.

```
/it kom ... ez ői me:d e ryəl/
       'it came ... as (that) they made a rule...'
      /so: it kom laik ozze ||| fres 'mainidge(r) kom/
       'so it came like as (that) er... [a] fresh manager came'
      /wel it kom e 'ryalin laik wit 'jy:nian ozze | čem
       well it came a ruling like with the union as (that) er...
      ez we:nt/
      them as weren't (those who weren't)...'
      /it kam 'liliz te:(r)n/
      'it came Lilly's turn'
      /ne: | wen it komp 'θang(e)lin 'si:zn/
      'now.. when it came the angling season,...'
      etc.
Also, in the sense 'to occur to, to realise':
      /it kep 'komin 'o'ver ez e(r) 'paywint dog/
      'it kept coming over us (occurring to us) her polling
      the dog (cutting its hair off)'
      /it kom ty: mi e(r) sez wi an e 'waier e:(r)d
      ""It came to me," her says (she said), "we have a
                                           wire-haired
      foks'terie(r)/
      fox-terrier"'
      It occurs extensively in idioms, and especially in certain
constructions concerned with travel and the weather:
     /wag it/
                           'wag it = play truant'
      /tsak it/
                           'chuck it = resign, give up work'
      /rof it/
                           'rough it = live roughly, make do
                           with what little is available'
     /mes e'mong it/
                           'mess among it'
      /mok e'mong it/
                           'muck among it'
     both = 'be in a mess, make do or get by in the
     circumstances (which are messy or mucky)
     /(get) kat; it/
                           '(get) catch it = be in trouble,
                           be punished'
```

```
'(get) cop (catch) it = be in trouble, be punished'
/(get) kap it/
/eiv it/
                        'heave it = spare no expense'
                        'stop it (off) = stop, desist'
/stop it (o:f)/
/baik it/
                        'bike it = cycle'
/baz it/
                        'bus it = go by bus'
/flau it/
                       'fly, go by aeroplane'
                        'run'
/ron it/
/wo:k it/
                       'walk'
/eiv it dein/
                       'heave it down = pour with rain'
/its 'peltin it de:n/'it's pelting it down = pouring with
                       rain'
/its 'Oro:in it de:n/'it's throwing it down = pouring with
                       rain'
/its 'ti:min it de:n/'it's teeming it down = pouring with
                       rain'
/snof it/
                       'die; go out (of a flame)'
/av bin 'battin at it o: 'moə(r)nin/
                       'I've been batting at it (working hard)
                       all morning
/dang it/
                       mild form of 'damn (it)!'
/pak it in/
                       'stop, resign'
/it pr:z (wi mi:) 'gr:in E:t et dat taim/
                       'it pulls (with me) going out at that time' (i.e. I do not like it,
                       it is a great effort)
/ko: it kwits/
                       'call it quits = we are even'
/Oreid on it/
                       'tread on it = make a mistake'
etc.
```

Such usages can be heard in modified speech too, although adjustments to pronunciation are obviously made.

6.7.1.6. Dependent pronouns and pronominal syntax:

A number of important sentence patterns are based upon the use of a <u>dependent pronoun</u>. Observers of non-Standard English will be familiar with the pattern in <u>And the manager he said...</u>,

where a subject noun is immediately followed by a dependent pronoun. Melchers quoted Wright's observation that in Scottish and northern dialects "a pronoun is often used to introduce a statement, the specific subject being added later, as it runs very well does that horse", but commented that she "had expected the construction to be more frequent". Although what is "frequent" can be a matter of opinion, I would say that the construction is certainly frequent in the dialect of Farnworth and district.

The pattern:

SUBJECT NOUN + PRONOUN

is common, e.g.

/ns:t 'botem 'level ikl syd 'twenti'fye(r) men/
'now the bottom level it'll hold twenty-four men'
Also common is the pattern:

SUBJECT PRONOUN ... NOUN

- e.g. /ou we(r) lank 'li:tnum ez ou se | uz legz/
 'they were like lightning, as they say .. his legs'
 This pattern often occurs with an auxiliary verb or repeated
 auxiliary following or preceding the noum:
- 2. /bɛt i:d don som 'lɜʏfin ed ɜʏd pæi(r)/
 '[I'll] bet he'd done some laughing had Old Parr'
 Some of these constructions are emphatic, perhaps especially
 those with the inverted AUXILIARY + SUBJECT at the end. Others

^{1.} Wright, J. (1905: section 402); Melchers (1972: 139). Melchers, however, was not undertaking a syntactic analysis, and her generalisation must be treated with caution.

do not seem to be particularly emphatic at all, the constructions being common and occurring on some occasions when no emphasis would seem to be required. Indeed, the informant who produced the first of the two preceding examples immediately reinforced it by adding he did and all "he did that, he did too/indeed!"

In emphatic constructions, the pattern

PRONOUN ... PRONOUN

is found too, e.g.

/well it 'se:ndid pa'ri:t oat did/
'well it sounded all right that did'

The patterns described so far deal with the subject of a sentence. However, there are similar patterns to be observed in the syntax of the object as well:

OBJECT PRONOUN + NOUN

e.g. /a ne:(r) si:d nayt lank it | oat sap 'winde/
'I never seed (saw) nowt (nothing = anything) like it
..that shop window'

The object might take the form of a gerund:

PRONOUN + GERUND

e.g. /ai en'dzoid it 'gr:in e:t/
'I enjoyed it going out'

The pattern:

OBJECT NOUN ... PRONOUN

is found when the noun object is placed before the subject and verb:

/'kofi bi:nz oi jr:s oral om syt'said/
'coffee beans they used to dry them outside'
This last example may be part of a wider tendency on the part
of dialect speakers to state what is of prime concern initially,

i.e. coffee beans, and then to proceed with a clause in which a pronoun substitutes for the noun already mentioned. The same tendency is found in a construction such as the following, where the pronoun occurs after a preposition:

/e pe:r e oroa(r)z a 'didn jr:s get | e wik e:t
'a pair of drawers (underpants) I didn't used to get ...

on om did a/

a week out on (of) them, did I?

It predicts an entire clause in:

/on al tel je e'be:t it wor i: did won taim/
'And I'll tell you about it what he did one time...'

Further material which illustrates the patterns just described is appended here in translation, to save space:

'And the manager he said...'

'Farnworth and Kearsley, they have some funny boundaries'

'And young Donald he could eat...'

'And the captain, Bert Barron, he lives at the top of the road'

'And their.. centre forward he reckoned to score a goal or two'

'Well this Old Walt Webb.. he'd a great big moustache'

'the fruit stalls and the fish stalls they used to start selling everything off cheap' [from a much more modified speaker]

'but McKerracker's their weaving it was... it was coloured weaving there'

'they'd no interest in you the teachers hadn't'

'it ran some years did that gas engine'

'he was all right...old Reagan was'

'he was on the police force some years, that fellow was'

'it was [a] beautiful summer, that was'

'they're both dead now these two lads'

'he died Roland.. then Elsie died the daughter'

'...that went up one time, the Pretoria' (i.e. there was an explosion at a pit called the Pretoria)

'so when we got there.. me and my mate...' [the mate had never been mentioned before. The earlier pronoun therefore cannot be predicted from the preceding context]

'and these <u>dirt</u> they put them in the <u>gob</u>' (mining terminology. <u>Gob</u> refers to where the coal has been dug out, cf. <u>gob</u> 'mouth')

'well eventually they closed it.. the beaming side' [cotton mill]

'I quite like it - me - tartar sauce'

'anyhow this stream there were trout in it'

'Him and his wife, when they used to get falling out, her used to...'

etc.

The use of dependent pronouns is also typical of modified speech.

6.7.2. Possessive Pronouns:

Singular 1. /main

2. dain

3. lz

e:(r)z

Plural 1. $\alpha:(r)z$, $\epsilon:(r)z$

2. joe(r)z

3. de:(r)z/

In modified speech, the second singular is lost, first person plural is /'3Y(W)e(r)z/, and second person plural is /jYe(r)z/ or /'jY:We(r)z/.

6.7.3. Reflexive Pronouns:

Set 11	Set 2 ²
/mi'sɛl	mı
ði'sɛl	ðı
ιs'sεl, ιz'sεl	ım
e(r) sel	e(r)
it'sɛl	it :
æ:(r)'sεl, ε:(r)'sεl	ez, oz
je, sel	j o
ðe(r)'sɛl	em. om/

Set 2 forms are unstressed forms of the personal pronoun. In set 1, -/sal/ may replace -/sɛl/. Only set 1 forms may be used emphatically. That apart, it is difficult to be certain when set 2 forms will and will not occur. Some verbs have a tendency to be followed by set 2 forms, but not to the exclusion of set 1 forms; others have been recorded with set 1 or set 2 pronouns only. Examples:

```
Set 1: /oi did it oe(r)'sel/
                                    'they did it themselves'
        /elp ði'sel/
                                    'help thyself'
        /t 0cgz de(r)'sal/
                                    'the eggs themselves'
        /'polist iz'sel o:f/
                                    'polished himself off' =
                                      'killed himself'
        /bi mi'sel/
                                    'by myself'
        /kild is'sEl/
                                    'killed himself'
        /i: 'kodid is'sEl/
                                    'he <u>codded</u> (deceived) himself'
        etc.
```

^{1.} The first element of the pronouns is genitive throughout the declension - contrast SE <u>himself</u>, <u>themselves</u>. Also there is no singular-plural distinction in the second element. See further Strang (1974: 141, 198).

^{2.} The use of the personal pronoun with reflexive meaning is a feature of some antiquity, cf. Strang (1974: 198).

```
Set 1 or set 2:
      /weis mi/ cf. /weis mi'sel/
                                          'wash myself'
      /Sif(t) di'sel/
                                          'shift thyself!'
      cf. /i: 'wodnt Sift im/
                                          'he wouldn't shift him
                                                        (move)'
      etc.
Often set 2:
      /swil mi (oa(r))/
                                 'swill me (over)'
      /a si:t mi de:n/
                                 'I seet (sat) me down'
      /wen a se: tse:ndz je a mi:n tse:ndz je/
      'when I say change you I mean change you'
      /ores mi/
                                 'dress me'
      /lai ði de:n/
                                 'lie thee down'
      etc.
Set 2 only recorded:
      /ple:z im/
                                 'plays him'(plays, of a child)
                                 'played them' (of men = 'were
      /ple:d em/
                                              off work')
                                 'I <u>mun</u> (must) think me on (remember)'
      /a mon tink mi on/
      /ki:p oi stil/
                                 'keep thee still!'
      etc.
```

It will be clear from these examples that there are verbs which function reflexively in the dialect which do not do so in SE, in addition to verbs which do not occur in SE at all, e.g. cod, above, or /se:p oi'sel/ 'shape thyself!' = 'increase your efforts, improve, try, try harder, do it properly'.

Plural forms of the reflexive pronoun intermediate between dialect set 1 forms and NS forms occur occasionally in the corpus:

```
/3Y(w)e(r)'sɛlz/
/je,'sɛlz/
/ŏe(r)'sɛlz, ŏe(r)'sɛlf/
```

^{1.} Cf. Strang (1974: 153).

In modified speech, the following forms occur:

```
/mi'sɛlf

je'sɛlf

(h)ım'sɛlf

(h)e(r)'sɛlf

it'sɛlf

sy(w)e(r)'sɛlf, -sɛlvz

je'sɛlf, -sɛlvz

ðe(r)'sɛlf, -sɛlvz, (ðem'sɛlvz)/
```

6.7.4. Demonstrative Pronouns:

Singular	<u>Plural</u>	
/ðis	ði:z	(+ /lə(r)/)
ðat	ðem	(+ /ðiə(r)/)
jon	jon /	•

This and these are frequently reinforced with the adverbere. There may follow that and them, but does so less frequently than here follows this and these.

You refers to 'that person' (male or female) as well as to objects. In addition to functioning as a straightforward demonstrative pronoun, you may represent a known referent, with the result that the reference will not always be clear to the outsider. In this latter respect, its function resembles that of the demonstrative adjective, and certain stressed uses of the personal pronoun.

Examples:

/www kept 'wee:(r)nin im e'be:t dis 'i:je(r)/
'we kept warning him about this here' (= the already mentioned defective halter)

Cf. section 6.4.3.
 Cf. section 6.7.1.4. See also the adverbs in section 6.8.4.

```
/ði bel jr:st av ði:z ie(r) | wot ði ko:d flo:ts/
'they all used to have these here. what they called floats'
/wi ad ðis ie(r) te [ift/
'we had this here to shift'
/i: wants ðat ðier ez i:z nor ad/
'he wants that there as (which) he's not had'
/jon ez || 'r:izits 'raitin tr:/
'yon as (that woman who). who-is-it's writing to'
/aks jon e'bɛ:t it/
'ask yon (him) about it'
/wots 'rɛkn e'bɛ:t jon/
'what does thou reckon (think) about yon (that)?'
etc.
```

In modified speech, the use of the adverb with the pronoun is retained, but there is less of it. You disappears, but them is widely retained.

6.7.5. Interrogative Pronouns:

```
'wot/
/wits/
'which'

'who', also equivalent to SE
    'whom', as there is no distinction
    in the dialect between subject
    and object cases of this pronoun.

/Y:z/
    'whose'
```

6.7.6. Relative Pronouns and Relative Clauses:

The chief syntactic patterns are those described in subsection 1, and those illustrated in subsection 2 by examples containing the main relative pronoun as. However, some uses of what and which are distinctive, and cannot be replaced by as.

6.7.6.1. No relative pronoun:

Constructions of a type in which relative pronouns are often found also occur without relative pronouns. One might perhaps speak of optional deletion or omission of the relative pronoun - although as Strang writes, "It is important to realise that contact-clauses are ancient structures of independent origin, not just relatives with pronouns left out" - or of zero-relatives or contact-clauses. Examples:

/a θιnk it we(r) 'wo:kdin it ke:m from/
'I think it were Walkden it came from' [modified speech]

/dats e'be:t '00:ni 'feitin teks ple:s/
'That's about the only fighting takes place'

/welt | bes bi:f oe kad bet | wer/
'Well,.. the best beef thou could buy.. were...'

/ðe(r)z e bon ot 'te:blz main/
'There's a bun on the table's mine.'

/at ad e 'honkl e 'manidger et 'mozgro:vz 'fɛ:nðri/
'I had a huncle (an uncle) a manager at Musgrove's foundry.' [The whole relative construction is omitted here.]

/i: met no: e frend weks in e 'blaksmiθ/
'He might know a friend works in a blacksmith['s].'

/gi mi iz 'dine(r) ðez brayt/

'Give me his dinner thou has brought'

/i: ad e 'nefjr: ad e | big ken'feksene(r)3 sop 'He had a nephew had a big confectioner's shop

et 'e:(r)msen/

at Urmston.'

^{1.} Strang (1974: 142f). Jacobs (1973: 106) comments on the historical omission of the relative pronoun, especially from ME until C18. There is a powerful historical precedent for such constructions.

^{2.} Cf. Strang (1974: 68).

^{3.} Cf. <u>ibid</u>., 68.

/δε no: z δεm 'siksti od 'sevnti/
'thou knows them (those who are) sixty-odd seventy'
etc.

It will be seen from these examples that zero-relatives occur in subject function as well as in object function. Constructions without a pronoun are regular after there is, there are, there were:

/ðe(r) wer 'o:nit 'lo:kalz went in/
'there were only the locals went in'
/ðe(r)z fi:t ki:ps | 'trattin ap en de:nt | 'labi/
'There's feet keeps tratting up and down the lobby'
/ðer e || tr: || blaks bin ap'gre:did/
'There are two blocks been upgraded.'
etc.

As an alternative to a relative clause, two main clauses may be juxtaposed. In the following example, there are no pauses, and the second clause has the same intonation pattern that a relative clause would have:

6.7.6.2. "<u>As</u>":

As is the chief relative pronoun in the dialect, whether referring to people or things. It is equivalent to SE who, whom, that, which. Examples:

^{1.} Cf. Strang (1974: 68).

- (i) /δε no: z 'dʒımı | ιm ez we(r) dıəf/'Thou knows Jimmy, him as were deaf?'
- (ii) /im ez wekt et 'kloraid/
 'Him as worked at the Chloride?'
- (iii) /dem ez wi 034t di we(r)t best/
 'them (those) as we thought they were the best'
- (iv) /t θ3 vd 'le:dι e:r ez ad ιt we(r) 'bιzι 'di:ʃιn em ε:t/
 'the old lady her as had it were busy dishing them out'
 (i.e. the old lady who owned the pea-stall was busy serving peas)
- (v) /bot δεm | ez δι we(r) 'praivet o:nd | δι səloz
 gi:t e 'bεtθe(r) krak et wip δen wot loe(r)d
 'εlzmie(r)z did/

'but them as they were private owned (those which were privately owned) they always geet (got) a better crack of the whip than what Lord Ellesmere's did.'

etc.

The use of a personal pronoun, which duplicates or reinforces the subject or the relative pronoun itself (v), would seem to represent an extension of the use of <u>dependent pronouns</u>, which was discussed under <u>Personal Pronouns</u>. The use of <u>her</u> in (iv), <u>they</u> in (iii) and (v), and <u>they</u> within the relative clause following the relative pronoun in (v), is optional in each case. A further typical construction is the following:

/3Yd 'bili 'somne(r) ðat wer im ez o:nd it i:
'kodnt :onðe(r)'stond/

'old Billy Sumner that were him as owned it he couldn't understand...'

The subject is stated, information or explanation in a relative clause or parenthetical sentence (perhaps containing a relative

^{1.} See section 6.7.1.6.

clause, as in the example here) is interposed, and the original clause is then taken up again with a subject pronoun.

As + POSSESSIVE ADJECTIVE = 'whose'. Two answers to SED question IX, 9, 6, which seeks the relative whose or an equivalent, from my Farnworth informants were:

/Y:z 'onkl snoft it læ:s wik/
'whose uncle snuffed it (died) last week'

/ez iz 'onkl we(r) ŏre:nt læ:s wik/

'as his uncle were drowned last week'

Both constructional types are used. The latter is the more traditional.

I have not noted any cases at all in which a preposition precedes the relative pronoun as in SE. To take just two examples from many:

/ont fæ:(r)m ez wi jv:s go: e'pikin op at/
'on the farm as we used to go a-picking up at'

/it 'Oendzin sys ez i: wo(r) 'waindin in/

'in the engine house as he were winding in'

These constructions might be contrasted with such SE alternatives as to which we used to go to pick up, at which we used to pick up, and in which he was winding. If a speaker of SE were to produce a construction approximately equivalent to the following:

/o:l 'bili.kan | fe(r) 'bry:in op pin/

'...old billy-can, for brewing up in'

and were to retain the form in, then a relative clause would presumably be used.

A subject pronoun takes the object case when followed by a relative pronoun, e.g.

/im ez tard oz 0art dar en o:/
'him as (he who) told us thought that and all (too)'

^{1.} Cf. Orton and Dieth (1962: 100).

The relative clause may be linked to the following clause without an intervening verb in constructions of the type

<u>all as</u> or <u>what</u> + SUBJECT + <u>DO</u> + SUBJECT + VERB Thus:

/bez i: did i: pi:st iz 'rimbandz en por iz 'tsekbandz on en be lai dat 'spine(r)/

'all as he did he pieced his rimbands and put his checkbands on and all like that, the spinner.'

/wel wot diz ty: doz di piks e 'kopel e kod jedz op/ 'well what these two does they picks a couple of cod heads up'

In constructions of the type

/its mi: ez iz 'dr:in it/
'it's me as is (I who am) doing it.'

the verb agrees with the relative pronoun, which is third person - contrast the first person agreement in S.E.

The relative pronoun <u>as</u> is also used in modified speech, but not as often as in the dialect.

6.7.6.3. "What":

What is used somewhat less frequently than as in the dialect, but is very common in modified speech, with the meaning 'who, which, that'. Examples:

/ðɛm wot se ðat/
'them what (those who) say that'

/men wot wekt/
'men what worked'

/tal or won mon to de: wot | gaz E:t et 'sandr wit klagz on/

'[I'll] tell thee one man today what goes out at Sunday with the clogs on'

etc.

Again, a pronoun may be used after the relative:

/en at 'o:fn 'wonder iv de(r)z 'ani ge:(r)lz 'livin
te'de: wot di no: wot 'bande(r)z we(r)/

'and I often wonder if there's any girls living today what they (i.e. who) know what banders were'

From the foregoing examples it will be clear that what is used just as readily to refer to people as to things.

What has an optional pronoun antecedent in such constructions as:

/on al tel je e'be:t it wor i: did won taim/
'And I'll tell you about it what he did one time...'

What can be used in the sense of "whatever" or "that which":

/de wot os larks/ 'do what thou likes' /wot we: want ns:z/ 'what we want now is'

etc.

Cf. also the example containing what in subsection 2 above.

In the comparison of adjectives and adverbs, what may follow tin, till, nor and than, e.g.

/i:z 'sθronge(r) tin wot at am/
'he's stronger tin (than) what I am'

Similarly:1

/dr: it laik wot i: tard oi/ 'do it like what (as) he told thee'

^{1.} These uses of the relative seem to have a tendency to nominalise - as in <u>I do not know what he did</u> - rather than to form a link between a clause and an antecedent noun or pronoun.

6.7.6.4. "That":

That is very occasionally encountered in the dialect, but is quite common in modified speech. As in the cases of as and what, that may apply to both persons and things. Examples from modified speech:

/men ðat woz idi'ze:bld/
'men that was (who were) disabled'
/mi 'moðe(r)z 'siste(r) ðat wekt in ðe mil/
'my mother's sister that worked in the mill'

6.7.6.5. "Who":

Who is occasionally used, generally in more modified speech, but whom is not employed. An example of whose as an alternative to as + POSSESSIVE ADJECTIVE has been given in subsection 2 above.

/oi brant di:z men n: we(r) ded e:t/
'they brought these men who were dead out'

6.7.6.6. "Which":

Which is not often used in the base dialect as a straightforward relative pronoun with a clear object as its antecedent, although on occasions when it does occur, it may mean 'who' as well as 'which', e.g.

/mi dad | wit; we(r) | e 'faije.mon öan/
'my dad, which were (who was) a fireman then...'

It occurs more often in modified speech, e.g.

/ai | wit; woz de 'kleve(r) won ev de ty: ev oz/
'I, which was the clever one of the two of us'

However, there is a most remarkable and extensive use of which in the base dialect, whereby it may refer to an antecedent, often of clausal proportions (i.e. 'which, which thing'), or predict a following notion, which logically might be construed as an "antecedent" nonetheless; yet in some cases, the referent can be so difficult to define that which often appears simply to link clauses - sometimes in a rather loose manner - having a value similar to conjunctions such as and, but, for, since, because, that or as. Examples:

Clausal antecedent:

/mainz iz 'difront te'de: bi'kɔz ðe(r) | 'nasen:laizd wits 'evri:bodi no:z/

'Mines is different today because they're nationalised, which everybody knows.'

Preceding a notion:

/praps faiv pe(r)'sent wit; δι did wekt e:r et | 'fakθriz re:nd e'be:t/

'Perhaps five per cent which they did worked er at the factories round about.'

More obscure referents and conjunctional uses:1

'And when we had to go to tell him one of the looms had gone wrong, he had a stock phrase: "Let it so-and-so-well stop", he always said that, you see. Which it had to stop because it was broken down.'

...he tried to make you look small by having to repeat it, you see. Which, it worked, the weavers would blush and go all confused.

^{1.} Phonetic or phonemic script may perhaps, in the discussion of some syntactic features, be sacrificed to weight and variety of material without appreciable loss.

[Having mentioned that falls in the pit could result in minimum money]: "Which naturally, thou couldn't expect it every week bowt (without) a bit of a fall.'

'And the first pint used to come off the winning landlord of the pub. See what I mean? Which a pint was only 3.1/2d then or 4d - came off the winner.'

'...because the Protestants, which we'd say about 14 schools, they used to walk at the Saturday.'

'And then [they] used to call at that afternoon, which he weren't open at the first thing of a morning when they used to call.'

[In a large pit]: 'And thou would each separate then, which thou could walk the full length, well lit up!'

'...and the womenfolk awhoam (at home), which every woman then used to wear thick, woollen stockings.' etc.

It is very difficult to describe the full range of this phenomenon, although some indication of the extent of its use is perhaps given, when I report that one informant used which 48 times on a single side of tape of less than 30 minutes' duration.

A relative clause may follow a main clause rather than interrupt it:

/en ŏe(r) kom te 'dʒi:zos 'kɔle(r)z wer ɔ:f wit\ wi kɔ:d om/ 'and their come-to-Jesus collars were off, which we called them.'

Contrast, however, the example in subsection 5 above.

6.7.6.7. Derived forms:

So in the above forms is always unstressed. Although other formations of the types -ever and so -ever are conceivable, the ones given above are the customary forms. The type so + RELATIVE PRONOUN is preferred - cf. also /se wie(r)/ so where 'no matter where, wherever', /se wen/ so when, /se e:/ so how. They are used as general relatives to introduce a concessionary clause. /se wot/ so what therefore has the meaning 'no matter what, whatever'. Contrast the S.E. formations whatsoever, etc. 2

```
6.7.7. Indefinite Pronouns and Adjective-Pronouns:
       /som/ 'some'; /'som(b)di, 'som:bodi/ 'somebody';
       /'somot/ 'something'.
       /'ani, 'oni, ('Eni)/ 'any'; /'æ:(r)mdi, 'ani,bodi,
       ('eni,bodi)/ 'anybody'; /3Yt/ 'anything'.
       /no:n, non/ 'none'; /'no:bdi, 'no:bodi/ 'nobody';
       /nayt, 'nothing'.
       /o:/ 'all'; /('avri,θin), 'εvri,θin/ 'everything';
       /'svri,bodi/ 'everybody'; /'svri/ 'every'; /i:ts/
       'each'.
       /'plenti, ('pleni)/ 'plenty': used adjectivally3
       and pronominally, e.g. / plent: jy:s go in fe:(r)
       'Sot.laitin/ 'plenty used to go in for.. shot-lighting'
       /'moni, ('meni)/ 'many'; /mots, mits/ 'much';
       /e lot/ 'a lot': /'sevrol/ 'several'.
       /e'ny:f/ 'enough'.
       /fjr:/ 'few'; /e fjr:/ 'a few'.
       /bya0. bo:0/ 'both'.
```

^{1.} Contrast stressed so = 'so, therefore', which could be followed by the same words in completely different constructions.
2. Cf. Strang (1977: 144).

^{3.} See section 6.4.

/'e:ðe(r), ('aiðe(r), 'i:ðe(r))/ 'either'; /'no:ðe(r), 'ne:ðe(r), ('naiðe(r), 'ni:ðe(r))/ 'neither'.

/sit[, sot[/ 'such'.

See the expanded forms of the relative pronoun for whatever. 1

None requires further comment, in that its distribution restricts the distribution of <u>any</u> compared with S.E. The dialect equivalent of

S.E. I haven't any
is /av 'getn no:n/
 'I've getten (got) noan (none)'

Similarly, <u>none</u> may be used after a negative in the dialect.

The same applies to <u>nothing</u> and <u>nobody</u>, e.g.

/ðat mon 'wodnt pæ:(r)t wi nayt/

'that man [known referent] wouldn't part with nothing'

Anything and anybody are therefore similarly restricted in distribution.

In modified speech, the same basic distribution of none, nobody and nothing applies. /37t, n37t, 'somot/ may be retained, and the same forms and uses of one and plenty may be found.

^{1.} See section 6.7.6.7.

6.8. Verbs:

6.8.1. Present Tense:

Endings in -s, except on second and third person singular and third person plural forms described in subsection 1, are governed by conditions set out in subsections 2 and 3. They are not general to all persons in all uses of the present tense. Contrast Melchers' findings in respect of Yorkshire dialects:

"As stated in EDG section 435 about the North Midland dialects all persons tend to take [s], [z] or [əz] in the present tense, but as far as I can see the use is not restricted to certain conditions. In general there is great confusion in this respect."

6.8.1.1. Regular present tense endings:

Anomalous finites are treated separately below. The following paradigm is representative other than under conditions stated in subsections 2 and 3.

/Oink/ 'think'

 /aι θιηκ
 we: θιηκ, ('θιηκη)

 δε: θιηκ
 jo: θιηκ, ('θιηκη)

 i: θιηκ
 δe: θιηκ, 2 ('θιηκη)/

With the verb go, an ending may follow $/\alpha/$ but not $/\gamma:/$:

/gr:, go/ 'go'

/al gy:, gα we: gy:, gα, (gαn)
δε: gαz jo: gy:, gα, (gαn)
i: gαz δε: gy:, gα,² (gαn)/

^{1.} Melchers (1972: 138). Cf. also Hargreaves (1904: 93). The restrictions on -s endings described in the present study differ to some extent from those described by Wright in EDG section 435.

^{2.} Forms without -s endings apply after the pronoun they. As will be explained below, -s endings are likely to be used after noun subjects.

Since /o/-forms of go may be stressed, the verb is not always anomalous in the dialect. However, /v:/-forms are certainly very common. The verb do is conjugated in the same way as the verb go, although -en plural endings are more common in the former case, and /do/ forms, if stressed, are restricted in distribution. The verb do must be treated as an anomalous finite in respect of both its positive and negative forms. 1 Other verbs with infinitives ending in /v:/ (e.g. /pv:/ 'pull') retain the same vowel throughout the conjugation.

Verbs with a stem ending in a vowel or a lenis consonant other than /z, ds/ take the ending /z/ in the second and third person singular; after /s, z, /s, /s, and /s/ the ending is /s/, e.g. /s 'aks/s' 'her (she) asks'; after fortis consonants other than those just given, the ending is /s/. The sequence /s/ is rarely pronounced in the dialect, and forms such as /s/ are pronounced /s/'kɔs/s/ or /s/| e.g.

/it 'kosiz mi: e 'blodi mint/
'it costs me a bloody mint'

Cf. the sequence /sks/ in the somewhat modified

/Si 'asiz fo dis mon/
'she asks for this man'

The third person plural often takes the same ending as the third singular. Indeed, in many of the following examples an -s ending is altogether typical. However, it does not occur in ordinary present tenses (cf. subsections 2 and 3) after the pronoun they itself - hence the forms in the above paradigms.

^{1.} See section 6.8.4.11.

```
/i: no:z wot θingz bi'longz/
 'he knows what things belongs (he is well informed)'
/ðe(r)z 'bokits ez ronz on tse:nz laik/
 'there's buckets as (which) runs on chains, like'
/ðe(r)z faiv 'onðe(r)t komz/
 'there's five hundred comes'
/δεm men goz/
'them (those) men goes'
/koz dats roid ez eingz iz/
'because that's the road (way) as (that) things is'
/mainz iz/
'mines is'
/wen demz dan/
'when them's done'
/so: wist at si: wot 'tse:ndziz komz inte 'bi:in δεn/
'so we shall have to see what changes comes into being then'
/õεm Y:z kopt/
'them who's copped (those who are caught)'
/δεmz wot δι 'karıd/
'them's what they carried'
```

Plural endings in -en may be used after we, you (plural) and they. The form of the ending is [n] after a vowel, and /en, on/ or a syllabic /n/ after a consonant, e.g. [we: 'want?n] 'we want'. The use of these endings must now be accounted residual, at least in the urban area. Only one very elderly informant, raised by grandparents, used these endings widely in normal conversation:

```
/we: 'wantn/ 'we want'
/ou 'laikn/ 'they like'
etc.
```

Some other informants used <u>-en</u> plural endings only on one or two very common verbs, although their use with <u>have</u> is still widespread. Examples:

/e: 'moni don je want/
'how many do you want?'

/oin mayt de wi: it/

'they have <u>nowt</u> (nothing) to do with it'

Wakelin refers to the use of -en plural endings on verbs.

Lancashire appears not to feature in Wakelin's concept of the kernel area for what he terms an obviously "recessive feature", but the feature is certainly still present in the dialect of Farnworth and district, and was probably much more widespread at one time. Admittedly, one informant of local parentage, who spoke residual dialect himself, said he had never heard such forms; however, another informant recalled quite clearly that they were once more common, citing examples and giving particular persons as sources. Local dialect literature and works concerned with local history and traditions suggest that -en plural endings were once common, e.g.

Neaw yo' known God loves annybody...3

'Now you [plural] know God loves anybody...'

Similarly, Bamford, writing in the middle of the last century, gave paradigms of modal and auxiliary verbs which show -en endings throughout the plural of the present and preterite for each verb. Wright also observed that the present plural of

^{1.} Writing of a Cheshire dialect, Boyd (1954: 17) observed that some elderly people used -en plural endings invariably, but that they had been lost by most of the then present generation, except with a few very common verbs.

^{2.} Wakelin (1972a: 120).

^{3.} Gaskell (1964: 57). 4. Bamford (1850: 265f).

verbs ends in $-\underline{en}$ in South-East and South Lancashire, that the plural of <u>have</u> has \underline{n} in nearly all Midland counties, and that the preterite plural sometimes ends in $-\underline{en}$ in Lancashire. 1

In addition to the usual /os no:z/, /je no:/, and the older /jo: no:n/, the following second person forms of the verb know are also found: /os no:d/ (definitely present tense), /os no:/, /je no:z/.

6.8.1.2. The "habitual present":

When describing habitual behaviour, or their more permanent tastes and opinions, informants use an $-\underline{s}$ ending (i.e. /s, z, $\iota z/)$ on the first person singular. Examples:

```
/a larks it in'de(r)z/
'I likes it indoors'

/a de:nt ni:d tert a sops main/
'I don't need to eat - I sups (drink) mine'

/oats war a li:vz it switst on/
'that's why I leaves it switched on' [household routine]

/a Ornks e'be:t it 'monr e tarm/
'I thinks about it many a time'

etc.
```

The second and third person singular already have -s as their usual present tense ending. However, on the basis of my corpus (which might not be sufficiently large for this generalisation), -s endings are not used in the plural to express habit or custom, although I recorded:

^{1.} Cf. Wright, J. (1905: 8).

```
/wi goz de:n ðie(r) 'regle de:nt wi/
'we goes down there regular(ly), don't we?'
/ðats wot ðe: 0inks 'pi:pel/
'that's what they thinks, people'
```

I am inclined to view such occasional $-\underline{s}$ endings as formations by analogy.

6.8.1.3. The historical present:

Past events are sometimes related in the historical present - a state of affairs which is very common in the languages of the world. The endings used are either the normal present tense endings given in subsection 1 above, or -s (i.e. /s, z, tz/) throughout the conjugation (even after the pronoun they), although second person forms are rare in narratives (except of course in direct speech within narratives, which is not the same thing). The tense is especially typical of jokes, narratives, and indeed any occasion when immediacy and dramatic effect are required in the relation of past events. A speaker may relate a story in the historical present, perhaps having begun with one or more past tense forms, or, as is more likely, he may switch to and fro between historical present and past tense forms - sometimes rather abruptly, which on occasion is reminiscent of Old Norse It is interesting to note that Wackernagel-Jolles, in an examination of spoken, regional North German, drew attention to the extensive use of the present: it is the tense of discussion, statement and explanation, and may be used as a historical present. Examples of the historical

l. Cf. Gordon (1966: 313).

^{2.} Cf. Wackernagel-Jolles (1971: 236).

present:

1. Note the two forms of we get, and the eventual switch to a past form:

/so: wi gets | egent kok | en and 'biliz et
'so we gets.. again (against) the Cock [Hotel].. and Old Billy's
front | wi 'dzogin e'we: bet wons wen wi get
at the front.. we're jogging away all at once when we get
tet kok lad it wipt in/
to the Cock lad [exclamation] it whipped in'

2. Note changes similar to those in (1):

/www get | 'gr:in e'gen 'eni.jsr | en wi get te 'e:(r)mston 'we get.. going again anyhow.. and we get to Urmston ... ŏe: gets it || 03 vs laik en levz mi: | o: wi ... they gets into.. the house like and leaves me.. oh we polts | ŏis 'po:ni e:t et Safs | on | wi tr:k pulls.. this pony out of the shafts.. and.. we took it it jæ:(r)d/ it into the yard' etc.

6.8.1.4. Expanded or periphrastic forms:

I have not noted any tendency on the part of the dialect towards non-expanded forms. Expanded forms are used as in S.E., and their formation is the same:

BE + PRESENT PARTICIPLE

DO + INFINITIVE

The present participle is formed by adding -/in/, e.g.

/seit, 'seitin/ 'shout, shouting'.

^{1.} Cf. Wakelin (1972a: 121f).

6.8.2. Future Tense:

The future tense is formed by using:

- 1. Either SHALL or WILL + INFINITIVE: 1 or
- 2. BE + /be:nt/ 'boun(d) 2 to' or /be:n(t) fot/ 'boun(d) for to' + INFINITIVE; or
- 3. BE + /'gY:int/ 'going to' + INFINITIVE

WILL may be used with all persons as the future auxiliary.

SHALL is used in the first person singular and plural. (2)

and (3) are used in the same way, and tend to indicate a

fairly immediate future. Examples:

/wist gy: e'gen neks je(r)/
'we shall go again next year'

/wot wilt dy:/
'what will thou do?'

/am be:nt tak 'dgeri æ:t tet | @ais'kri:m mon/
'I'm boun(d) to take the jerry (chamber-pot) out to
the ice-cream man'

/am be:nt 'finis op wi 'se:in öis ne:/
'I'm boun(d) to finish up with (by) saying this now'

/we(r) 'gy:int æ: e @ink e'be:t it/
'we're going to have a think about it'

etc.

There is a progressive form in such constructions as:

/al bi 'si:in oi e'foe(r)t wiks oe(r)/

'I'll be seeing thee afore (before) the week's over'

For forms of these auxiliaries see sections 6.8.4.6. and 6.8.4.7.
 Partington (1920: 6) gives boun and compares it with Icelandic búinn, past participle of búa to prepare. See also C.O.D. bound. However, the meaning 'ready to start', which is given in both works, does not fully reflect the form's rôle in periphrastic future constructions in the Farnworth dialect, where its meaning is 'going (to)'.

In modified speech, the same alternatives apply, except that the more residual forms of SHALL occur less frequently. Modified pronunciations of /bɛ:nt/ and /bɛ:n(t) fot/ are /bɜʏnt, bɜʏn(t) fot/.

6.8.3. Past Tenses:

6.8.3.1. Preterite:

Plural endings in -en were only used by one informant, and must now be accounted a very residual feature indeed. I have only heard them on strong verbs, e.g. /we: 'ky:men/ 'we For further discussion see the description of the Present Tense.2 Generally speaking, strong verbs change their root vowel, and the subject is not reflected by the verb ending For details of the verbs involved, and the vowel alternations, see the Table of Irregular Verbs below. verbs are treated elsewhere. 3 The verb slink, preterite and past participle /slonkt/, appears to be subject to a double (strong and weak) past tense formation. 4 A number of irregular verbs change neither the vowel nor the ending, e.g. /fot/ = infinitive, preterite and past participle of a verb meaning 'fetch'. These verbs are included in the table. are verbs in which final /d/ becomes /t/ in the preterite and past participle, e.g. /bild - bilt; graind - graint/ 'buildbuilt: grind-ground'.

^{1.} Cf. section 6.8.4.6.

^{2.} Section 6.8.1.1.

^{3.} See section 6.8.4.

^{4.} Verbs belonging to different historical classes may have a vowel change + alveolar, e.g. /ki:p - kept/ 'keep-kept', but it is curious that slink should do this.

In the case of regular verbs, the preterite and past participle are formed in /t/ after /p, k, f, θ , s, ζ , t ζ / and quite often after /m, n, 1, r and ι /. In the case of /r/, a special note is needed. Since historical /r/ is often not pronounced nowadays in relatively unstressed syllables at the end of a word, or after / ι e/, then /t/ may occur after /e, ι e/. Examples of preterites and past participles in /t/ after /m, n, 1, e(r), ι e(r), ι e/:

/re:nt/ 'rained'; /'apnt/ 'happened'; /'fm:snt/
'fastened'; /ors:nt/ 'drowned'; /'fri:tnt/
'frightened'; /te:(r)nt/ 'turned'; etc.

/'apmt/ 'happened'; /So:mt/ or /Se:mt/ 'shamed,
ashamed'; /tlaimt/ 'climbed'; /fre:mt/ 'framed';
/tle:mt/ 'claimed'; etc.

/kilt/ 'killed'; /nielt/ 'kneeled'; /ke:lt/ kaled '(to be) overtaken, lost one's turn'; /'bɛi(j)elt/ 'boiled'; /'ratlt/ 'rattled'; /'tiklt/ 'tickled'; etc.

/fiə(r)t/ 'frightened'; /tliə(r)t/ 'cleared'; etc.
/'tɛi(j)e(r)t/ 'tired'; /'wonŏe(r)t/ 'wondered';
/'boge(r)t/ 'buggered'; /'bɔðe(r)t/ 'bothered'; etc.
/'karıt/ 'carried'; /'werıt/ (also /'werıtıd/)

'worried'; and probably also /'bɛrit/ 'buried'.

In modified speech, there is a tendency to use /t/ after the same set of consonant phonemes as in S.E. However, this is a tendency, not an absolute rule (e.g. /kilt/ 'killed' from a speaker of a variety of N.S.), just as the use of /t/ after /m, n, l, e(r), ie(r), ie(r), ie(r), in the base dialect does not apply in some cases, is optional in others, and customary in others still. After final /d, t/, the preterite and past participle

ending is -/1d/, but a final /d/ must or may convert to /t/
in bend, bind, build, grind, lond (land, lend), rebuild, send,
spend, unbend and unbind. In all other verbs, the preterite
and past participle ending is /d/. Examples of regular
formations:

```
'step-stepped' /step - stept/
'fold-folded' /f3rd - 'f3rdid/
'beg-begged' /beg - begd/
```

Past events may be related in the historical present.

Details are given in the description of the <u>Present Tense</u>.

On occasion it will not be clear which tense is being used, e.g.

/so: wi kom bak e'gen || gets 'setld in e'gen |
'so we come/came back again, gets settled in again,
wen or ste:(r)t e'gen/
when they start again'

6.8.3.2. <u>Perfect</u>:

Irregular formations of the past participle are given in the Table of Irregular Verbs below. Strong verbs show vowel changes when their participles are compared with the infinitive and/or preterite forms, e.g. drink. Some additionally end in /n/, e.g. let, while others end in /d, t/, e.g. keep. Other irregular verbs may show no change at all, e.g. fot 'fetch'; or they may have a participle in /n/, e.g. shut; or /d/ of the stem may become /t/, e.g. grind.

Endings on regular participles follow the pattern given for regular preterites in the preceding subsection, i.e. they

^{1.} See section 6.8.1.3.

are /t, ιd or d/, and are formally predictable, except for some variation between /t/ and /d/ after /m, n, l, e(r), $\iota e(r)$, $\iota e(r)$, $\iota e(r)$, $\iota e(r)$, $\iota e(r)$,

A construction involving the past participle is:

HAVE + been and + PAST PARTICIPLE

Thus:

/av dgos bin en 'kE:ntid om/
'I've just been and counted them'
/i:z bin en gon en don it/
'he's been and gone and done it'

The construction may be used to lend emphasis, express surprise, and so on. A past participle is used after BE, where speakers of other varieties of English might prefer a progressive or continuous form, in the following:

/ŏε we:nt sted op/
'thou wasn't stood up'

/i: we(r) sit int sink/
'he was sit (sat) in the sink'

/a 'pelez 'wantid bi stod op/

'I always wanted to be stood up' [when digging coal]

A past participle also occurs in:

/ðe(r) we(r) 'peloz men 'we:tin fe(r) wek | stod it 'there were always men waiting for work, stood in the pit jæ:(r)d/pit yard'

It is impossible to say whether stood is part of a construction with BE in this last case.

A distinctive construction is:

HAVE + been + given over + PRESENT PARTICIPLE

stopped
(PAST PARTICIPLE)

```
e.g.
/av bin giv oə(r) 'wekin 'sitti:n je(r)/
i.e.
'It is 18 years since I stopped working.'
```

A further distinctive construction involves the progressive form of the perfect. Very many verbs may be substituted for <a href="https://hoover.ncb/hoover

```
/av bin 'Y:verin o: 'moə(r)nin/
'I've been hoovering all morning.'
```

This sentence, and others of the same type, may mean what they literally appear to mean. Alternatively, they may mean virtually the opposite: the above construction means 'I have been intending to hoover all morning, but unfortunately I have not yet done so'. The construction might be followed by a further remark, which in the event might well serve to clarify it, but equally it can stand alone. Furthermore, the construction, which is common, is not disambiguated by stress or intonation as far as I can tell. Context would appear to be crucial. I believe that the construction could derive from the fuller alternative:

/av bin fer 'Y:verin o: 'moe(r)nin/
'I've been for hoovering all morning.'

is also a possible interpretation).

There is occasional evidence of double past forms:

```
/'weritid/ 'worried' preterite and past participle
/'slonkt/ 'slunk' preterite and past participle
/i: jv:s sed/ 'he used to said'
/'evri taimp 'fo:mon 'bleitse(r) jv:s won op(r)t 'monidge(r)/
'every time the foreman bleacher used to won over (win against) the manager'
/a 'jv:ste 'no:tist/
'I used to noticed' (theoretically 'I used to have noticed'
```

The use and formation of the perfect tense is generally as in S.E. However, there seems to be a trace of BE as an auxiliary. There are occasions when it is difficult to be certain: firstly, a past participle may be used adjectivally, in which case it will be preceded by BE; secondly, /1z/, when unstressed, could represent either <u>is</u> or <u>has</u>; thirdly, with rare constructions, the possibilities of error and reformulation cannot be overlooked. Examples, including pluperfect forms:

1. [Modified speech]:

/wit; iz moore les bi'kom mi 'seken ho:m/
'which is [has?] more or less become my second home'

- 2. /ai o: ði:z θingz iz 'o:tθe(r)t | 'tɛribli/
 'aye all these things is altered.. terribly'
- 3. /ɔ:l ðɛmz don ø'we: wi:/
 'all them's done away with'
- 4. /o: δi:z θingz iz 'vanis ne:/
 'all these things is vanished now'
- 5. /pel ðem taimz iz | flayn e'we:/
 'all them times is.. flown away'
- 6. /won ev iz me:ts iz kom/
 'one of his mates is come'
- 7. /it wer 'o:fn kom of/
 'it were often come off'
- 8. /ði we(r) kom op ðie(r)/
 'they were come (had come) up there'
- 9. /ðat we(r)t bes wot wer 'eve(r) kom iə(r)/
 'that were the best what were ever come here'
 (i.e. 'that was the best horse which had ever come here')
- l. could involve <u>is</u> or <u>has</u> (unstressed), although become is a classical contender for BE as auxiliary - cf. N.H.G.

werden, 0.E. weor an. 3. is presumably a case of a past participle used adjectivally. However, the other examples are more convincing, and the pattern involved would appear to be that BE may function as the perfect tense auxiliary with intransitive verbs of motion or change of state, rather as in N.H.G. and N.F. This is a pattern which is also thought to have applied in 0.E. with the emergence of the perfect and pluperfect. There is not as much evidence as scholars would wish, but Zimmermann writes that "...be was used with certain intransitive verbs denoting motion or change, so-called 'mutative' verbs, e.g. cuman, weor an". The use of BE as the perfect tense auxiliary in the dialect would bear further investigation, although such use is not extensive.

Miscellaneous occurrences of the past participle:

/it wer e:f pe:s siks te:(r)nd/

'it were half-past six turned' = 'turned half-past six'

/e nais 'te:stid 'latis/
'a nice tasted lettuce'

6.8.3.3. Pluperfect:

On the possibilities of BE as auxiliary, see the preceding subsection. The pluperfect may be formed as in S.E., but in certain contexts

HAD (+ NOT) + HAVE + PAST PARTICIPLE is customary. This formation is typical of conditional clauses.

^{1.} Zimmermann (1973: 107).

Examples:

```
/iv 'i:d: 'dv:tsen 'adnt e kom/
'if Edith Dootson hadn't have come'
/iv ad e θ3vt/
'if I'd have thought'
/iv we:d e no:n/
'if we'd have known'
/iv i:d e bin kilt e(r) sed wid e stil at te l3vft/
'"If he'd have been killed," her (she) said, "we'd have still had to have laughed."'
etc.
```

The same construction is very prominent in modified speech, where /e/ may become /ev/, e.g.

/a wis ad ev por e 'kæ:(r)digen on/
'I wish I'd have put a cardigan on.'

It is used after <u>if</u>, <u>I wish</u>, <u>I wished</u>, <u>if only</u> in both residual dialect and modified speech.

6.8.3.4. Regular verbs:

The following <u>Table of Regular Verbs</u> contains a selection of regular forms, including:

- a) verbs which are irregular in S.E., e.g. blow
- b) verbs which are also irregular in the dialect, e.g. choose
- c) verbs more often or alternatively irregular in S.E., such as thrive, bide, blend, etc.
- d) verbs with a different basic form from the S.E. equivalent, such as scrat, yar.

TABLE OF REGULAR VERBS

Meaning	Infinitive	Preterite	Past <u>Participle</u>	Notes
'bide'	/baid	'baidid	'baidid	
'blend'	blend	'blendid	'blendid	
'blow'	blo:	blo:d	blo:d	
'call'	kə:	kə:d	kə:d	
'cast'	km:st	'kæ:stid	'kæ:stid	
'catch'	katS	katSt	kat \f	
'choose'	t[Y:z	t[Y:zd	t[Y:zd	Also irreg.
'clothe'	klo:ð	klo:ðd	klo:ðd	
'crow'	kro:	kro:d	kro:d	
'deal'	dıəl	diəld	dıəld	Also irreg.
'dig'	dig	dıgd	dıgd	
'draw'	ðra:	ðrə:d	ðrə:d	Less often
'fall'	fo:	fo:d	fo:d	irreg. Also irreg.
'flee'	fli:	fli:d	fli:d	Verb rare,
'forecast'	foe(r)ke:st	'foə(r),kæ:stid	'foə(r)ıkæ:stı	also irreg.
fotch = 'fetch'	fots	fɔtst	fɔtʃt	Cf. fot (irreg.)
'gild'	gild	gildid	gildid	Rare
'grow'	gro:	gro:d	gro:d	
'hang'	ang	angd	angd	Also irreg.
'hear'	19(r)	ıə(r)d	iə(r)d	Also irreg. cf. also yar below.
'heave'	ειν	ειvd	εινd	
11 1		εινα	εινα	
'hold'	зyd	tardid	'ardid	Also irreg.
'kneel'				Also irreg.
	э ү d	rardid	rdid	•
'kneel'	ard nuel no:	'aydid niəlt	'sydid niəlt	•
'kneel' 'know'	ard nuel no:	'ardid nielt no:d	'ardid nielt no:d	•
'kneel' 'know' 'learn; teach	3Yd nuəl no: 'læ:(r)n	'aydid nielt no:d lm:(r)nt	'ardid nielt no:d læ:(r)nt	•

^{1.} The irregular form is used adjectivally: /pa(r)s orp:n/'horse-drawn'.

•		•	Past	
Meaning	Infinitive	Preterite	<u>Participle</u>	Notes
<pre>overlie = / oversleep'</pre>		oə(r)'laıd	o⊖(r)'laıd	Less often irreg.
'saw'	so:	so:d	b:ca	Also irreg.
'say'	se.	sed	sɛd	Also irreg.
scrat = 'scratch'	skrat	skratid	skratid	Also irreg.
'see'	si:	si:d	si:d	Also irreg.
'sew'	so:	so:d	so:d	Also irreg.
'shake'	∫e:k	<pre>\$\{\epsilon\} e:kt</pre>	<pre>\$e:kt</pre>	Also irreg.
'shave'	Seiv	∫ε ι vd	{eivd	·
'shed'	<pre>\$i:d</pre>	'Si:did	'Si:did	Also irreg.
'shine'	ςειη Sain	ςειnd Saint	ςειnd Saint	Also irreg.
'shit'	Sit	'Sitid	'Sitid	More often irreg.
'shoe'	7 x:	<pre>{ Y: d</pre>	<pre> Y:d </pre>	
'show'	So:	<pre>{o:d</pre>	<pre>\$o:d</pre>	Also irreg.
'shred'	Sred	'Sredid	'fredid	
'smell'	smel	smelt	smelt	
'sow'	so:	so:d	so:d	Occasionally irreg.
'speed'	'spi:d	'spi:dld	'spi:did	
'spell'	spεl	spεlt	spelt	
'spill'	spil	spilt	spilt	
'spin'	spin	spint	spint	More often irreg.
'spoil' 1	'sperel	'sperelt	'sp&lelt	
'steal' 1	'sterel	'sterelt	'stɛɪelt	More often irreg.
'swear'	swe:(r)	swe:(r)d	swe:(r)d	Also irreg.
'sweat'	swet	'swetid	'swetid	Also irreg.
'teach'	teits	teitst	teitst	Also irreg.
'tear'	te:(r)	te:(r)d	te:(r)d	Also irreg.
'thrive'	θraιν	θraιvd	θraινd	
'throw'	θro:	θro:d	θro:d	Less often irreg.
'wear'	we:(r)	we:(r)d	we:(r)d	Also irreg.
'weave'	WELV	weivd	weivd	Also irreg. More often
'wet'	wet	'wetid	wetid	more often irreg.
'work'	wek	wekt	wekt jæ:(r)d/	Alaa immaa
<u>yar</u> = 'hear'	jæ:(r)	jæ:(r)d	log v (I.) (T.)	Also irreg.

^{1.} Forms may also be phonemicised with /(j)/ after /ει/.

6.8.3.5. <u>Irregular verbs</u>:

In the following table it will be noted that a number of verbs which are irregular in S.E. are not present, e.g. to This is because such verbs are not in the corpus, and because special questions designed to elicit them failed to do (When asked directly about verbs such as to arise, so. informants said that they knew what the words meant, but that they did not use them.) These observations do not of course preclude the possibility of such a verb being used on some particular occasion. In the event, it is reasonable to predict that the preterite would follow the S.E. pattern, but be pronounced with a pronunciation appropriate to the area. However, one would also have to predict that the past participle might possibly take the same form as the preterite - it will be seen from the Table of Irregular Verbs that many verbs pattern in this way in the dialect. Some verbs are known to informants in a mainly passive way, perhaps from religion, e.g. to beget. Informants feel, as I do, that these verbs cannot really be accounted a part of the dialect. On the other hand, the table contains one or two verbs which are not irregular in S.E.: e.g. the strong verb to squeeze.

A number of verbs have both regular and irregular forms. Where appropriate, this fact is indicated in the notes. In view of the existence of several forms for some parts of some verbs, the notes offer brief guidance as to the rarity or commonness of forms where possible. These judgements reflect: a) the number of occurrences of a form in the corpus (including forms deliberately elicited by questionnaire techniques); b) informants' opinions concerning which form(s) they use more often.

TABLE OF IRREGULAR VERBS

			Past	
Meaning	Infinitive	<u>Preterite</u>	Participle	Notes ,
'be'	/bi:/, /bi/	/we(r)/	/bin/	•
'bear'	/be:(r)/	/boə(r)/	/boə(r)n/	
'beat'	/bjet/ /bi:t/	/bjet/ /bi:t/	/'bjetn/ /'bi:tn/	Older forms More common
'become!	/bi kom/	/bi'kom/	/bi 'kom/	
'begin'	/bi'gin/	/bi'gon/	/bl gon/	
'bend'	/bend/	/bent/	/bent/	Also regular
'bet'	/bet/	/bεt/	/bet/	,
'bethink'	/bi'θink/	/bι'θзγt/	/bi'03yt/	
'bid'	/bid/	/bid/	/bid/	
'bind'	/baind/	/bon/ /be:nd,be:nt/	/bon/ /be:nd,be:nt/	Older forms Also regular
'bite'	/bait/	/bit/	/'bitn/ /bit/	
'bleed'	/bli:d/	/blɛd/	/blɛd/	,
'break'	/breik/	/bro:k/	/'brokn/ /bro:k/	
'breed'	/bri:d/	/bred/	/bred/	
'bring'	/bring/	/brayt/	/brayt/	
'build'	/bild/	/bilt/	/bilt/	
'burst'	/bast/	/bost/	/bost/	Also regular
'buy'	/bei/	/bsyt/	/bs rt/	
'choose'	/t[Y:z/	/tso:z/	/'tso:zn/	Also regular
'cling'	/tling,kling/	/tlong,klong/	/tlong,klong/	
'clink'	/tlink,klink/	/tlank,klank/	/tlonk,klonk/	· ·
clod='throw'	/tlad,klad/	/tlod,klod/	/tlod,klod/	
'come'	/kom/	/kY:m/	/ky:m/ Olo	der./ky:m/ rare
		/kom/	/kom/	past participle. Now more com- mon
'cost'	/ko:st/	/kp:st/	/ko:st/	
'creep	/kri:p/	/krept/	/krept/	
'cut'	/kot/	/kot/	/kot/	e service de la company de la
'dare'	/dm:(r), da / /de:(r), de(r)/	/dest,dost/	/dm:(r)d/ /de:(r)d/ /dest,dost/	Also regular
'deal'	/diəl/	/dɛlt/	/dclt/	Also regular
'dig'	/dig/	/dog/	/dog/	Also regular
WIE.	/ == 6/	<i>y</i> o/	, - 0,	

^{1.} For further forms and discussion see section 6.8.4.10.

			Past	
Meaning	<u>Infinitive</u>	Preterite	Participle	Notes
'dive'	/daiv/	/do:v/	/do:v/	Also regular
'do'	/dr:,do/(/de/)	/did/	/don/	<u> </u>
		/don/1		Less common preterite
'draw'	/ðr::/	/òrɔ:d/ /òrv:/	/ðrɔ:n/ ²	Also regular Less common preterite
'dream'	/ðri:m/	/θrεmt/	/oremt/	
'drink'	/ðrink/	/ðrank/	/ðrank/ ³	
'drive'	/ðrειν/	/oro:v/	/'oro:vn/	
			/'ðrovn/ /'ðrivn /	Now the most
	,		/ðro:v/	common form
'eat'	/ert/	/εt/	/'εtn/	The most common set
	/i:t/	/ειt/ /i:t/ /εt/	/i:t/ /εt/	
'fall'	/fo:/	/fo:d/ /fel/	/fo:n/ /fcl/ Less	Also regular usual.Modified
'feed'	/fi:d/	/fed/	/fed/	
'feel'	/fiel/	/felt/	/fclt/	
'fight'	/feit/	/fsyt/	/fayt/	
'find'	/faind/	/fon/ /fe:nd/	/fon/ Older /fe:nd/	Still common
'flee'	/fli:/	/fled/	•	rare. Also
'fling'	/fling/	/flong/	/flong/	egula r 4
'fly'	/flai/	/flayn/ /fly:/	/flayn/ /flo:n/	Older forms
		/IIY:/	/flo:n/ /fly:/	Uncommon
'forbid'	/fe(r)'bid/	/fe(r)'bid/	/fe(r)'bidn/	Preterite rare
'forget'	/fe(r)'gst/		/fe(r)'gstn/	The most common set
		/fe(r)'got/	/fe(r)'gi:t/ /fe(r)'gotn/ /fe(r)'got/	Modified Modified

The use of the past participle as a preterite is quite uncommon

4. /flayn/ was also given, cf. 'to fly'.

in the dialect - the reverse is more often the case.

Especially when used adjectivally, e.g. /'pa(r)s'oro:n kæ:(r)t/
'a horse-drawn cart'.

^{3. /&#}x27;oronkn/ may be used adjectivally, e.g. /e 'oronkn tsap/ 'a drunken chap'.

Meaning	<u>Infinitive</u>	Preterite	Past <u>Particip</u>	<u>le Notes</u>
'forgive'	/fe(r)'gl, fe(r)'glv/	/fe(r)'gid/ /fe(r)'gin/ /fe(r)'giv/	/fe(r)'g	ın/
'forsake'	/fe(r)'se:k/	/fe(r)'sY:k/	/fe(r)'s	Y:k/ e:kn/
fot='fetch'	/fot/	/fət/	/fot/	
'freeze'	/fri:z/	/fro:z/	/'frozn/ /fro:z/	The more common Cannot be used adjectivally of a person
'get'	/gɛt/	/gi:t/ /'gɛtn/ /gɛɪt/	/'getn/ /gi:t/	The most common set Uncommon
'give'	/gi,giv/	/gan/ /gin/ /gid/ /giv/ /gi/ /ge:v/l	/gan/ /gin/ /gid/ /giv/ /gi/ /'givn/	Traditional Traditional.Common Traditional.Common Quite common Less common Modified
'go'	/gy:,go/	/went/	/gon/	
'grind'	/graind/	/graint/ /gre:nd,gre:nt/,	/graint/ /gre:nd,gre	Preferred forms
'hang'	/ang/	/ong/	/ong/	Also regular. The irreg.forms2are more common
'have'	/a,æ:,av/	/ad/	/ad/	
'hear'	/lə(r)/	/e:(r)d/	/e:(r)d/	Somewhat modified. Also regular. See also yar below.
'hide'	/aid/	/1d/	/'idn/ /id/ /'idid/	Rare
'hit'	/lt/	/1 t/	/1 t/	
'hold'	/3¥d/	/εld/	/Eld/	Also regular
'hurt'	/et/	/et/	/et/	
'keep'	/ki:p/	/kɛpt/	/kept/	
'kneel'	/niel/	/nelt/	/nɛlt/	Also regular

 ^{/&#}x27;gidn/ was also recorded for the preterite from one informant.
 The strong forms may be used with the meaning 'to execute a person'.

			Past	
Meaning	<u>Infinitive</u>	Preterite	Participle	Notes
'knit'	/ni t/	/nit/	/nit/	Also regular
'lay'	/le:/	/le:d/	/le:n/ <u>to 1</u> :	s of the verb ie are often
'lead'	/li:d/	/1ed/	/led/ used:	cf. <u>lie</u> below.
'lean'	/lein,li:n/	/lent/	/lent/	Also regular
'leap'	/li:p/	/lept/	/lept/	
'leave'	/lev/	/lɛft/	/left/	
'lend'	/lənd/ /land/ /lɛnd/	/lont/ /lant/ /lent/	/lont/ /lant/ /lent/	Older Older
'let'	/li:t/	/li:t/	/'lɛtn/ Older	forms: still
	/lɛt/	/lɛt/		common. informant gives tn/ for p.p.
'lie'	/lai/	/laid/ /le:d/	/le:n/ can h /le:d/ same	regular forms be used with the meaning as the ag forms.Cf. <u>lay</u>
'light'	/li:t/	/li:t/ /l:	i:t/,/'li:tn	Older ^l
	/lait/	/lit/	('litn)/ /lit/ More	modified
'lose'	/lo:z/	/lo:st/	/lo:st/	infinitive
'make'	/me:k,mek/	/me:d/	/me:d/	
'mean'	/miən,mi:n/	/ment/	/ment/	i .
'meet'	/mi:t/	/met/	/met/	
'mislay'	/mis'le:/	/mis'le:d/	/mis'le:n/	Also regular
'mistake'	/mis'te:k/	/mis'ty:k/	/mis'tan/	Older
			/mis'ty:k/ /mis'te:kn/	Modified
'misunderstand' /misionőe(r)'stod/ /misionőe(r)'ston(d), /misponőe(r)'stod/ misionőe(r)'stan(d)/				
'mow'	/mo:/	/mo:d/	/mo:n/	Also regular
'outbid'	/e:t'bid/	/e:t'bid/	/e:t'bid/	
'outdo' 'outgrow'	/ε:t'dY:/	/s:t'did/	p	nly past articiple licited
'overcast'			/oə(r)'kæ:st/0	

^{1.} The first set of forms is identical to forms of the verb to let. <u>Light</u> means 'to set fire to; to alight, to land', and /li:t on/ means 'to meet by chance'.

Meaning	Infinitive	<u>Preterite</u>	Past <u>Participle</u>	Notes
'overcome'	/oə(r)'kom/	/oə(r)'ky:m/	/oə(r)'kom/	
		/oə(r)'kom/		
'overdo'	/cə(r)'dy:/	/oə(r)'dıd/	/oə(r)'don/	
'overhang'	/op'rang/	/oe'rong/	/oə'rang/	
'overhear'	/oə'rlə(r)/	/oə're:(r)d/	/oə're:(r)d/	Modified. Cf. o'eryar below
<pre>overlie = 'oversleep'</pre>	/oə(r)'laı/	/oə(r)'le:d/	/oə(r)'le:d/ /oə(r)'le:n/	More often
oversieep			\09(L).Te:U\	regular
'overrun'	/oə'ran/	/oə'ran/	/oə'ran/	
'overshoot'	/oə(r)'\x:t/	/oə(r)'[ot/	/tc?'(r)eo/	
'oversleep'	/oə(r)'sli:p/	/oə(r)'slεpt/	/oə(r)'slɛpt/	
'overtake'	/oə(r)'te:k/	/oə(r)'tY:k/	/oə(r) tan/	
			/oə(r)'takn/ /oə(r)'tɛkn/	
			/oə(r)'tY:k/	
o'eryar = 'overhear'	/oə(r)'jæ:(r)/	/oə(r)'jæ:(r)d/	/oə(r)'jæ:(r):	n/
'peel'	/'psiel/	/pilt/	/pilt/	
'put'	/pat/	/pat/	/pat/	
'read'	/ri:d/	/red/	/red/	
'rebuild'	/ri:'bild/	/ri:'bilt/	/ri:'bilt/	
'reset'	/ri:'sɛt/	/ri:'sɛt/	/ri:'sɛt/	
'rid'	/rid/	/rid/	/rid/	
'ride'	/raid/	/ro:d/	/'ridn/	
			/ro:d/	
'ring'	/ring/	/rong/	/rong/	
'rise'	/raiz/	/ro:z/	/'rizn/ /ro:z/	
'run'	/ron/	/ron/	/ron/	
'saw'	/sp:/	/sɔ:d/	/so:n/	Also regular
'say'	/se:/	/sɛd/	/sed/	Also regular
<pre>scrat = 'scratch'</pre>	/skrat/	/skrat/	/skrat/	Also regular
'see'	/si:/	/sin/	/sin/	Also regular
	, ~-·,	/ 7	/si:n/	More modified
'sell'	/sɛl/	/sayd/	/sayd/	
'send'	/send/	/sent/	/sent/	

Meaning	Infinitive	Preterite	Past <u>Participle</u>	Notes
'set'	/set/	/si:t/ /sɛt/	/si:t/ /sɛt/	Older.Cf.sit
'sew'	/so:/	/so:d/		ar form more
'shake'	/\$e:k/	\2 x : K\	/'[Y:kn/ /[Y:k/ /'[e:kn/	Older. Also regular Modified
'shame'	/se:m/	/So:mt/	/So:mt/	Old forms. Also regular
'shed'	/Si:d/ /Sed/	/sed/	/sed/	Older Also regular
'shine'	/sein, sain/	/son/		ar forms more
		/Son/	/son/	Less common
'shit'	/Sit/	/Sit/	/'Sitn/ /Sit/	Older
'shoot'	/\(\tau : t/	/sot/	/spt/	; !
'show'	/so:/	/So:d/	/so:n/	Also regular
'shrink'	/Srink/	/Sronk/	/Sronk/	
'shut'	/sot/	/Sot/	/'Satn/ /Sat/	Older
'sing'	/sing/	/song/	/song/	1
'sink'	/sink/	/sonk/	/sonk/	
'sit'	/si:t/ /sit/	/si:t/ /sit/	/si:t/ Older /sit/	set of forms
'sleep'	/sli:p/	/slept/	/slcpt/	
'slide'	/slaid/	/slid/	/slid/	
'sling'	/sling/	/slong/	/slong/	
'slink'	/slink/	/slonkt/ /slonk/	/slonkt/ /slonk/	Preferred forms
'slit' 'smite'	/slit/	/slit/	/slit/ /'smitn/ Past	participle only
'sow'	/so:/	/so:d/	/so:n/ Regula	ar form preferred
'speak'	/spcik/	/spo:k/	/'spokn/ /spo:k/	Less usual
'spend'	/spend/	/spent/	/spent/	
'spin'	/spin/	/spon/	/spon/	Also regular

^{1.} Only in the phrase "(not) to be smitten with".

<pre>Meaning Infinitive Preterite 'spit' /spit/ /spat/ 'split' /split/ /split/</pre>	Past Participle Notes /spit/ Less usual /split/ /spred/ /sprong/ /'skwozn/ Dialectal set
/spat/	Less usual /split/ /spred/ /sprong/
'split' /split/ /split/	/spred/ /sprong/
, _ , _ , _ , _ , _ , _ , _ , _ , _	/sprong/
'spread' /spred/ /spred/	
'spring' /spring/ /sprong/	/!skwogn/ Diplectal set
'squeeze' /skweiz/ /skwoiz/ /skwiiz/ /skwoiz/	/'skwo:zn/ Modified set
'stand' /ston,stond/ /stod/ /stan,stand/	/stad/
'steal' /'starel/ /'starel/	/'starlen/ Also regular
/sto:1/	/starn/ / sto:l / More modified /'sto:len/ More modified
'stick' /stik/ /stok/	/stak/
'sting' /sting/ /stong/	/stang/
'stink' /stink/ /stonk/	/stank/
'stride' /s0raid/ /s0roid/	/sθro:d/
'strike' /s0ralk/ /s0rok/	/s0rok/
'string' /s0ring/ /s0rong/	/s0rong/
'strive' /s0ralv/ /s0ro:v/	/'s0rivn/
'swear' /swe:(r)/ /swoə(r)/	/swoə(r)n/ Also regular
'sweat' /swet/ /swet/	/swet/ Also regular
'sweep' /swi:p/ /swept/	/swept/
'swell' /swel/ /sweld,swe	lt/ /'swo:len/ Also regular
'swim' /swim/ /swom/	/swom/
'swing' /swing/ /swong/	/swong/
'take' /tak/ /ty:k/	/tan/ Older
/tek/	/'takn/ Common /'tɛkn/ Common /tr:k/
'teach' /tsits/ /tsyt/	/tayt/ Also weak. The verb /læ:(r)n/ is often used.
'tear' /te:(r)/ /to=(r)/	/tɔə(r)n/ Not much used. Rip is preferred. Also regular
'tell' /tsl/ /t3rd/	/tard/
think' /01nk/ /03Yt/	/03 Y t/

			Past	
Meaning	Infinitive	Preterite	Participle Participle	Notes
'throw'	/θro:/	/0ry:/	/0ro:n/	Regular forms more usual.
'thrust'	/θrast/	/0rast/	/0rast/	
'tread'	/0red,0reid/	/0rod/ /0rod/	/θrod/ /'θrodn/	
'unbend'	/om'bend/	/om'bent/	/om'bent/	Also regular
'unbind'	/om'baind/	/om'be:nd, om'be:nt/	/om'be:nd, om'be:nt/	Also regular
'understand'	<pre>/onde(r)'ston .onde(r)'stond .onde(r)'stan .onde(r)'stand</pre>	/onde(r)'stod/	√anðe(r)'stad	
'undertake'	/anðe(r)'te:k/	/onðe(r)ty:k/	<pre>/onde(r)'tan/ /onde(r)'takn, /onde(r)'tekn, /onde(r)'tr:k,</pre>	/
'undo'	/on'dr:/	/on'did/	/an'dan/	
'upset'	/op'set/	/op'si:t/ /op'sct/	/op'si:t/ /op'sɛt/	Older forms
'wear'	/we:(r)/	/woə(r)/	/wəə(r)n/	Also regular
'weave'	/welv/	/weivd/ /wo:v/	/'wovn/ /wo:v/ /'wo:vn/	Also regular Modified Modified
'wed'	/wed/	/wed/	/wed/	
'weep'	/wi:p/	/wept/	/wept/	
'wet'	/wet/ /wi:t/	/wet/ /wi:t/	/wet/ /wi:t/	Also regular Older
'win'	/win/	/won/	/won/	
'wind'	/waind/	/wan/	/won/	Older forms. Still common.
		/we:nd/ /wond/	/we:nd/ /wond/	Less common
'withdraw'	/wiððro:/	/wið'ðry:/	/wið'ðrə:n/	Also regular
'wring'	/ring/	/rong/	/rong/	
'write'	/rait/	/ro:t/	/ro:t/ /'ritn/	Less common
yar='hear'	/jæ:(r),ja/	/jæ:(r)d/	/jæ:(r)n/	Less often regular

Users of modified speech employ a number of forms marked "modified" in the above table, together with other forms not included here which are as in S.E., but with a pronunciation appropriate to the area. Quite a number of the irregular dialectal forms given in the table are retained by many modified speakers. Especially common are forms of the type:

/drink dronk dronk/
/ring rong rong/

and so on. There are speakers with considerably modified accents (Educated Northern), and a high degree of formal education, who use these forms with utter consistency. There is another group of modified speakers, however, which, despite the modification that it makes, nonetheless preserves the more general principle that the preterite and the past participle are the same. Thus:

/drink drank drank/
/ring rang rang/

and so on with all verbs of the same pattern.

It is probable that a number of older verb forms survive on the outskirts of the Greater Bolton area, that is the more rural outskirts. A gentleman from Westhoughton (not an informant), for instance, gave infinitives and preterites which were in no way different from those in the table above, but in addition to various familiar past participles, he produced several not recorded previously, all of which are characterised by an /n/ ending:

It should be noted that these Westhoughton forms are not the result of a systematic study, but they are perhaps sufficient to suggest that a thorough investigation of the outlying areas might yield a good deal more material.

6.8.4. Modal Verbs, Auxiliary Verbs, Anomalous Finites:

The description is of those verbs which are known as modal and auxiliary verbs, and as anomalous finites. The forms of the negative interrogative are of two types in question tags:

- 1. I can do it, can't I?
- 2. I can't do it, can I?
- 3. I can't do it, can I not?

After a positive statement, the type <u>can't I</u>? is compulsory; after a negative statement, the tag may be positive <u>can I</u>?, or negative, in which case it will be of the type:

AUXILIARY + PRONOUN + not.

The same rules apply in modified speech.

In questions other than tags, residual dialect and modified speech both often tend to use the pattern

Can I noan/not have one?

i.e. AUXILIARY + SUBJECT + NEGATIVE PARTICLE ... rather than the type

Can't I have one?

which is regular in spoken standard usage. The former pattern is - like the rules governing question tags - a general feature of the dialect, and many examples of it can be found. The actual dialect forms are given under the relevant verbs in the following description.

There are some second person singular interrogative forms in the dialect, which are very distinctive from a comparative point of view. In these forms, thou would appear to have coalesced with or become incorporated into the verb.

Thus:

```
/kont/ 'can thou?' - cf. /kon os, ken os:/ and especially /'kontte/, which also occur

/'katnt/ 'can thou not?'

[k'o?t'] 'could thou?'

[di?t'] 'did thou?'

/'kotnt/ 'could thou not?'

etc.
```

In all such forms, the final /t/ may be very heavily preglottalised (although in continuous conversation glottalisation is sometimes less marked), or the final cluster contains strong glottal stricture, e.g. ['k'at?nt'] 'can thou not?'. Although interrogative forms in which the pronoun coalesces with the auxiliary verb predominate in question tags, they are sometimes used in ordinary second person singular questions too, especially if positive - cf. however the proviso in respect of negative forms in the preceding paragraph.

Examples:

1. /kont ri'membe(r)/ 'can thou remember?'
 /ôs ken ri'membe(r) dʒo: bre:n 'katnt/
 'thou can remember Joe Brown, can thou not?'

In the first example, /kon ôs/ is in free variation with
/kont/, the latter occurring more frequently, however.

With all modal and auxiliary verbs, forms in which the
second person singular pronoun coalesces with the verb
are in free variation with AUXILIARY + PRONOUN constructions
in positive questions. However, the latter may be preferred
where emphasis on the pronoun is required, e.g. /ken ôs:/
'can thou?'. In the positive interrogative, forms of the

type /kont/ 'can thou?' are more frequent than forms of the type /'kontte/ 'can thou?'. The latter type is very residual, and seems to have been largely replaced by the former.

6.8.4.1. Can

Present: /kon, (kan)/, throughout.

Relatively unstressed: /ken/, throughout.

Present negative: /koent, ko:nt, (km:nt)/, throughout.

Present interrogative: /kon, (kan)/, throughout.

Additional second person singular forms incorporating the subject pronoun: /'kontte, 'kantte, kont, (kant)/ 'can thou?' Relatively unstressed: /ken/, throughout.

Present interrogative negative:

(a) /kon, (kan)/ + SUBJECT + NEGATIVE PARTICLE, throughout.

Relatively unstressed: /ken/.

(b) /koent, (ke:nt, ko:nt)/, throughout.

Additional second person singular forms incorporating the subject pronoun: /'kɔtnt, 'katnt/'can thou not?'.1

Preterite: /kod/, throughout.

Preterite negative: /kont, 'kodnt/, throughout.
/'kodn/ is also possible.

^{1.} Further investigation would probably also establish /kɔənt, kæ:nt, kɔ:nt/ as additional second person singular forms incorporating the subject pronoun.

Preterite interrogative: /kod/, throughout.

Additional second person singular forms incorporating the subject pronoun: /'kodte, ko(t)t, $(k\omega dt)/^{1}$ 'could thou?'

Preterite interrogative negative:

- (a) /kod/ + SUBJECT + NEGATIVE PARTICLE, throughout.
- (b) /kont, 'kodnt/, throughout.

Additional second person singular form, incorporating the subject pronoun: /'kotnt/'could thou not?'

The distribution of this verb is as in S.E., except that <u>can</u> is the only modal verb for asking permission in the dialect.

In modified speech, forms with /3, 33, 32/ give way to ones with /a, æ:/, and thou forms are not used.

6.8.4.2. Might

This verb does not have the same range of use as S.E. <u>may</u>. The function of asking permission is carried out by <u>can</u> in the dialect. The range of meaning of <u>might</u> is roughly that given under senses one and four of <u>may</u> in Hornby (1975), i.e. to indicate possibility or probability, and to suggest "there is good reason".²

I have only recorded one present tense form, a presumably reduced, relatively unstressed form in:

/je mi tal/ 'you may tell (you can be sure, you can deduce it from the fact that...)'

The phonemicisation here is meant to suggest the heavy preglottalisation of the final consonant, i.e. /ko(t)t/ = [k'o?t']. Put another way, there are two grammatical units, could + thou, cf. /kot/ 'cut'.
 See the entry for may, Hornby (1975: 533f).

/me:/ and /me: not/ are introduced in modified speech.

Preterite: /mɛt/, throughout.

This form is very prone to strong preglottalisation of the final consonant, or to strong gemination before vowels and hesitation forms, e.g. [ä met? t'e dön] 'I might have done'. The preposition to would appear to be used after might as well.

Preterite negative: /mɛt nɔt/, throughout

Preterite interrogative: /mɛt/, throughout.

The form is again prone to gemination, e.g. /'metti:/ 'might he?'

Additional second person singular form incorporating the pronoun: /'mette/ 'might thou?'

Preterite interrogative negative:

I have only elicited this form as a question tag, where it is /'metnt/ throughout.

The second person singular form elicited incorporated the pronoun, / metnt/ 'might thou not?'

In modified speech, /me:/ forms are used too, although can retains the function of asking permission. /met/ --> /mait/.

6.8.4.3. <u>Mun</u>

Mun has the meaning 'must'.

^{1.} See section 6.10.

^{2.} Further investigation might establish $/m\epsilon(t)t/$ as an additional second person singular form incorporating the subject pronoun.

Present: /mon/, throughout.
Unstressed /mon/.

Present negative: /me:nt/, throughout.

I have heard /mont/ too, but /me:nt/ is unquestionably the regular form for the area, and for all persons of the verb.

Present interrogative: /mon/, throughout.

No additional second person singular form incorporating the pronoun was elicited.

Present interrogative negative:

(a) /me:nt/, throughout.

Additional second person singular form incorporating the pronoun: /'metnt/'must thou not?'

(b) /mon/ + SUBJECT + NEGATIVE PARTICLE, throughout.

The distribution of <u>mun</u> is similar to that of S.E. <u>must</u>. <u>Mun</u> is replaced by <u>must</u> in modified speech; and in residual speech, although <u>mun</u> is still used quite a lot, /most/-forms are also found. Note in the following account of <u>must</u> that there are dialectal forms of <u>must</u> too in the second person singular interrogative.

6.8.4.4. <u>Must</u>

Present: /most/, throughout.

Present negative: /'mos(e)nt/, throughout.

^{1.} Further investigation might also establish /me:nt/ as an additional second person singular form incorporating the pronoun.

Present interrogative: /most/, throughout.

Additional second person singular form incorporating the pronoun: /'moste/ 'must thou?', and probably also /most/.

Present interrogative negative:

(a) /'mos(e)nt/, throughout.

Additional second person singular form incorporating the pronoun: /'mos(e)nt/ 'must thou not?'.

Additional first and third person singular form (i.e. before a pronoun beginning with a vowel): /'mos(e)n/.

(b) /most/ + SUBJECT + NEGATIVE PARTICLE, throughout.

6.8.4.5. Ought

Ought is followed by to or for to.1

Present: /3 vt/, throughout.

Present negative: /'aytnt/, throughout.

Present interrogative: /3Yt/, throughout.

Additional second person singular form incorporating the pronoun: /'artte/ 'ought thou?'

Present interrogative negative:

I elicited these forms only in question tags. In other types of question, informants seemed to wish to use only negative forms of should, have to and must.

^{1.} For forms of these prepositions see section 6.10.

/'sytnt/, throughout.

Additional second person singular form incorporating the pronoun: /'artnt/ 'ought thou not?'

Past: There is a negative past formation:

DIDN'T + OUGHT, throughout.

e.g. /i: 'didnt syt to don it/

'He didn't ought (ought not) to have done it.'

This construction can be used interrogatively.

It is also used in modified speech, where /syt/-forms are replaced by /o:t/-forms.

6.8.4.6. Shall

Present: (a) Forms incorporating the pronoun. The first person singular and plural forms are common-The second and third person forms are rare, and were elicited by using notions such as the following. A parent proposes to buy his child/children a present, if he can afford Thus, he says to one child: /oest a won iv wi ken 'monidg it/ 'Thou shalt have one if we can manage (afford) it.' Similarly, a negative form can be elicited in

> /i: [m:nt av won iv alv syt de wi it/ 'He shan't have one if I've owt (anything) to do with it (I won't let him have one).' However, the second and third person forms were not elicited without difficulty.

Singular 1. /as(t), æ:s(t), ais(t)
2. ŏɛs(t), ŏɛ:s(t)
3. i:s(t), es(t)

wis(t), we:s(t) Plural 1.

such a context as:

jes(t) 2.

ðis(t)

Final /t/ is compulsory before a vowel, but is generally not apparent before a consonant. In the form /ais(t)/, /ai/ is realised by monophthongal variants, or by variants in which the second element is of low prominence.

(b) Stressed form, first person singular and plural: /Sal/. (In section (a) forms, the modal itself is not stressed.) If shall is used when not emphatic, /Sol, Sel/, it constitutes more modified usage. (Perhaps these forms might be elicited for other persons of the verb using devices such as those in subsection (a)).

Present negative:

- (a) PRESENT (a)-FORMS + NEGATIVE PARTICLE.
- (b) Alternative and more modified form: /se:nt/.

Present interrogative: /Sal, (Sol, Sel)/ - only elicited for the first person singular and plural.

Present interrogative negative:

I only elicited these forms for the first person singular and plural.

- (a) /[al/ + SUBJECT + NEGATIVE PARTICLE
- (b) /se:nt/

Preterite: /Sad, Sed/, throughout.

The unstressed form is common.

Preterite negative: /'Sodnt, Sont/, throughout.

^{1.} Cf. section 5.4.2.

Preterite interrogative: /scd/ throughout.

Additional second person singular forms incorporating the pronoun: /'Sadte, Sa(t)t, Sadt/'should thou?'

Preterite interrogative negative:

(a) /'Sodnt, Sont/, throughout.
/Son/ is possible before pronouns beginning
with a vowel.

Additional second person singular forms incorporating the pronoun: /'Sotnt, ('Sotnte, 'Sodnt, Sont)/

(b) /Sod/ + SUBJECT + NEGATIVE PARTICLE throughout.

Shall is used rather less than in S.E. The dialect uses the present forms chiefly in the first person singular and plural. I am not aware of any use of second person interrogative forms as in S.E. Shall you be coming?

The dialect uses will and would for some uses of S.E. shall and should, and shall and will overlap in function within the dialect itself. Both verbs are used to form future tenses and conditionals, and will and would can be used with the first person as well as shall and should. Examples of shall, should:

/wist gy: e'gen neks je(r)/
'We shall go again next year.'
/ast not de dat in 'eri e'gen/
'I shall not do that in [a] hurry again.'
/ai Sed se/
'I should say.' (in the sense "would")

```
/or sod e don/
'They should have done.' (in the sense "ought to")
etc.
```

Should in the sense "ought to" is common. Also very common is the first person singular should in conditionals, when the speaker is stating/estimating/guessing what he would do, or what state he would be in:

```
/a Sed Sot it/
'I should shut it.'

/a Sed dok/
'I should duck.'

/Sed a no:n left/
'[I] should have noan (none) left.'

/ar Sed sr/
'I should say.'

/be:t 'twenriert je r a Sed grv e ges doon/
'about twenty-eight year (years) I should give a guess, John.'

etc.
```

In these examples, should has the meaning 'would'.

In modified speech, /ast/ and /wist/ may still be heard, but they tend to be replaced by /Sal/. Thou-forms are not used, nor is the negative interrogative form /Son/.

6.8.4.7. Will

Present:

(a) Forms incorporating the pronoun:

Singular: 1. /ail, æ:l, al

2. δε:1. δε1. δal

3. iel, il e:(r)l, el itl

Plural: 1. we:1, wil, wel

2. jo:1, joəl, jel

3. ðe:1, ðil, ðel

/ai/ in /ail/ is realised by monophthongal variants, or variants in which the second element is of very low prominence.

(b) /wil/, throughout.

Present negative:

(a) /we:nt, we:nt/, throughout.

The former is the more representative.²

(b) PRESENT TENSE FORMS INCORPORATING THE PRONOUN
+ NEGATIVE PARTICLE, throughout.

Present interrogative: /wil/, throughout. /wilt δε:/
occurs as well as /wil δε:/.

Additional second person singular forms incorporating the pronoun: /'wiltte, wil(t)t/ 'will thou?'

Present interrogative negative:

(a) /we:nt, we:nt/, throughout. /we:n/ is possible

^{1.} Cf. section 5.4.2.

^{2.} There could be some variation in frequency of occurrence within the area. For Farnworth, Kearsley and Bolton /we:nt/ is preferable. Stronger support for /we:nt/ comes from Little Hulton (and possibly Wigan).

before a pronoun beginning with a vowel.

Additional second person singular forms incorporating the pronoun: /'wetnt, 'witnt/'will thou not?'

(b) /wil/ + SUBJECT + NEGATIVE PARTICLE, throughout.

Preterite:

(a) Forms incorporating the pronoun:

Singular: 1. /aid, æ:d, ad

2. deid, ded

3. i:d, id
 e:(r)d, e(r)d
 it ed

Plural: 1. we:d, wid

2. jo:d, jed

3. de:d, did /

/at/ in /atd/ is realised by monophthongal variants, or variants in which the second element is of low prominence. 1

(b) /wad/, throughout.

Preterite negative:

- (a) PRETERITE FORMS INCORPORATING THE PRONOUN + NEGATIVE PARTICLE, throughout.
- (b) /'wodnt, wont, ('wodn)/, throughout.

Preterite interrogative: /wod/, throughout.

Additional second person singular forms incorporating the pronoun: /'watte, 'wadte, wa(t)t, (wadt)/ 'would thou?'.

^{1.} Cf. section 5.4.2.

Preterite interrogative negative:

(a) /wont, 'wodnt/, throughout. /won/ may be used before a pronoun beginning with a vowel.

Additional second person singular forms incorporating the pronoun: /'wotnt, 'wodnt/ 'would thou not?'

(b) /wod/ + SUBJECT + NEGATIVE PARTICLE,
throughout.

The distribution of will and would is similar to that in S.E. The overlap between will and shall in the dialect has already been mentioned in the description of shall. In modified speech, /we:nt, we:nt/ -> /wo:nt/, and /won/ is not used.

6.8.4.8. Be

Infinitive: /bi:, bi/

Present participle: /'bi:in/

Past participle: /bin/

Present:

(a) Singular: 1. / am

2. æ:(r)t, at

3. 1z

Plural: 1. æ:(r), a

2. æ:(r), a

3. æ:(r), a /

/at/ is often pronounced $[a\Re^{\epsilon}]$. In the third person plural, if the subject is not the pronoun

they, the third person singular form of the verb may be used, e.g. /klo: oz iz/ 'clothes is (are)'.1

(b) Forms incorporating the pronoun:

Singular: 1. / aim, æ:m, am

δε:t, (δε:z), δεt

i:z, (ιz) ; e:(r)z, e(r)z; ιts 3.

1. we:(r), we:(r), we(r) Plural:

2. joə(r), je(r)

3. $\delta e:(r), \delta e:(r), \delta e(r)$ /

It will be remembered that the phoneme /ai/ includes such variants as [ä:], 2 and the phonemicisation /arm/ here implies monophthongal variants of /at/, or variants with a second element of very low prominence. long vowel in /we:(r)/ and /de:(r)/ can break to a diphthong, leaving [we'a] and [be'a] when /r/ is not pronounced. These forms might seem to suggest the need for a phoneme /ea/, but in a sense they may also be felt to transcend the unit word, which is the limit of the present phonology.³ In terms of the present phonology, [we'a] might be viewed as a reduced form of /'we:e/. or as a form of /we:/, both of which are possible phonemicisations if /r/ is not pronounced. The occasional form /δε:z/ may have resulted from a generalisation of the second person singular ending to the verb be. Strictly, /oc:z/ means 'thou has'.

Present negative:

(a) PRESENT FORMS INCORPORATING THE PRONOUN + NEGATIVE PARTICLE, throughout.

Cf. the present tense forms in section 6.8.1.1. See section 5.4.2. for details.

^{2.}

^{3•} As stated in section 5.1.

(b) Singular: 1. 2. 3. /int, 'iznt/
Plural: 1. /\omega:(r)nt
2. \omega:(r)nt
3. \omega:(r)nt /

Present interrogative:

Singular: 1. / am

2. a, at, e

3. lz

Plural: 1. æ:(r), a, e

2. æ:(r), a, e

3. æ:(r), a, e

The plural forms /a/ may be followed by either a stessed or unstressed pronoun. In the second person singular, additional forms, in which the pronoun is incorporated into the verb, are: /'atte, a(t)t/ 'art thou?'. /a(t)t/ is by far the most common second person singular interrogative form. The third person plural form may be the same as the third singular if the pronoun they is not used, e.g.

/iz ðem ty: 'gy:in/
'Is them (Are those) two going?'

Note also the contractions:
/wiə(r)t/ 'where art thou...?'
/wot 'le:(r)nin/ 'what art thou learning?'
/wiə(r)t ðe:/ 'where art thou?'
/ɛ:t 'fiəlin/ 'how art thou feeling?'

^{1.} Cf. the present tense forms in section 6.8.1.1.

Present interrogative negative:

(a) Singular: 1. / æ:(r)mt, æ:(r)nt

2. æ:(r)nt

3. int, in

Plural: 1. æ:(r)nt

2. æ:(r)nt

3. æ:(r)nt /

Additional second person singular forms incorporating the pronoun: /'atnt, æ:(r)nt/'art thou not?'. The form /in/ is used optionally before a vowel.

(b) PRESENT INTERROGATIVE FORM + SUBJECT + NEGATIVE PARTICLE, throughout.

Preterite: /we(r), we:(r)/, throughout.

Modified speech can have /woz, wez/ for all persons, although many people retain /we(r)/ for all persons.

Preterite negative: /we:(r)nt/, throughout.

Modified speech can have /'woznt/ for all persons, although many people retain /we:(r)nt/ for all persons.

Preterite interrogative: /we(r), we:(r)/, throughout.

Additional second person singular forms incorporating the pronoun: /'wette, we(t)t/'were thou?'

Preterite interrogative negative:

(a) /we:(r)nt/, throughout. /we:(r)n/ is possible
before a subject beginning with a vowel.

Additional second person singular forms incorporating the pronoun: /'we:(r)ntte, 'wetnt, we:(r)nt/ 'were thou not?'

(b) /we(r)/ + SUBJECT + NEGATIVE PARTICLE, throughout.

6.8.4.9. <u>Have</u>

Infinitive: /a, av, e/. There is no tendency for only the form /av/ to occur before a vowel. /a/
may lengthen before a vowel.

Present participle: /ain, 'avin/. Only variants of /ai/ which have a second element of low prominence, or monophthongal variants, are used.²

Past participle: /ad/

Present:

- (a) Singular: 1. / av
 - 2. az
 - 3. az
 - Plural: 1. an, av
 - 2. an, av
 - 3. an, av /
- (b) Forms incorporating the pronoun:

Singular: 1. / aiv, e:v, av

- 2. δε:z. δεz
- 3. i:z, (\(\pi_z\))
 e:(r)z, \(\epsilon(r)z\)
 its

Plural: 1. we:n, we:v, win, wiv

- 2. join, joiv, jen, jev
- 3. dein, deiv, din, div

/aiv/ is realised phonetically by monophthongal variants, or variants having a second element of low prominence. The switch to a front first element and a second element of some prominence is a mark of modification.

^{1.} Cf. Taylor (1901: no pagination) on the forms of <u>have</u>. Taylor writes that his work reflects the speech of that area of Lancashire between Bolton and Manchester.

^{2.} Cf. section 5.4.2.

^{3.} Cf. section 5.4.2.

Present negative:

- (a) PRESENT FORMS INCORPORATING THE PRONOUN
 + NEGATIVE PARTICLE, throughout.
- (b) Singular: 1. / 'avnt
 - 2. 'aznt
 - 3. 'aznt

Plural: 1. 'avnt

- 2. 'avnt
- 3. 'avnt /

These forms would constitute modified usage in many contexts, e.g. the dialect uses have, usually + getten (got) + noan/none

rather than <u>haven't</u> + <u>any</u>. Thus <u>SED</u> questions IX, 6, 2-3 will tend to produce answers such as:

/av 'getn no:n/

'I've getten (got) noan (none)'

/av 'neve(r) si:n it/

'I've never seen it'

rather than forms of the type <u>haven't</u>, hasn't. 1

Present interrogative: PRESENT TENSE (a)-FORMS, throughout.

Additional second person singular forms incorporating the pronoun: /'aste, ast/ 'has thou?' /ast/ is often /as/ before a consonant.

Note also the contractions:

/wots bakt/ 'what has thou backed?'

/woes don/ 'what has thou done?'

etc.

^{1.} Cf. section 1.2.1.1.

Present interrogative negative:

- (a) PRESENT TENSE (a)-FORMS + SUBJECT + NEGATIVE PARTICLE, throughout.
- (b) Singular: 1. /ant, 'avnt/
 2. /'as(e)nt/-pronoun incorporated

Plural: 1. /ant

2. ant

3. ant /

Modified forms are /'aznt/ for the third person singular and /'avnt/ for all plural forms.

Preterite:

- (a) /ad, (εd)/, throughout. /εd/ is very residual now. Note /at/ 'had to'.
- (b) Forms incorporating the pronoun: add /d/ to the stressed or unstressed forms of the personal pronoun, le.g. /δε:d, δεd/ 'thou had', as listed for the preterite of will.²

Preterite negative:

- (a) PRETERITE (b)-FORMS + NEGATIVE PARTICLE, throughout.
- (b) /'adnt, (ant)/, throughout. These forms are not always appropriate - cf. the comment on present negative forms above.

Preterite interrogative:

/ad/, throughout.

Additional second person singular forms

Cf. section 6.7.1.1.

^{2.} See section 6.8.4.7.

/'atte, 'adte, a(t)t, (adt)/ 'had thou?'

Preterite interrogative negative:

- (a) /ad/ + SUBJECT + NEGATIVE PARTICLE, throughout.
- (b) /'adnt, ant/, throughout.

Additional second person singular forms incorporating the pronoun: /'adnt, 'atnt/ 'had thou not?'

There are many analytic constructions containing HAVE, e.g. /av e well 'have a wash = wash'. Further: 'have a look', 'have a see' (look), 'have a stop', 'have a bake', 'have a listen', 'have a smell', 'have a taste', 'have a feel', 'have a lie down', and even with more unusual concepts such as 'have a box' (box in a boxing ring), etc.

6.8.4.10. Dare

One of the two verbs <u>dare</u> is anomalous, the form of the verb being the same for all persons when the infinitive which follows (or is implied) is not preceded by <u>to</u> or <u>for to</u>. The informants for this study do not appear to distinguish between present and preterite forms, and attempts to arrive at a more systematic classification by eliciting a wide range of <u>dare</u>-forms were not successful. Complicating factors included:

- rejection by informants of some questions and contexts as implausible;
- 2. confusion and unease on the part of the informants;
- 3. the existence of a range of vowel variants in free variation (but not known for certain to be in free

variation to begin with, of course);

- 4. the existence of a regular verb dare;
- 5. a lack of forms which were clearly preterite in reference due to: the factors already listed; a seeming disinclination to use the preterite as such and a preference for other constructions such as <u>didn't dare</u> and <u>were gam</u> (was game); the use of the historical present in narratives, so that one is not always sure whether a form is meant to be present or past.

The second factor is very understandable if we allow that the linguist may well have been looking for distinctions which did not exist. There is a confusion of forms with this verb, just as there is in the case of S.E.

/dæ:(r)/ occurs widely, and might seem to be essentially a present tense form (although see the discussion of the preterite which follows). /de:(r)/ is slightly modified, and /de:(r)/ decidedly so. /dæ:(r)/ and /de:(r)/ both show a marked tendency for the vowel to shorten, resulting in half-length and short variants. One may therefore also phonemicise /da/, probably without /r/, and /de(r)/. The realisations of /a/ are, however, high.

The forms /dast, dost, dest, (dest)/ are also used as present tense forms. /dast/ and /dest/ can have half-length variants, in addition to short ones (again high in the case of /a/), and might occasionally be phonemicised as /dæ:(r)st/ and /de:(r)st/. The use of all these forms in the present tense is not open to debate, but clear instances

of the preterite are harder to come by. To take just one example, an informant gave:

/i:d e kom v i: dest/
'He'd have come if he durst.'

but the informant was equally happy to substitute /da/ for /dest/. Was /da/ being used as a preterite here or not? It is difficult to decide, but quite possibly it was. It is equally difficult to decide whether /dest/ is a preterite or not when the same form can also be present. Elicitation of preterites is difficult, as suggested in points four and five above. My impression - and it can be no more than that - is that /dest, dost, dast, (dest)/ can function as preterites, but that they are rare as such. The regular verb dare may be used in the preterite, of course. Finally, as in the example above, /dx:(r)/-types may perhaps function as preterites too.

Negative forms without <u>do</u> are common in the present tense. /dm:(r)nt/ is a negative form applicable to all persons, /de:(r)nt/ is slightly modified, and /dc:(r)nt/ decidedly so. The present negative may equally well be rendered by /'das(e)nt, 'des(e)nt, 'dos(e)nt, ('dcs(e)nt)/. /'das(e)nt/ and /'des(e)nt/ may have half-length vowels as well as short ones, and again realisations of /a/ are high. My impression is that negative forms with and without /s/ can on occasion serve as preterites. An alternative past formation is <u>didn't</u> + <u>dare</u>.

Positive forms may be used in the interrogative, and the vowels are often short, especially in tags, e.g.

/'dari:, 'deri:/ 'dare he?'. /dost/ and /dest/ 'does thou dare?' were given as additional second person singular forms incorporating the subject pronoun, and further investigation might reveal others. The negative interrogative may be formed in:

- (a) POSITIVE FORMS + SUBJECT + NEGATIVE PARTICLE
- (b) NEGATIVE FORMS.

/'das(e)nt/ and /'des(e)nt/ were given as additional second person singular forms including the pronoun, and further investigation might reveal others.

Presumably these interrogative forms can have both present and past reference too.

6.8.4.11. Do

Infinitive: /dY:, (do, de)/.

Present participle: /'dr:in/.

Past participle: /don/.

Present:

Singular: 1. / dy:, do

 $2 \cdot d\Omega z$

3. doz

Plural: 1. dy:, do, don

2. dy:, do, don

3. dy:, do, don /

/o/-forms which alternate with /v:/-forms may, as in the case of go, be stressed, but with the verb

^{1.} Cf. section 6.8.1.1.

do the form /do/ is somewhat restricted in distribution. In /al let on no: we a dy:/ 'I'll let thee know if I do' where do stands for another verb and is stressed, the form with the long vowel is required.

Present negative:

Singular: 1. / de:nt

2. dont, 'doznt'

3. dont, 'doznt

Plural: 1. de:nt

2. de:nt

3. de:nt /

Note also /ai do'no:/ 'I don't know'.

/'ai 'dy: 'not 'no:/ 'I do not know' is the only use of DO + not. It is very emphatic.

Present interrogative: Present forms, throughout, except that the first person singular is /dx:/ only.

Additional second person singular forms incorporating the pronoun: /'doste, dos(t)/'does thou?'. The final /t/ of the latter form is quite often not pronounced before a consonant. Unstressed: /des/'does thou?'. Contractions: /wots mi:n/'what does thou mean?'; /wie(r)s get oat fre/'where does thou get that from?'; etc.

Present interrogative negative:

(a) Singular: 1. / de:nt, (de:n)

2. dont

3. dont, don

Plural: 1. de:nt

2. de:nt

3. de:nt /

Forms without /t/ are optional forms occurring before a vowel. Additional second person singular form incorporating the pronoun: /'dos(e)nt/'does thou not?'

(b) PRESENT INTERROGATIVE FORMS + SUBJECT + NEGATIVE PARTICLE, throughout.

Preterite: /did/, throughout.

Preterite negative: /dint, 'didnt/, throughout.
Occasionally /'didn/.

Preterite interrogative: /did/, throughout.

Additional second person singular forms incorporating the pronoun: /'ditte, di(t)t/'did thou?'. Sometimes in running speech the devoicing of the final /d/ in the second person singular form incorporating the pronoun is not complete, e.g. [did? ko:] 'did thou call?'. Also, before a vowel, [did? tire] 'did thou hear?'. /didt/ is therefore perhaps also a possible phonemicisation of 'did thou?'.

Preterite interrogative negative:

(a) /dint, 'didnt/, throughout. The first and third persons singular may also be /din/, or /'didn/, i.e. /t/ can be elided before a vowel.

Additional second person singular forms incorporating the pronoun: /'ditnt, 'didnt/'did thou not?'

(b) /did/ + SUBJECT + NEGATIVE PARTICLE,
 throughout.

6.8.4.12. <u>Go</u>

This verb is only partly anomalous in the dialect - see the forms of \underline{go} in the description of the present tense endings.

Infinitive: /gv:, go/.

Present participle: /'gy:in/.

Past participle: /gon/.

6.8.4.13. <u>Used</u>

Used is followed by \underline{to} + infinitive or \underline{for} to + infinitive.²

Preterite: /jr:s(t)/, throughout.

Preterite negative: /'jx:znt/, throughout. The construction didn't + used is often used.

I have only noted interrogatives formed by using do as an auxiliary.

^{1.} See section 6.8.1.1.

^{2.} For forms of these prepositions see section 6.10.

6.8.5. <u>Imperatives</u>

As in S.E., the imperative is formed without endings. The use of a pronoun - in the object case - is optional, e.g.

/ði: kom tə(r)/
'Thee come here!'

A number of periphrastic constructions are found:

/bi 'gy:in wom/
'Be going whoam (home)!'

/bi 'komin on wom wi oi/
'Be coming on whoam (home) with thee!'

/lats bi 'no:int wost/
'Let's be knowing the worst!'

Similarly in Let's be having thee!; Let's be seeing it!;

Let's be getting done!; etc. Strang has observed that

periphrastic imperatives of the type be going were permitted in Elizabethan English.

Quite a number of imperatives are formed with GET + PAST PARTICIPLE:

/ger em rong/
'Get them rung! = Telephone them!'
/ger em tipt/
'Get them tipped!'
/get or'sel werst/
'Get thyself washed!'
/get 'stæ:(r)tid/
'Get started!'
/get it 'etn/
'Get it eaten!'
/get gon/
'Get gone! = Go!'
etc.

^{1.} See Strang (1974: 150).

Despite the similarity to the passive, which may also be formed with GET + PAST PARTICIPLE, the above constructions are active in force. Thus <u>Get them tipped!</u> means on this occasion 'you tip them', and not 'have them tipped by someone'.

The form of commands in modified speech follows the above closely.

6.8.6. Constructions with GET:

The use of GET as a passive auxiliary is discussed elsewhere. Other constructions also use GET + PAST PARTICIPLE but are active in meaning, e.g.

/gst oi'ssl wsist/
'Get thyself washed!'

In addition to imperatives, 2 note also:

/al gst im 'takn/
'I'll get him taken'

/wen i:d 'getn it 'etn/
'When he'd got it eaten'

and many others. I'll get him taken is, like all these constructions, active in force, and means 'I'll take him', and not 'I'll have him taken by someone else'. Similarly to get washed up, get done, get thee/thyself washed/dressed/shaved, get a thing eaten/drunk/shifted, etc. etc.

GET is followed by an infinitive form in /gst kat it/ 'get catch it'; /gst kap it/ 'get cop (catch) it',

^{1.} See section 6.8.8.

^{2.} For further examples see section 6.8.5.

both of which mean 'be in trouble, be punished'.

A very distinctive construction is

TO /get e'ge:t/ 'get <u>agate</u>' + PRESENT PARTICIPLE which means 'to start doing something, which subsequently becomes habitual, to get into the habit of doing'.

Examples:

/en i: gor e'ge:t dise'plerin/
'and he got agate disappearing'
/gi:t e'ge:t 'ly:kin 'æ:f0e(r)t ge:(r)lz/
'geet (got) agate looking after the girls'
etc.

The expression:

/e(r)z 'belos 'gerin e'ge:t on mi/
'her's (she's) always getting agate on me'
means 'she's always nagging me'. When someone is going
home, one may say:

/all gy: e'ge:t wi ði/
'I'll go agate with thee'

which means that the speaker will accompany the guest part of the way home.

Get has the approximate meaning 'start' in /wen or jy:s get 'fo:in e:t/
'when they used to get falling out'

and the meaning 'start' or 'reach the point of' in we'd get falling asleep sometimes. Get may be used in the sense of 'reach, become' with ages, e.g.

/ez sy:n ez i: gi:t 'siksti/
'as soon as he geet (got) sixty'

GET + PRESENT PARTICIPLE can mean 'get a job doing X', e.g.

/get 'ko:lin/

'get coaling = get a job as a coalface worker'

Get is very extensively used in the dialect, and most of the meanings listed under this verb in dictionaries, including entries marked <u>colloquial</u>, are to be found. In questions, the dialect does not use <u>have</u> alone, if the question is of the type

Have you a + NOUN?

but always uses a construction of the type Have you got a
+ NOUN?

There are a number of constructions containing get, which are not followed by an object, adverb, or prepositional phrase:

/a kæ:nt gɛt/
'I can't get (by/past/through this, etc.)'
/at mon gɛt don/
'I mun (must) get done (i.e. finish whatever it is)'
/www.'gɛtn sot/

'we getten (got, [have] got), shut (rid) (of whatever or whomever it is)'

similarly with:

/get rid/
'get rid (of it/them/etc.)'
/os gi:t jy:st/
'thou geet used (to it)'

Apart from get agate, the above constructions are also to be found in modified speech.

6.8.7. Affirmation and Negation:

6.8.7.1. Affirmative and negative particles:

The system of affirmative and negative particles is more complex than the S.E. equivalent. The less restricted of the two affirmative particles is /ai/ 'aye, yes', which is the general word for yes. The second particle is /jai/ 'yes, oh yes, etc.' and this is used to contradict a negative proposition, or in answer to a negative question. When one person says that she cannot find the scissors, she receives the reply:

/jai ðe(r) ðie(r)/

'Yigh (yes, oh but) they're there.'

Similarly, an informant asked his wife if it had been four shillings that they had received when he had once been out of work. His wife said: "I don't know". The informant replied:

/jai wen mi: en 'tomi 'blaywe(r) went de:n/
'Yigh (yes you do know), when me and Tommy
Blower went down.'

There are other straightforward cases of <u>yigh</u>, but the tapes contain an interesting borderline case of <u>yigh</u> in answer to a "negative" question. An informant was relating a story about a woman in another part of the country, who - to his dismay - could not make a potato pie: "I said: 'Well you can make a pie-crust, can't you?'" There was evidently some doubt about this. He continued, in reported speech:

/o: jai e(r) ked me:k won e ðɛm/

'Oh <u>yigh</u> (yes) her (she) could make one of them.'
Classically, <u>yigh</u> follows a negative question and not a
negative tag. However, the question was almost certainly
couched in positive terms as a matter of politeness, and it
is clear from the story that there was some doubt as to
whether the woman could make a pie-crust. <u>Yigh</u> contradicts
the implication that she could not make a pie-crust.

Yigh may also follow upon a negative in the speaker's own directly preceding utterance, as he changes his mind or corrects himself. When I suggested to an informant that he was asking me to remember something which happened too long ago, he replied:

/m:m not 'gr:in bak o: dat long [LAUGHS]
'I'm not going back all that long!

∭jal 'ttl bl ∞(r) 'twentl je(r) we:n lt/ Yigh, it'll be over twenty years, won't it?'

The use of the particle <u>yigh</u> after negative propositions and questions in contrast to the more general <u>aye</u> is directly comparable with the N.H.G. <u>doch - ja</u> and N.F. <u>si - oui</u> distinctions. Additionally, however, the dialect has two types of negative particle. The general, less restricted type is variously /nay, ne:, ne:/ 'no'. /nay/ is sometimes pronounced with a low first element, and a suggestion of an /r/ might occasionally be suspected after the long vowel variants. In contrast with this type is

^{1.} Cf. Shorrocks (1978c: 12). 2. See section 5.4.4.2(ii).

the particle /ne:/, which means 'no' when contradicting.
When a judge once told a man that he knew more than he would
tell, the man replied:

/a st ne: bat gom || a sez am not
'I say [=said]: "Nay (no) by gum!"..I says: "I'm not
'avin oat/
having that!"'

Again, an informant may use the contradictory particle when changing his mind or correcting himself:

/o: at || ne: it 'didnt/
'oh <u>aye</u> - <u>nay</u> (no) it didn't!'

Although there are difficulties involved in defining the term <u>sentence</u>, especially for the spoken language, it has to be observed that sentence grammars are quite unable to handle these particles. Both <u>yigh</u> and <u>nay</u> operate most typically across sentence or utterance boundaries between two speakers, e.g.

- A. /a koent dy: it/ 'I can't do it.'
- B. /jai & kon/ 'Yigh (Oh yes) you can!'

 Text grammars are more promising because, in a case such as the example just given, yigh is formally predictable from

the example just given, <u>yigh</u> is formally predictable from the negative in the preceding utterance. But there are yet other uses of <u>yigh</u> which are dependent upon the pragmatic situation rather than the preceding text. For instance, an informant wanted to show me a photograph of a horse which he had once owned. His son went to look for the photograph on a nearby piece of furniture, but when the son did not immediately locate the desired photograph, the informant

exclaimed:

/jai it iz its on ðiə(r) 'somwiə(r) 'goə(r)dn/
'Yigh (yes) it is, it's on there somewhere, Gordon.'

In this case, yigh contradicts the failure to find the photograph. It is as if the son had said: "It's not here" or "I cannot find it". Similarly, an informant observed to a child in respect of myself:

One informant sometimes used <u>yigh</u> as a narrative device. For instance, he told a story about tantalizing an old cat with a lump of pluck dangled from a piece of string. The cat jumped up, seized the pluck, would not let go, and lost its teeth in the pluck. Having told the part about the cat losing its teeth, the informant added: "Yigh, he jumped up and it wouldn't let go...". Yigh serves here to counter or pre-empt any incredulity on the part of the listener. Again, it is as if the listener had said, or - more likely - might be thinking "I don't believe you".

From the foregoing cases, it will be clear that yigh can be used to counter implied negation, or aspects of the pragmatic situation. Similarly, use of <u>nay</u> is sometimes dependent on the physical context. A dialect speaker who had noticed that one was executing or contemplating a course of action of which he disapproved, might contradict this

(intended) course of action by saying:

/ne: ne: | ost ne:(r) bs:nt gy:/

'Nay! Nay! (No! No! Surely) thou art never (not)
boun (going) to go!'

Of the two particles <u>yigh</u> and <u>nay</u>, the latter may be heard in modified speech, the former not.

6.8.7.2. Negative adverbs:

The negative is formed by the use of the adverb /no:n/ or /not/ 'not'. Contracted forms of anomalous finites are given elsewhere.

/no:n/ may be used as a pronoun as well as an adverb:

adverb: /am no:n 'gy:in/
'I'm noan (not) going'

pronoun: /i:z 'getn no:n/
'he's getten (got) noan (none)'

Note also the interesting adverbial usage:

/oat 'didnt op'set em no:n/
'that didn't upset them noan (none)'

/'ns:(r), nm:(r), 'nsve(r)/ 'never' can also be used with the meaning 'not' rather than 'never':

/a ne:r i:t no: 'dine(r)/
'I never eat (ate) no dinner.'

The informant is saying here that he did not eat any dinner on a particular occasion after having seen a pit accident. The reference to a single occasion is totally unambiguous.

The adverb <u>neither</u> (/'no: $\delta e(r)$, 'ne: $\delta e(r)$, ('nai $\delta e(r)$, 'ni: $\delta e(r)$ /) is used with <u>nor</u> (/ne(r)/) in the

^{1.} See section 6.8.4. including the restriction on negative contracted forms of the anomalous finites.

construction neither ... nor, e.g.

/its 'ne:ðe(r) won θing ne(r)t 'toðe(r)/
'it's neither one thing nor the tother (other)'

6.8.7.3. Multiple negation:

Treble and quadruple negatives may be heard in the dialect, e.g.

/am not 'neve(r) 'gr:int do navt ne'mve(r)
'I am not never going to do nowt (nothing) no more
fe(r) oi:/
for thee.'

Double negatives, however, are by far the most common, and may be collected literally by the score from children, speakers of residual dialect, and speakers of varieties of Regional Standard alike:

[residual] /www ns:(r) por op ôte(r) ne'mye(r)/
'we never put up there no more (we didn't stay there again).'

[modified] /a jy: nor 'avin non/
'Are you not having none?'

Further examples in translation:

'I'm not never going to do that no more.'
'Don't never bother him - he's eating.'
'We couldn't see <u>nowt</u> (nothing).'
'there's never been nobody there'
'we didn't know <u>nowt</u> (nothing)'
'we wouldn't have gone if we'd known that lot neither'
'she doesn't say <u>nowt</u>'
'he didn't know nothing about it'
'well I've not neither'
'but I don't want to neither'

```
'I said: "Never no more".'
'we couldn't do nowt else'
'he wouldn't put them on neither'
'he wouldn't tell nobody what it was'
'he couldn't get them out no road (any way)'
'not railed round nor nowt'
'there were no chairs ready nor nowt'
'I never saw nowt like it.'
'[?] weren't in the army no great length of time'
'I never said nowt (didn't say anything)'
'I'd never done no fa[rming]'
'thou couldn't have wished for nowt nicer'
'Lee, thou art not having noan (none)!'
'that man wouldn't part with nowt'
'I wasn't so good on it neither.'
'I daren't say nowt.'
'I never thought nowt about that.'
'I can't speak no different.'
'Well Old Parr never had noan only with bloody
greasy legs'
'you've not said nowt yet'
'I didn't know nowt else only the pit'
```

and very many more.

Multiple negation, as linguists have pointed out, whether in respect of dialects or historical varieties of English, is cumulative or reinforcing in effect. It is not to be criticised on grounds of logic and the notion that two negatives make a positive. Yet despite the clearly emphatic function of multiple negation in some contexts, double negation is so common that it is perhaps best described simply in terms of pattern or agreement. The quadruple negative at the beginning of this section is without doubt emphatic, but we couldn't do nowt else is probably no more

emphatic than S.E. we couldn't do anything else.

A further aspect of double negation is the use of a negative question tag after a negative statement, as described in the next subsection.

6.8.7.4. Question tags:

Lester, looking at English from a transformational point of view, gave examples of the use of question tags:

- 1. It is hot isn't it?
- 2. It isn't hot is it?
- 3. It rained didn't it?
- 4. It didn't rain did it?

and commented: "In short, given the main sentence we can always predict exactly what the tag will be." In the dialect this is not quite the case, as there are two types of negative tag, one of which follows a positive statement and the other - optionally - a negative statement. In cases 1 and 3, a negative tag is used, the tag taking the form:

/int, 'iznt, in/ + /it/ 'isn't it?'
/dint, 'didnt, din/ + /it/ 'didn't it?'

i.e. a negative contracted form of the anomalous finite.²
In cases 2 and 4 (2 being quite likely <u>It's not hot</u>, although <u>It isn't hot</u> is possible too), the tag may be positive, as in S.E., or negative. If negative, however, the form must be:

AUXILIARY + SUBJECT PRONOUN + /not' 'not'

^{1.} Lester (1971: 164). The examples are from p. 161.

See further section 6.8.4.

i.e. did it not?, is it not?. Such negative tags recall the dialect's use of multiple negation. The two types of negative tag are never confused. In the dialect, then, we cannot "predict exactly what the tag will be" after a negative statement, but rather that it will take one of two forms.

6.8.7.5. Other negative patterns:

Nor occurs after neither, as described above. It also patterns after other negatives:

> /'wadnt ste(r) wan roid ne(r)t 'tabe(r)/ 'wouldn't stir one road (way) nor the tother (other)' etc.

although or is also met after negatives other than neither. The dialect often uses negative indefinites:

/www.an.no:n/

'we have none'

rather than NEGATIVE VERB + POSITIVE INDEFINITE. Negative indefinites occur after negative verbs too, as may be seen from the examples of multiple negation, above.

Negative forms of anomalous finites, including second person singular interrogative negative forms, are given elsewhere.2

A negative verb may be used with the adverb /'m:(r)dli/ 'hardly', in contrast to S.E. usage:

Cf. section 6.7.7.

See section 6.8.4. 2.

```
/av 'never 'æ:(r)dli don it/
'I've never hardly done it = I've hardly ever done it.'

/e(r) 'didnt 'æ:(r)dli want it/
'her (she) didn't hardly want it = she hardly wanted it.'

/a 'kodnt 'æ:(r)dli ɛit/
'I couldn't hardly eat = I could hardly eat.'
etc.
```

6.8.8. Passive:

The passive is formed in the dialect with:

GET + PAST PARTICIPLE

or

BE + PAST PARTICIPLE

The agent is preceded by the preposition with or by. GET functions as the auxiliary verb more often than in S.E., where it is "avoided in formal style" and "is usually restricted to constructions without an expressed animate agent". There is usually no difference in meaning caused by selecting BE or GET: /we(r) kilt/ and /gi:t kilt/ both mean 'was killed'. However, in one instance a distinction must be made:

/at we(r) orapt an/
'I were dropped on = I was surprised, astonished'
/at gi:t orapt an/
'I geet (got) dropped on = I was caught (e.g. by the police)'

Further, GET may be used in the dialect with an animate

^{1.} Quirk, Greenbaum, Leech and Svartvik (1976: 802).

agent expressed:

/gi:t t3Yt wit 'tsittse(r)z/
'geet (got=was) taught with (by) the teachers'
although there are many cases in which there is no agent
expressed:

/ðat 'didnt get pot de:n did it/
'That didn't get put down, did it?' [of a horse]
/a gi:t sent | 'fe:(r)ðer int pit/
'I geet (got) sent further into the pit.'
etc.

GET is followed by a past participle in other constructions, which therefore resemble the passive in form, but which do not have an agent. 1

The preposition which introduces the agent is either with or by. With may be followed by an animate or inanimate agent, as may by:

/wi e dog/ 'with (by) a dog'
/wi e kæ:(r)/ 'with (by) a car'
etc.

A distinctive passive construction is:

BE + /ko:d/ + PRESENT PARTICIPLE to be called doing something

which means 'to be supposed to be doing something'.

/i:z ko:d 'komin/
'he's called coming'

/we: we(r) ko:d 'sydint 'donki/
'we were called holding the donkey'
etc.

^{1.} See section 6.8.6. for details.

I recorded just one case of the question:

/at 'redi fe(r) 'somet 'getin/

'Art thou ready for something getting?'

The informant confirmed that this meant 'Are you ready for something (food) to be got for you?' and that one might add /oi/ 'thee' after getting, although this was not necessary. Contrast: Art thou ready for getting summat?, which would be an active sentence.

HAVE is in free variation with GET in get my hair cut and get my car mended.

The above principles of passive formation are all carried over into modified speech.

6.9. Adverbs:

Negative adverbs are treated in the discussion of $\underline{\text{Affirmation and Negation.}}^1$

6.9.1. Form:

A great many adverbs in the dialect have the same form as the adjective. Thus:

/a tard or 'konfr'densol/
'I told thee confidential (confidentially)'
/ar 'teknikl daob/
'[a] high (highly) technical job'
and so on.

The use of adverbs with the same form as their equivalent adjectives is reminiscent of N.H.G., and contrasts with the greater reliance on -ly endings in S.E. However, there is a certain amount of variation between zero and -ly endings in the dialect, and some adverbs only exist with an -ly ending, e.g.:

/av not sin im 'le:tli/
'I've not seen him lately.'

The use of adverbs which have the same form as the adjective is further illustrated by the fact that good functions as an adverb in addition to /wiel/ 'well':

/e god pe:d d3ob/ 'a good (well) paid job'
/dy: it god/ 'do it good'

^{1.} See section 6.8.7.

```
/e 'veri god .venti'le:tid Sm:ft/
'a very good ventilated shaft'
/e god kot kyet/
'a good cut coat'
etc.
```

6.9.2. Comparison of Adverbs:

The absence of -ly endings often permits the use of an inflection rather than a periphrastic construction:

> /ois ie(r)z 'naise(r) bilt/ 'this here's nicer (more nicely) built'

The prepositions or conjunctions used in the comparison of adverbs are those given for the comparison of adjectives. For instance:

> /'fe:(r)de(r) de:n til dat/ 'further down till (than) that'

Note also:

```
/'o:fnere.not/
                          'oftener than not'
  /'bst0eren/
                          'better than'
```

Irregular comparatives and superlatives are given under the Comparison of Adjectives. 2 To these may be added:

```
/fm:(r)/ - /'fe:(r)\deltae(r)/ - /'fe:(r)\deltaist/
 'far'
                  'further'
                                      'furthest'
```

See section 6.4.1. See section 6.4.1.

^{2.}

```
6.9.3. "Here", "There", "Yon", "Yonder":
         /lə(r)/
                          'here'
         /ðiə(r)/
                          'there' (more modified /oe:(r)/)
         /jon/
                          'yonder, there'
         /'jonde(r), ('jonde(r), 'jonde(r), 'jonde(r))/
'yonder'
In adverbial phrases of place, one of the above is frequently
added after the place.1
         /et | 'wogdin ðiə(r)/
         'at Walkden there'
         /et 'tlifn ðiə(r)/
         'at Clifton there'
         /et bak et 'le:be(r) klob ðie(r)/
         'at the back of (behind) the Labour Club there'
        /et te:(r) | 'sando:1 oue(r)/
         'at the er .. Sandhole there' [a pit]
        /loa(r)d seri:t 'jonde(r)/
         'Lord Street yonder'
        /et 'wogdin 'jonde(r)/
         'at Walkden yonder'
        /'km:(r)zli 'jənðe(r)/
         'Kearsley yonder'
        /δε:l si: ım 'εντι de: ın 'mæ:(r)kıt sθri:t
         'thou will see him every day in Market Street
        jon wit klagz on/
        yon (yonder, there) with the clogs (clogs, his clogs) on.
        /et 'swith jon/
        'at Swinton yon'
```

^{1.} Cf. the use of the adverbs <u>here</u> and <u>there</u> with the demonstrative adjective and demonstrative pronoun, section 6.4.3. and section 6.7.4. respectively.

/wen & we(r) 'me:kin & is 'mo:te(r) we: ler |

'when they were making this motorway here

et | blak be(r)s 'jonde(r) | 'ke:(r)zli & le(r) /

at the Black Horse yonder .. Kearsley there,...

Here is used to indicate the place where the speaker already is, or somewhere very near at hand indeed. Anywhere further away than here is there, you or yonder. I cannot detect the slightest distinction between the latter three.

6.9.4. The "Known Referent":

The reference associated with here is probably always obvious, but there, you and yonder can refer to a known referent, which may be obvious to intimates, but is not to strangers. One dialect speaker, when visiting anyone, would often refer to /ds:n 'jonðe(r)/'down yonder'. This was an allusion to his own home. But yonder can equally well refer to a launderette (as used by one lady with highly modified speech), or anywhere else for that matter. The linguistic context will sometimes be helpful, but not always.

6.9.5. <u>Intensifiers or Degree Adverbs</u>:

Quite a number of adverbs function as degree adverbs or intensifiers. Some are highly restricted in their distribution, perhaps modifying only a single word:

^{1.} Exact distances and pointing range would not appear to be relevant features.

^{2.} Cf. the demonstrative adjective, demonstrative pronoun and personal pronoun - sections 6.4.3., 6.7.4. and 6.7.1.4. respectively.

```
/'bliderin kard/ 'blithering cold'
/'dedli 'perznes/ 'deadly poisonous'
etc.
```

Others modify or intensify a range of words, but often from the same semantic field:

```
/'Oompin great/ 'thumping great'
/'Oompin big/ 'thumping big'
```

Similarly, thundering, tremendous and whopping.

and

etc.

/'teribl, 'terible'
/'shocking'

are restricted to phrases such as terrible bad, terrible hard, shocking bad, shocking awful.

The following have the meaning 'very, really, extremely, etc.':

```
'all'
/2:/
                'dead'
/ded/
/'gre:dli/
                gradely = 'very, proper, excellent,
                           real, good, genuine'
                'great'
/greit/
                'jolly'
/'dʒəli/
/'prop(p)e(r)/ 'proper'
/'ratlin/
                'rattling'
                'real'
/riel/
/ri:t, (reit)/ 'right'
/'veri/
                'very'
/'martr/
                'mighty'
/'blodi/
                'bloody'
```

Certain of these - proper, real, right, and, with some informants, bloody - are very widely used. /mots/ 'much'

and /e lot, o:l lot/ 'a lot, whole lot' may appear before comparatives. /'fe:(r)li/ 'fairly' and /'priti/ 'pretty, fairly' are also widely used and have the meaning 'fairly, rather', and 'very' if one allows for a certain predilection for understatement. leter(r)/ 'fair' is also an intensifier. A few examples will suffice:

/oats bin 'gre:dli god/
'that's been gradely (very) good'

/'priti nie(r)/
'pretty, fairly, quite near'

/e 'ratlin god dy:/
'a rattling (very) good do (party, occasion, meal, event, etc.)'

/i:z 'bi:in e fe:(r) long taim e'bs:t it/
'he's being a fair (considerably) long time about it'

/oi fe:(r) went ery: it/
'they fair went through it = spent it very quickly'
etc.

Due to the formation of adverbs without -ly, the constructions:

ADJECTIVE + ADJECTIVE + NOUN

and

ADVERB + ADJECTIVE + NOUN

cannot always be distinguished, as in a tremendous big clock.

Three adverbial constructions which add emphasis are included in the next subsection.

Pairly good or not bad can represent a high degree of praise, just as a good meal might be relished with /av 'te:stid wos ('moni e taim)/ 'I've tasted worse (many a time)'.

6.9.6. Some Other Adverbs and Adverbial Constructions:

/'o:los/ [various other forms] 'always'; /'apn, 'apm/
'happen = perhaps'; /'re:ðe(r)/ 'rather'; /'o:nt/ 'only';
/'bake(r)ts/ 'backward'; /'fore(r)d/ 'forward(s)'; /bot/
'but, only'; /fe(r) 'regle/ 'for regular = regularly';
/fe(r) [ve(r)/ 'for sure = certainly'; /fe(r) 'settin/
'for certain = certainly'; /(t)moe(r)n/ 'tomorrow';
/te'moe(r)n/ 'tomorrow'; /(t)ni:t/ 'tonight'; /te'ni:t/
'tonight'; /'nobot/ 'only'; /'pæ:(r)tli wot/ 'partly-what
= in part, partly'; /'saide(r)ts/ 'sideways'; /ðier e'gen/
'there again = then again'; /'de:nbrv:/ 'downhill';
/'opbrv:/ 'uphill'; /'opraits/ 'upright'; /wom/ 'home';
/sin/ 'since'; /wonst/ 'once'.

Negative adverbs and the phenomenon of multiple negation are treated elsewhere. Words for <u>yes</u> and <u>no</u> are treated as particles elsewhere, rather than as adverbs. ²

Three common constructions which lend emphasis are the following:

- 1. SUBJECT + NEGATIVE AUXILIARY + half and + VERB or SUBJECT + BE NEGATIVE + half and + ADJECTIVE³
- 2. /'boni wiel on/ 'bonny well and' + PAST PARTICIPLE³
- 3. <u>about</u>, e.g. /i:z e'bɛ:t kılt ıt/ 'he's about killed it = "he will have killed it, that's what he'll have done" [informant's paraphrase]'

^{1.} See sections 6.8.7.2-3.

^{2.} See section 6.8.7.1.

^{3.} See section 6.11.

As may function as a relative adverb meaning 'when'. e.g.: /oat taim ez a we(r) bad/ 'that time as (when) I were bad (was ill)' That often has the meaning 'so', e.g.: 'so big'. /dat big/ The construction that... as corresponds to S.E. so... that, e.g. /ot gi:t oar syd ez ot 'adnt 'konfidons i oe(r)'sal/ 'they geet (got) that (so) old as (that) they hadn't confidence in theirself (themselves) The as element is not always present in such constructions. That, and all, and that and all frequently serve as intensifying tags. Emphasis is often given to an account of an event by adding: /i: did dat/ /i: did en p:/ /i: did dat en o:/ all of which mean 'he did too!' or 'he did indeed!' So has the meaning 'very' in constructions of the type: /its not so red of in ie(r)/ 'it's not so red-hot in here' This is an understatement meaning that it is decidedly cold. The interrogative adverb why is sometimes for why, e.g.

'I'll tell thee for why (I'll tell you why)'

/al tel oi fe(r) wai/

In addition to /'ds:n'ri:t/ 'downright',
/'ri:t'ds:n/ is also regular, e.g.

/am 'ri:t'de:n vekst/
'I'm downright vexed.'

To get faster forward, an idiom meaning 'to make progress, advance more rapidly', is also syntactically distinctive, e.g.

/δεd get 'fæ:sθe(r) 'fored iv δε py:d δι
'thou would get faster forward if thou pulled thy
'dʒakit ɔ:f/
jacket off'

About may modify constructions of the type:

INDEFINITE ARTICLE + MEASUREMENT + NOUN

by immediately preceding them. Examples:

/or ad e'be:t e Ori: wik sOraik/
'they had about a three-week strike (a strike of about three weeks' duration)'

/ko:l | komz tet 'botom et pit | on belts | 'coal comes to the bottom of the pit on belts -

'apn e'bsyt | e foe(r) fot belt/
happen (perhaps) about a four-foot belt (a belt
of about four feet)'

6.10. Prepositions:

A reasonably exhaustive specification of the prepositions of the dialect would be dependent upon a fuller investigation of the lexis than can be carried out here. This is because:

- 1. prepositions occur in idiomatic phrases;
- 2. prepositions occur after verbs, and many such uses are distinctive.

Examples of 1:

/i:z o: oat ouer e'be:t im/
'He's all that there about him = he is self-important.'

/i:z no:n long for lo(r)/
'he's noan (not) long for here = he will soon die.'

Examples of 2:

/we:t on, ov/
'wait on, of = wait for'

/al Oreis em e:t fo(r) wek/

'I'll <u>trash</u> them <u>out</u> for work = I'll wear them out at work (- they are good enough for that, but not for anything else)'

See also off below.

Nor, till, tin and than are used prepositionally in the comparison of adjectives and adverbs. Than is sometimes called a quasi-preposition in accounts of S.E. In the dialect, the prepositions must be followed by the object case. Sentences of the type He is bigger than I are impossible. There is a straight choice in the dialect between the patterns He is bigger than I am/than what I am (conjunction) and He is bigger than me (preposition).

The dialect often does not use a preposition with adverbial phrases of time:

```
/õεm de:z/
'them days = in those days'
/δε ked gy: ont mo:s | 'some(r) taim |
thou could go on the moss - summer-time -
en & ked av e nais de:/
and thou could have a nice day'
/'win0e(r) taim/
'winter-time = in winter-time'
/t taim/
'the time = at the time'
/itl not bi mai de:/
'it'll not be my day = in my day'
/'krismos/
'Christmas = at Christmas'
/δε 'stæ:(r)tid e:f pas faiv/
'thou started half-past five'
```

and very many others. Cf., however, the use of at below.

The following list of prepositions is partial, and concentrates on distinctiveness <u>vis-à-vis</u> S.E. Within the list itself, the forms and meanings given are those which are comparatively distinctive.

a: /e/ 'a'. Used with verbal nouns and participles,

e.g. /'takint θρθ(r)s e'bi:in [Y:d/
'taking the horse a-being shoed (shod)'

Such constructions occur most commonly of all after the verb GO, with the meaning 'to, in order to, -ing'. Further material in translation:

```
'they used to have these outlooks [look-outs], you know, a-waiting of...'

'went a-complaining'

'they came a-keeping a pub'

'and the lad ran home a-telling his mother'

'they came a-looking for me'

'go a-working'

'we only used to go a-gathering these nettles'

'they go a-playing at cricket'

'men going to these rucks (coal waste heaps), a-grading it'

'aye, we used to go a-fishing'

'oh, from there, I went to the Drake [mill], a-winding at the Drake'

etc. etc.
```

Note also: /lev im e'bi:/ 'leave him a-be (leave him alone, leave him be)'.

above: /e'bov/ is used quite extensively with the meaning 'for longer than', e.g.

/it re:nd e'bov e wik/
'it rained above a week'

according to: /e'kbə(r)din te/ 'compared with':

/it ed bi 'priti &i ste:ks | e'kbe(r)din
'it would be pretty high stakes..according
te te'de:z | 'b&tin/
to today's.. betting'

afore: $/e^{i}fve(r)$, $e^{i}foe(r)/'before'$. This preposition is used extensively, e.g.

/δε jy:s gy: en ε:r e'foə(r)t 'toŏe:(r)z/
'thou used to go an hour afore the tothers (others)'

after: /'m:f θ e(r)/ 'after, in pursuit of, in quest of'.

My impression is that BE + after 'be in pursuit

of' is used more extensively than in S.E. Examples:

/t dogz 'm:f0e(r)t 'bcielt am/
'the dog's after the boiled ham' (i.e. is trying to get at it)

/am 'æ:f0er im/

'I'm after him.' (either chasing him here and now, or intending to catch him at the first opportunity)

/we: wer 'm:f0e(r) som e om/
'we were after some of them (those)' (i.e. wanted to acquire some)

again: /e'gɛn, e'dʒɛn/

etc.

1. 'near, just by, next to, by the side of', e.g.

/i: livz e'gent ore:k/
'He lives again the Drake.' (i.e. in a street next to the Drake Mill)

2. 'compared to', e.g.

/a kom bak te:r e bit e ri:'aliti e'gen te'de:/
'I come back to a bit of reality again today.'

3. S.E. against

In case 2, <u>again</u> might appear to mean 'once more' as written here. However, the intonation pattern of the utterance rules out such an interpretation, and the meaning 'compared to' is assured. Cf. <u>over again</u>.

among: /e'mong/ 'among'. Note, however, the idioms:

/mok e'mong it/ 'muck among it'

/mes e'mong it/ 'mess among it'

both of which mean 'make do, get by in the circumstances (which are mucky or messy)' or 'be in a mess'.

```
as from: /az from, etc./ 'from', e.g.
         /az fromp mil a went de:nt pit/
         'as from the mill I went down the pit'
         (i.e. after being a textile worker I
         became a miner)
at: /et/.
         When a preposition is used with adverbial phrases
of time, it is often at. In this respect, at is equivalent
to S.E. 'at, on, in (the), during (the)' or no preposition
at all.
        Examples:
         /fe(r)t got bed et 'i:vnin/
         'for to go to bed at evening'
         /ot dar (m:f0e'ny:n/
         'at that afternoon'
         /et fas 0ing ev e 'moe(r)nin/
         'at the first thing of a morning'
         /wits we(r) don et 'sondi/
         'which were done at (the) Sunday' (every Sunday)
         /i:z was et dæ:(r)k/
         'he's worse at dark' ('in the, after')
        /wits ty:k e'be:t Ori: e:(r)z et 10æ:f0e'ny:n/
         'which took about three hours at the afternoon'
        /et le:t on/
         'at late on'
        /ty: e'klok et 'moə(r)nın/
         'two o'clock at morning'
        /et day:n/
         'at June'
        /et'fraiddi/
         'at Friday'
```

and very many others.

```
A use of at other than in a time phrase is:
          /a we(r) siks et front on im/
          'I were six at the front on him (I was six points in front of him)'
away: /e'we:/ 'away from'. Examples:
          /tr: mail e'we: 'we:(r)dli/
          'two mile (miles) away Wardley'
         /e'foo(r) os gi:t e'we:t pit le:n/
          'afore (before) thou geet (got) away the pit lane'
at t' back on, of: /et bak on, ev, e/ 'behind'. Examples:
         /et bak on je/
          'at the back on you'
         /'me:(r)t[in re:nd et bak e won e'noðe(r)/
          'marching round at the back o' one another'
         /bak e der ielz o:t taim/
          '[at the] back o' their heels all the time'
at t' back side on, of: /et bak said on, ev, e/ 'behind'.
Example:
         /oi jy:s stak it op et won said | en tsokt
          'they used to stack it up at one side .. and chuck the
         smo:1 | et bak said ev it/
         small..at the back side of it'
bar: /bæ:(r)/ 'bar, except'.1
barrin': /'bæ:rin/ 'barring, excepting, unless there are,
```

except. not including'.1

^{1.} Marked as colloquial S.E. in Hornby (1975: 62).

bar for: /be:(r) fe(r)/ 'except, excepting, apart from'.
Example:

/be:(r) fe(r)t det/
'bar for the dirt'

beawt: /bs:t/ 'without'. Example: /bs:t 'anz 'Soge(r)/

'without any sugar'

belonging: /bi'longin/ 'belonging to'. Example:
 /oats bi'longin oz/
 'that's belonging us'

beside: /bi'said/ 'in addition to'.
The use is the same as that of S.E. besides.

between and: /bi'twi:n en/ 'before, by'.

This is a common preposition. Examples:
 /bi'twi:n en nain e'klok/
 'between and nine o'clock'

/wal bi'twi:n on oat taim || i:d 'getn/
 'well between and that time..he'd getten (got)...'

'cross: /kro:s/ is an aphetic form of across.

down: /de:n/ 'down to, down in'.

The preposition does not necessarily relate to the geographical or topographical position of the speaker. Examples:

```
/i: weks dε:n 'mantsisθe(r)/
          'he works down Manchester'
         /am 'gy:in de:nt myə(r)/
          'I'm going down the Moor' (down Farnworth)
In the second sentence, /t/ does not represent to the -
there is no alternative construction with /tet/ 'to the'.
except for: \langle \epsilon k' s \epsilon p f e(r) \rangle 'except, excepting', e.g.
         /Ek'sep for im/
          'except for him'
for: fe(r)
         1.
            S.E. for;
             'from', as fro and from below.
         2.
        /fot, fe(r)t/ 'for to, to, in order to'.
         This preposition is used extensively in the dialect
before infinitives.
                       Examples:
         /wi we(r) dlad fe(r)t ger e:t/
         'we were glad for to get out'
         /am ri:t fe:n fe(r)t si: je/
         'I'm right fain (very glad) for to see you'
         /it as fot bi:/
         'it has for to be'
         /i: wants fe(r)t gy: | δε no:z/
         'he wants for to go..thou knows'
         /oat ment fo(r)t si/
         'that meant for to say'
         /av got fot go/
         'I've got for to go'
```

and very many others.

Similarly: /jY:s fe(r)t/ 'used for to'.

'fore: /fvə(r), foə(r)/ are aphetic forms of afore, or possibly before. Presumably either is possible, cf.
'cause from acause or because.'

fro: /fro, fre/ 'from', e.g.
/komz fre 'baytn/
'comes from Bolton'

from: /from, from, from/ 'from', is interchangeable
with fro.

at t' front on, of: /et front on, ev, e/ 'in front of'.
Examples:

/et front on im/
'at the front on him'
/e me:s ron et front ev mi/
'a mouse run (ran) at the front of me'

'gain: /gen/ is an aphetic form of again.

i', in: /1,1n/ 'in, into; (to; during; at)'.

The form with /n/ is compulsory before a vowel,
and optional before a consonant. Examples:

/'pr::n 'sombdl l 'pi:slz/
'pulling somebody in (to) pieces'
/ol Sod stop o: oi:z 'alriSimen 'komin it 'konθri/
'they should stop all these Irishmen coming in the country'

^{1.} See section 6.11.

```
/a bompt in im/
         'I bumped in (into) him.'
         /i: livd it top e 'Orafe(r)d sOri:t Oio(r)/
         'he lived in (at) the top of Trafford Street there'
         /ast ad e god 'dge:(r)ni it de:/
         'Has thou had a good journey in the day?'
The form /in/ is used when in is an adverbial particle:
         /wi gor em in bit nait/
         'we got them in by the night'
         It was noted above that prepositions such as in
are not always required in adverbial phrases of time.
However, in phrases such as in a morning, meaning 'every
morning', in is required:
         /we: jy:s tak em a:t in e 'moe(r)nin fer e wo:k/
         'we used to take them out in a morning for a walk'
         /or kamt fas ting in e 'mae(r)nin/
         'they come (came) the first thing in a morning'
```

i'stead o': /i'sted e/ 'instead of', is used as in S.E.

nobbut: /'nobot/ 'except, apart from', e.g.
/wer o: 'gr:in 'nobot e fjr:/
'we're all going nobbut a few'

<u>nor</u>: /ne(r)/ 'than', is used in the comparison of adjectives and adverbs, e.g.

/i: we(r) 'jonge(r) ner e(r)/
'he were (was) younger nor her'

o'er: /oə(r)/

'over, above', e.g.
/oə(r)t sink/

'over the sink'

2. 'about, concerning, on the subject of'
There are similar uses in S.E., but they are neither so
frequent nor so wide-ranging, e.g.

/lark i: tard or co(r)t 'darrn 'pro:ses/
'like he told thee over the dyeing process'

The preposition is used after various verbs, such as grumble o'er, enquire o'er, swank o'er. In the phrase

/am 'gr:in oə(r)t ro:d/
'I'm going over the road',

over the road may represent a known referent, e.g. some particular shop.

o'er again: /oər e'gen/ 'over against, in opposite situation to, in contrast with'.

This preposition is used quite a lot in the dialect:

/'tsekin cer e'gen nem/

'checking over again them' (keeping score in addition to the contestants, who were also scoring)

/e(r) ple:d oer e'gen im/

'her (she) played over again him' (i.e. played a corresponding part: male lead-female lead, husband-wife, or similar)

^{1.} Cf. section 6.4.3, and also sections 6.7.4, 6.7.1.4 and 6.9.4.

```
of, o': /ov, ov, ev, e/ 'of; (at; on; in; each, every)'
```

1. Used in the pattern

MEASUREMENT/NUMBER + of + INDEFINITE ARTICLE + NOUN Examples:

/'twenti'eit jæ:(r)dz ev e 'setin ev e lain/
'twenty-eight yards of a setting of a line'

/won men ed oro: e 'sekkel | 'Eitti:n 'intsiz 'one man would draw a circle - eighteen inches

er e 'kopel e fi:t ev e 'sekkel/ or a couple of feet of a circle'

/foo(r) fat siks ev e orm:ft/

'four foot six of a draught' (i.e. boats with a draught of 4'6")

/'e:ðe(r) said e'bayt | siks fot ev e pak 'either side about..six foot of a pack '

'e:ðe(r) said/ either side'

[modified speech]: /ar woz won ev narn'ti:n ev e 'famlr/
'I was one of nineteen of a family'

2. The pattern

NOUN + of + INDEFINITE ARTICLE + NOUN enjoys considerable use:

/it we(r) laik e mc:0 || ev e spe:d/
'it were like a mouth..of a spade' (i.e. the spade was mouth-shaped)

/it we(r) no: 'domi ev e d3ob/
'it were no dummy of a job' (i.e. the job was not easy)

/e 'rabis ev e fe:(r)m/

3. = 'at, on, in, each, every' in expressions of time (habitual). The same basic construction exists in S.E., but is not as frequent there as in the dialect.

Examples:

/ev e 'wikkend/
'of a week-end' (i.e. at week-ends)

/oe jv:st af get op et fver ev e 'mpe(r)nin/
'thou used to have to get up at four of a morning'
(i.e. every morning)

Of is not always used after plenty:

/oe(r) we(r) 'plenti syd 'brokn plant pots
'there were plenty old broken plant pots

ot flve(r)/
on the floor'

off: /o:f, (of)/ 'off'.

Can be used after the verb reckon, e.g.
/a de:nt 'rekn navt a:f e(r)/

'I don't reckon <u>nowt</u> (nothing) off her (I don't think much of her)'

o', on: /on, o, en, e/ 'on, onto; of'.

The forms with /n/ are compulsory before a vowel, and optional before a consonant. The form /2n/, like /1n/, is used when on is an adverbial particle.

1. 'of':

/al av 'e:ðer on em/
'I'll have either on them'
/t θe:v on it/
'the half on it'
etc.

```
2. on + a + day = on + day PLURAL
/on e 'sondi/
'on a Sunday' (i.e. on Sundays, every Sunday)
```

3. One goes /st we:ks/ or /st fe:r/ 'on the wakes, on the fair' in dialect, rather than to it. A preposition such as on may not be required, however, in expressions of time:

/oi pel be:k 'diferent de:z/
'they all bake different days'

only: /'o:ni, ('o:nli)/ 'but, apart from', e.g.

/oe(r) jr:s bi nsrt 'o:ni 'blemin fieldz/

'there used to be nowt (nothing) only blooming fields'
(the reference is not floral!)

only for: /'o:ni fe(r)/ 'but for, without'

/a de:nt 0ink a Sad e no:d im 'o:ni fer 'art/

'I don't think I should have knowed (known) him only for Harry'

<u>out</u>: /ε:t/

1. 'out of'. Examples:

/get e:t ro:d/
'get out the road (way)'
/ki:p e:t ro:d/
'keep out the road (way)'

2. (In the form out of) 'from' (cf. N.H.G. aus).

Examples:

/e tjr:n 0:ng n3r ez wi gi:t c:t e itand3e'nie(r)z/
'a tune thing now as (that) we geet (got) out of
Tangeniers'2

^{1.} The reference is to a primitive musical instrument.

^{2.} Presumably Tanganyika or Tangier.

/i: kom &:t e 'kampbol s@ri:t/
'he come (came) out of Campbell Street'

over: /'o've(r)/ is a modified form of o'er.

at t' side of, o': /et said ev, e/

- l. 'by the side of'. Example:
 /gι em won et said et 'θιero:l/
 'give them one at the side of the earhole'
- compared with'. Example:
 /am no:n se bad et said e som fo:ks/
 'I'm noan (not) so bad at the side of some folks'

sin: /sin/ 'since' has the same range of use as the S.E. equivalent since.

'stead o': /sted e/ is an aphetic form of i'stead o'.

than: /ðen/ 'than' is used in the comparison of adjectives and adverbs. There is an apparent contraction in: /'bɛtθeren/ 'better than'

till: /til/ 'than' is used in the comparison of adjectives and adverbs, e.g.

/'fe:(r)ðe(r) ds:n til ðat/
'further down till that'

tin: /tin/ 'than' is used in the comparison of adjectives and adverbs, e.g.

/'wimin we(r) wes tin men dem de:z/
'women were worse tin men them days (in those days)'
Tin also has the meaning 'until'.

to: 'to'.

When to has the form of a separate word, it is pronounced /te/, or occasionally /tr:/. However, it is often realised as glottalisation or devoicing at the end of the preceding word, very much as is the definite article. Glottal constriction may operate on the last consonant of the preceding word:

[ä went? bak? 'wo'eweks]
'I went back to Wallwork's'

Devoicing of a final voiced consonant is apparent in:

[at? kut?] 'had to cut'

[wi qf go] 'we have to go'

[wist af bi]

'we shall have to be'

Again as in the case of the definite article, the phonetic context will determine the nature of the suspended consonant:

[a:m gamp? gu']
'I am game to go'

Before a vowel, to may be realised as /t/:

[Jont? të: Jë:s]

'run to our house'

[t'im]

'to him'

although /ty:/ or /te/ is also possible before a vowel,

^{1.} Cf. section 6.2.1.1.

in which case a linking glide [w] will probably be heard between the two vowels. The sequence to the is realised either by the same glottal constriction which represents to alone, or by /tet/.

The sequence going to is sometimes realised as ['go:ne].

To is sometimes omitted in constructions of the type:

[wot? skm: dəz ðs: gm]
'What school does thou go [to]?'

To is optional after the verb help:1

/Elp em kil/
'help them kill'

To is used after might as well:

/δε mεt tez wielt gy:/
'thou might as well to go'

/os met ez wielt to 'wontid/
'thou might as well to have wanted...'

To appears to be fossilised in /latstav/ 'let us [or me] to have', e.g.

/latstav e ly:k/
'let's to have a look'

To was also noted in a number of other constructions and phrases:

/www did six days to the week them days (in those days)'

/te mai e'pinjen/
'to (in) my opinion'

^{1.} As in N.H.G., U.S. English and to a lesser extent S.E.

```
/te mai 'Oinkin/
           'to my thinking' = more usual:
          /te mai ro:d e 'Oinkin/ 'to my road (way) of thinking'
          /i:z e 'boge(r) te 'gambel/
          'he's a bugger to gamble (for gambling)'
          /fo:t sli:p/
          'fall to sleep (asleep)'
          To enjoys frequent usage in the sense 'compared
with', e.g.
          /\eth e(r) myə(r) pits re:n 'wigin te wət
          'There [were] more pits round Wigan to what
          \delta e(r) we(r) re:nd ie(r)/
          there were round here'
          To is also used in the sense of within, below.
             /tɔə(r)d(z)/
/tɔə(r)t/
/te'wɔə(r)d(z)/
tord(s):
tort:
toward(s):
          'compared with'. e.g.
   1.
          /oi do nayt toə(r)dz wot we: did/
          'they do nowt (nothing) towards what we did'
          'towards, in the direction of', as in S.E.
   2.
There is also the idiom:
         /wi man bi 'gerin taə(r)t wam/
          'we <u>mun</u> (must) be getting towards <u>whoam</u> (home)' = 'we <u>must</u> go now'
'tween: /twi:n/ is an aphetic form of between.
'tween an': /twi:n en/ is an aphetic form of between and.
```

 \underline{up} : /op/. May be used as a preposition in the manner of down.

wi', with: /ww, (wvð)/

- 1. 'with', as in S.E.
- 2. 'by'. In this sense it is used extensively in the passive:

/av won om op wit nek | wi e ro:p/
'I've wun (wound) them up with the neck..with a rope'

/i: id: saidid i:d av it | don wit '00:lid3 men/
'he decided he'd have it..done with the haulage men'
/wi/ also means 'by' in certain other constructions, which

are not passives:

/wi 'finist op wi 'porin it in &:(r) 'pokit/
'we finished up with putting it in our pocket'
/ðe(r)z 'o:ni ty: mæ:(r)ks ði ken tæl em wi:/
'there's only two marks they can tell them with'
/en am bæ:nt 'finis op wi 'se:in ðis næ:/
'and I'm boun to (going to) finish up with saying this now'

The vowel in /wi/ is fronted, and often lengthened before /j/ or a vowel; in the latter case, a linking [j] is usually interposed between the two vowels:

[wi j it] 'with it'
[wi: j it] 'with it'
[we': j a klog] 'with a clog'

The form /wi:/ may be found when with functions as an adverbial particle.

^{1.} See section 6.8.8.

within: 'beyond, against'. I have only recorded this usage of the preposition in modified speech, e.g.

/i:z not wi'oin 'swe:rin/

'he's not within swearing' (i.e. he swears, swears on occasion, is not averse to swearing)

Pronunciations of within without /ð/ are possible. Whilst I do not have the evidence to prove that within can be used with the meaning 'beyond, against' in the most residual speech, I can say with certainty that to can be thus used.

Many of the above uses of prepositions are carried over into modified speech. Sometimes, of course, a change of pronunciation is involved. One or two items, however, such as <u>till</u>, <u>tin</u>, would probably be too residual or considered too broad to be used in modified speech.

6.11. Conjunctions and Conjunctional Phrases:

The following list of conjunctions is compiled chiefly on a comparative basis. A conjunction such as though is a part of the dialect, but is not included here because its form and use appear to be much as in other varieties of English. That is used, but not so often as in S.E.: as is far more usual, whether alone or in combination with other elements. Than may be used nowadays in the comparison of adjectives and adverbs, but nor, till and tin are often found in the most residual speech. Although it may be suspected that a conjunction such as whilst is of low frequency in the dialect, speakers nowadays are subjected to such a wide range of information from such a variety of sources that it is difficult to say that some particular conjunction is definitely not a part of the dialect. usages which follow are common, and the examples widely representative, unless there is any statement to the contrary.

acause, acoz: /e'kɔ:z, e'kɔz/ 'because'.

Now rare, although possibly once more common.¹

Cf. cause, cos below.

afore: /e'fvə(r), e'foə(r)/ 'before', e.g.

/e'foə(r) öɛ gi:t e'we:t pit le:n/

'afore thou geet (got) away the pit lane'

^{1.} The forms feature more extensively in literature and in glossaries, e.g. Partington (1920:1).

Sometimes $f \circ (r)$ is encountered, in which case it is difficult to say whether <u>afore</u> or <u>before</u> is being used, e.g.

/foə(r) ði kopt ez/
'afore/before they copped (caught) us'

and: /on, en/ 'and'.

Sometimes there is extensive joining of clauses with and. 1 More specialised uses are:

(i) SUBJECT + NEGATIVE AUXILIARY + half and + VERB (+ too).

The construction intensifies the verb, e.g.

/it 'didnt e:f on stink tr:/
'it didn't half and stink too!' - i.e. 'it
stank dreadfully!'

Cf. further, won't half and, can't half and, BE NEGATIVE + half and + ADJECTIVE. The same constructions also occur without and.

(ii) In some dialect constructions, a negative clause may be followed by

and + SUBJECT + NEGATIVE FINITE VERB =
S.E. without + GERUND

- e.g. /a no: wie(r) on a one noiz av noin
 'I know whether they are thou knows...I've noan (not)

 wekt ie(r) se long en a deint no:/

 worked here so long and I don't know...' (= without knowing...')
- (iii) bonny well and is an intensifier and emphatic device, e.g. /al bet av bin 'boni wiel on ble:md/

'I'll bet I've been bonny well and blamed!'

^{1.} Peters (1968: 237) mentions this as a feature of O.E.

(iv) /www:(r)nt'gy:in wom til foer en 'e:fpast/
 'we weren't going whoam (home) till four and
half-past'

Any time around or between those mentioned is meant.

(v) <u>try and</u> + INFINITIVE 'try to'

go and + INFINITIVE 'go to'

Constructions of this type with <u>and</u> could be described as colloquial English.

and then: /en δεη/ 'so that, in order that'. e.g.

/al rait dis dein en den a deint fe(r)'get/
'I'll write this down and then I don't forget'

Although the construction might be construed as CONJUNCTION +

ADVERB, I am inclined to treat and then as one unit in this

type of construction. Other uses of and + then, i.e.

CONJUNCTION + ADVERB, are also found in the dialect. Nearer

to this latter usage is and then = 'moreover, furthermore,

in addition, and then again', e.g.

/it we(r) dgost e 'likl bit mye(r) 'moni|
'it were (was) just a little bit more money..

en öɛn mi 'broðe(r) wekt et öe se:m pit/
and then my brother worked at the same pit'
[the full form of definite article is due to careful explanation]

And then also functions as a final tag, e.g.

/lats tav e ly:k en oan#/
'let us (me) to have a look and then #'

^{1.} Hornby (1975: 29) does so.

The meaning of the tag is 'and then I will answer your question, and then I will see what I can do, etc.'

as: /ez/.

- 1. 'so that', e.g.
 - /its e 'blodi 'kare van ð: wants
 'it's a bloody caravan thou wants (needs)
 ez ð: ken go wie(r) ð: wants/
 as thou can go where thou wants'
- 2. 'that', e.g.

/o: e lot on om θιnk ez we:ks wi:ks
'oh a lot on (of) them (people) think as Wakes Weeks

z wen δι a δe(r) | 'olidi de:nt δι/
is when they have their..holiday, don't they?'

as how: /ez ε:, ez 3 y/ 'that', e.g.

/a θιnk ez 3 y i: mεt dy:/

'I think as how he might do'

as what: /ez wot/ 'in comparison to, when compared with', e.g.

/its o: t[e:nd3 te'de: ez wor it we(r)

'it's all change[d?] today as what it were (was)

'twenti je(r) sin/

beawt: /be:t/ 'without', e.g.

/it 'doznt we:(r) long be:t its 'detti/
'it doesn't wear long without it's dirty'.

twenty year sin (years since)'

Cf. without below.

becaud: /bi'ko:d/ 'because', e.g.

/bi'ko:d ðin 'getn/

'because they have got'

Forms in d are much less frequent than forms in z.

becau(se): /bi'kɔ:(z), bi'kɔ(z)/ 'because'.

Occasionally one hears forms without /z/. Such forms might constitute a reduction of <u>becaud</u> or <u>because</u>:

[bi'kɔ² it?]

'because it'

/bi'ko: $\delta e(r) we(r)/$

'because there were'

being, being as, being as how, (being how): /'bi:in,
'bi:in ez, 'bi:in ez &:, 'bi:in &:/ 'because, since, seeing
that', e.g.

/at ad tel im 'bi:in i:z 'aterris/
'I had to tell him being he's Irish'

between and: /bi'twi:n en/ 'by the time that, before'

This is very common indeed, e.g.

/bi'twi:n en a gi:t te 'mantsisθe(r)/

'between and I geet (got) to Manchester'

but: final /t/ is often /r/ before a vowel.

but what: /bot wot/ 'but' in the sense:

/os ns:r az e 'minit bot wat i:z 'msiderin di/
'thou never has a minute but what he's moidering thee'

```
'cause, 'cos: /kɔːz, kɔz/ 'because'.
        These are presumably reduced forms of either
                     Examples:
because or acause.
        /ko:z ðe mon/
        'because thou must'
        /kpz a want fe(r)t/
        'because I want for to ... '
chance: /tsan(t)s/ 'in case, lest', e.g.
        /tSants i: komz/
        'chance he comes'
considering: /ken'siderin/ 'in view of the fact that,
seeing that'. e.g.
         /i: si:mp no: e lot e'bɛ:t it ken'siðerin i:
        'he seemed to know a lot about it considering he
        weint dua(r)/
        weren't (wasn't) there'
directly: /de'rekli/ 'as soon as', e.g.
        /de'rekli it we(r) de:(r)k/
        'directly it were dark'
either: pronounced /'e:ðe(r)/, more modified /'aiðe(r)/,
occasional /'i:ðe(r)/.
except: /sk'sspt/ 'except that, only', e.g.
        /i: 'wodnt ev kom Ek'sEpt i: 'wantid e
        'he wouldn't have come except he wanted a
```

we: (r)d wt e(r)/
word with her'

except if: /ɛk'sɛpt ɪv/ 'unless', e.g.
/we(r) 'gY:ın ɛk'sɛpt ɪv ɪts 're:nɪn/
'we're going except if it's raining'

else: /εl(t)s/ 'or else', e.g.

/δε jy:st at dy: wot δε we(r) tayd εlts

'thou used to have to do what thou were told else

δε gi:t won rest et said et 'θιθτο:]/
thou geet (got) one right at the side of the earhole'

for all (as): /fer o:, oez/ 'for all (that), despite the fact that'.

Used more extensively than its S.E. equivalent.

Examples:

/e(r) we(r) oat 'sieries fer o: 'Evri, bodi
'her (she) were (was) that (so) serious for all everybody
we(r) 'layfin oe(r) jedz o:f/
were laughing their heads off'
/fer pez i: 'ri'taie(r)d laik/
'for all as he retired, like,...'

for if: /fer \(\text{v} \) 'in case, lest', e.g.

/fer \(\text{v} \) \(\text{tringz} \)

'for if it [the telephone] rings'

heaw: $/\epsilon$:/ 'how' = a rare form of as how.

if: /iv/ is the more usual pronunciation - it does not appear to be conditional upon a following vowel or lenis consonant. /if/ is also met, but can indicate code-switching.

This conjunction is sometimes absent in the dialect when it would be present in S.E., e.g.

/δε dʒɔgd e bit it wer ε:t/
'[If] thou jogged a bit, it were out'

less: /les/ a form of unless.

lest: /lest/ 'lest'.

The meaning is as in S.E.

<u>like</u>: /lark/ 'as', is used very frequently as a conjunction in the dialect, e.g.

/lark e sed br'foe(r) we:/
'Like I said before, we...'

neither: pronounced /'no: $\delta e(r)$, 'ne: $\delta e(r)$ /, more modified /'nai $\delta e(r)$ /, and occasional /'ni: $\delta e(r)$ /.

 $\underline{\text{nor}}$: /ne(r)/ 'than' in the comparison of adjectives and adverbs, e.g.

/a no:d moə(r) ner ad tɛl/
'I knowed (knew) more nor I'd tell'

only: /'o'ni, 'o'nli/ 'but, except that'.

The word is used very frequently as a conjunction, e.g.

/a de:nt 'rieli no: 'o'nli bre:n fle:(r)z
'I don't really know only brown flour is

se'pois to beit | Ooil et wiit dge:(r)m/
supposed to be the .. whole of the wheat-germ'

same as: /se:m ez/. See next entry.

samen: /'se:men/ 'just as, like, same as'.

This form was used by three informants as a variant of same as, e.g.

/'se:men a si/
'samen I say'

saying, etc.: /'se:in, etc./

Forms and function are as given for being, etc. above.

seeing, etc.: /'si:in, etc./

Forms and function are as given for being, etc. above.

sin: /sιn/ 'since', e.g.
/sιn a wer ε: jong/

'Sin I were how young?'

so as, so'd, so's: /soez, soid, soiz/ 'so that', e.g.
/soiz wi no: niid te/
'so's we [have] no need to...'

<u>till</u>: /til/ 'than' is used in the comparison of adjectives and adverbs, as <u>nor</u> and <u>tin</u> 2.

tin: /tin/

1. 'until', e.g.

/a livd ðið(r) tin a wer ɛit'ti:n/
'I lived there tin I were 18'

2. 'than' in the comparison of adjectives and adverbs, e.g.

/ad 're:ŏe(r) go: on e bo:t tin flat it/
'I'd rather go on a boat tin fly it'

up to: /op te/ 'until', e.g.
/op te a we(r) .siksti'sɛvn/
'up to I were 67'

where: /wiə(r)/ 'whereas', e.g.

/it let e bit ev e:r in || wiər iv i:
 'it let a bit of air in .. where if he
 ad iz kap on/
 had his cap on...'

which: /wits/.

There are uses of the relative pronoun which that resemble uses of conjunctions in S.E. $^{\mbox{\scriptsize l}}$

Equivalent meanings are approximately 'and; but; because; as'. Examples:

/δε dzos ty:k it en 'dostid it | wits δε 'thou just took it and dusted it..which thou

ken i madzint pail e dost wot did kom o:f/
can imagine the pile of dust what did come off'
(= approx. 'and')

/t land wer oind ben wit a bink it is ne:

'the land were owned then which I think it is now

bit bard e:(r)l e 'bradfe(r)d/

by the old Earl of Bradford'
(approx. 'as')

^{1.} See section 6.7.6.6.

without: /wi'δε:t/ is equivalent to S.E. without (PREPOSITION) + GERUND, e.g.

/δε 'kodnt 'karı ıt wı'δε:t ıt fɔ:d ε:t/
'thou couldn't carry it without it falled (fell) out'

Sometimes the dialect does not use a conjunction when S.E. probably or certainly would. It is generally pointless to speculate about which conjunction is "not being used", even on a comparative basis, except in the case of <u>if</u>, as conditional clauses are relatively clear logical constructions, e.g.

/3Yt went rong it we(r)t 'bonksmenz folt/
'[If] owt (anything) went wrong it were the Banksman's fault'

The following constructions involving measurements bear comparison with constructions containing or or to, but nonetheless clearly form a pattern in their own right:

/wit; went fe:(r) | Ori: foo(r) de:z/
'which went for .. three four days'

/æ: Sed se ois las ten 'fifti:n je(r)/
'I should say this last ten fifteen year (years)'

/its e'be:t || e:f pas nain ten e'klok/
'it's about .. half-past nine ten o'clock'

/en e'be:t 'siksti 'sevnti fi:t ei/
'and about sixty seventy feet high'

etc.

Clauses are sometimes juxtaposed without a pause or conjunction:

/'aniro:d i: kom it 06:s i: siz/
'anyroad (anyway) he come (came) in the house he says...

/ o:n did e'be:t 'eitti:n mon0 a kom e:t / [I] on[ly] did about 18 month (months) I come (came) out.'

Both of these examples might be analysed as consisting of two sentences, or might be compared with similar constructions in which the clauses are linked by a conjunction. The former suggestion might appear to be closer to the observable facts. However, a conjunction - presumably when - would be more usual in:

/wen je Oink e'be:t wot wi: jr:s [t] dr:| 'when you think about what we used to do..

we: we(r) jang/
[when] we were young'

Error is certainly a possibility here, and may have resulted from when's already having been used. However, it is difficult to insist upon such notions as error and reformulation, and a larger corpus would be required if more definite statements were to be made.

No conjunction is usually required in the construction what's to do:

/wots dy: je(r) not 'speikin to mi/
'what's to do you're not speaking to me?'
although as might be added after do.

There are cases of clauses not joined by a conjunction in the dialect, where it is by no means certain that spoken S.E. would use a conjunction either, e.g.

/a 0inkt | 'bigist ck'spens we(r) 'flaiin
'I think the..biggest expense were flying
it oie(r)/
it there'

It is quite likely that the dialect joins such clauses without a conjunction more often than does S.E., but a quantitative analysis lies outside of the scope of the present study.

Many of the distinctive conjunctional usages given in the above account are to be found in modified speech.

I have heard between and and many others even in speech which is highly modified at the phonological level.

7. CONCLUSION

Inevitably in a study of this length, a number of conclusions have been drawn earlier in the body of the work. To some extent too, a description may stand qua description, and does not always require further comment. However, it may be useful to draw together here some of the principal points which were made in the study, and to attempt certain overall conclusions. In particular, it is appropriate to make a concluding assessment of the account of morphology and syntax, since a hypothesis concerning variation at the grammatical level in English dialects was advanced earlier in the study.

Theory was defined in section 1. as being characterised by an explicit approach. I have tried to be explicit by specifying the methods adopted in this study (placing them, to some extent, in a wider linguistic context), and by describing fieldwork and transcriptional experiences at some length. Theory was also defined as "a certain method justified". The methodology adopted here seems to be justified for the following reasons:

- 1. An inventory of phonemes has been drawn up, and a phonemic inventory constitutes a systematic and economical explanation of a very substantial amount of data.
- 2. The corpus has been accounted for more fully than is usual by a systematic and economical arrangement in the

form of a relatively small number of modification rules operating on the description of the comparative distribution of phonemes.

3. A considerable number of grammatical rules have been postulated.

It was also asserted in section 1. that theory is \underline{a} prioristic because of the influence of the frame of The study began with an account of the area of Farnworth and district, which suggested that this area constituted a cultural region of considerable homogeneity. This account was part of the initial frame of reference. After transcribing and analysing the tape recordings of the speech of the area, it is possible to say that the linguistic evidence would seem to bear out this impression of relative homogeneity. Of the variants in the phonology, some have been accounted for as positional, i.e. as allophones, some have been accounted for as stylistic by means of the modification rules, others still remained in free variation. However, the last would occur, I imagine, in any detailed examination of a dialect, and they have not proved to be so heterogeneous as to prevent the setting up of an inventory The basic assumption concerning the homoof phonemes. geneity of the area would therefore appear to have been correct.

Sections 1.2, 2 and 3 have sought to show the many different ways in which and the degree to which methods,

fieldwork and the process of transcription determine or affect the data. Whilst the accounts of fieldwork and transcription given here are to some extent subjective, and open to considerable refinement, they are nonetheless sufficient to show that a study is determined fundamentally and in many different ways by methods, fieldwork and transcription. It follows, then, that if fieldwork and transcription are not described in detail, a study is liable to be vague, or inexplicit, and therefore theoretically weaker. The usefulness and epistemological status of a study were seen to be further conditional upon the subsequent archiving of tape recordings, transcriptions and background materials, to allow for further evaluation and analysis by others in the future.

In the discussion of approaches to dialectology, and the specification of the approach adopted here, an important point to emerge was the fact that varieties of speech tend to be mixed together in monolingual repertoires, so that it is impossible to elicit stylistically uniform speech. Since, however, the methodology of working with a corpus required that the corpus should be evaluated in toto, it followed that some attempt had to be made to account for more and less broad types of speech. The rules of speech modification set up in this study were modest in number, and general in character, so as to cover a large number of variants. It was indicated that a more particular approach might be adopted in further work by entering in matrices

all the different variants used by different informants, with the aim of correlating linguistic variants with variable features in informants' socio-economic profiles. Even so, the approach adopted here, utilising modification rules of a general character, has been able to account for the corpus much more fully than would have been the case in a more idealised study. Furthermore, the fact that variants in the speech of dialect speakers could be understood in terms of the speech of relatives, children, and other more modified speakers around them illustrated a point made in section 1.1.1.7, namely that the standard language does not operate directly on the dialects, but that rather they are influenced by the next social level above them. The need for a concept of the type "modification to or towards a variety or varieties of Northern Regional Standard" is similarly illustrated.

In section 5 a phonology of the dialect was presented. Some matters relating to the phonology, such as the need to establish precise symbols for glottalisation, and the likely need for /e/ and /e:/ as phoneme symbols, were apparent from a fairly early stage in the transcription process. 1 phonology was based upon both phonetic and phonemic considerations, and an attempt was made to account for the data. at the systemic, distributional and realisational levels.2

Cf. section 3.9. Cf. section 5.

Not surprisingly, a lot of variants were found in the transcriptions of the corpus of spontaneous speech. Whilst some were accounted for as being positional, and others as stylistic, still others remained. This illustrated the need for a phonetic as well as a phonemic approach - unless one made very major assumptions indeed about the importance and accuracy of the phonemic analysis. At the systemic level, it proved possible, even if sometimes difficult, to set up an inventory of phonemes. This fact confirmed the initial hypothesis concerning the relatively homogeneous character of the area of Farnworth and district. The systematic character and economy of the phonemic approach satisfied theoretical requirements up to a point, but the addition of modification rules enabled the study to meet the requirements of corpus methodology more fully than would otherwise have been the case, by accounting more completely for the corpus. It was shown earlier that stylistic variation within a corpus was unavoidable.

The preparation of the phonology was complicated by ongoing change. A possible residual /a:/-phoneme was subsumed under /ai/, whilst a clear boundary between /ɛ:/ and /æ:/ could not always be drawn. Variants intermediate between /ɛ:/ and /æ:/ could therefore pose problems for the phonemicist, just as half-length variants of the type [e·] did, for /e/ and /e:/ could be distinguished by length. A word such as dog might sometimes be phonemicised /dog/,

^{1.} For discussion see section 5.4.2.

and sometimes /dog/. Both occur, but a vowel midway between /o/ and /o/ is more difficult to handle. The opposition between two vowels may be neutralised in some phonemic environments, in some words, by some speakers, on some occasions... I have tried to indicate in the body of the thesis where such problems occurred. The subsection Diphthongs Not Accorded Phonemic Status was also a discussion of the problems of assigning phonetic variants to phonemes. The recession of final and preconsonantal /r/ caused problems of phonemicisation too, and the solution adopted here was to place the /r/ in brackets, usually after a diphthong.

Ongoing changes sometimes led to the question of different phonemic systems being in operation. Is the phoneme $/\epsilon$:/ still needed in modified speech? At what point might [h] or perhaps [ŋ] be accounted a phoneme? Should /tr, dr/ be accounted phonemes in modified speech, when dialect /0, δ / give way to /t, d/ before /r/? These questions transcend the limits of the present study, but it is nonetheless important to raise them.

^{1.} Cf. section 5.4.

distinctions <u>vis-à-vis</u> RP, again illustrating the point that change (if we assume that /ɔə, oə/ entered the system under the recession of post-vocalic /r/) is not necessarily or directly towards RP. The general absence of /h/ and /ŋ/- phonemes in modified speech also illustrated the need for a concept of the type "modification to or towards a variety or varieties of Northern Regional Standard".

The existence of long /æ:/ (and also long /o:/) before voiceless fricatives is an interesting feature of the area, which linguistic geographers have overlooked when regarding short versus long 'a' as a marker dividing the North from the South. The modification of dialect [a:]-types before voiceless fricatives to [a]-types is a modification away from RP, which once again illustrates the need for a concept of modification towards a Northern Regional Standard.

The account of the consonant phonemes of the dialect proved to be longer than in many studies. Had resort been had to instrumental analysis, more information still would presumably have been forthcoming. It was found that quite a lot could be said about the consonants in terms of aspiration, affrication, gemination, glottalisation, devoicing, voicing, nasal release, syllabic quality, elision, assimilation, distribution, and so on. Certain features of interest were quite general, e.g. gemination. More specific features of interest included the distribution of $/\theta$, $\delta/$, and the comparatively heavier functional load borne by the opposition $/\theta/\neq /\delta/$; the voiced flap allophone [t],

which exists as a variant of /t/ and sometimes in free variation with variants of /r/; the clusters /tl, dl/ which are quite distinctive in word-initial position; the uvular allophones of /k, g/; occurrences of initial /j/; and so on. Very many parts of the description yielded data which would be of interest to the historical linguist: occurrences of /j/ initially and in initial clusters; /f/ in /d3vf/ 'dough' and /'3vfn/ 'Westhoughton/; /θ/ in /lsiθ/ 'Leigh'; and so on.

/r/ and /l/, when following a vowel or diphthong phoneme, constituted environments which were likely to condition variants. The velar quality of /l/ was an important factor - not only because many associate such a quality more with Southern dialects, but because this helped to explain precisely why /l/ formed such a influential environment in respect of vowel and diphthong phonemes. Furthermore, the need to include the diphthongs /oə, ɔə/ in the system could be understood in terms of the recession of historical post-vocalic /r/ rather than as being due to the influence of other varieties of English.

The devoicing of /d/, or the occurrence of /t/ for /d/ in context under the influence of the definite article, the preposition to, or the second person singular interrogative, together with the grammatical exceptions to the use of / θ , δ / where RP has /t, d/, illustrated the point that the preparation of descriptions is not a purely linear process moving from phonology to grammar.

Whilst some features of the consonant system of the dialect may be peculiar to parts of Lancashire, and some perhaps highly localised, it nonetheless seems reasonable to suggest that many other dialects too might require a fairly protracted description of their consonant phonemes.

The present analysis did not include suprasegmentals, although it seems likely that a study of the suprasegmental features of the dialect would be interesting. It is also likely that instrumental analysis would make a significant contribution to the discussion of gemination, vowel length, and so on.

The account of the morphology and syntax of the dialect given in section 6 was offered first and foremost as a part of a description or grammar of the dialect, and secondly as a test of a hypothesis that English dialects differ significantly at the morphological and syntactic levels. Partial though the account remains, it surely contains sufficient material to confirm the hypothesis. The account of the definite and indefinite articles features which occur an enormous number of times in a large corpus - shows clearly that the articles differ extensively in both form and distribution from the articles in many other dialects of English. The account of the pronouns of the dialect - again features which occur frequently and on the basis of which one may therefore generalise more safely - also revealed very significant

^{1.} Cf. sections 1.1.1.8. and 6.1.

differences of form vis-a-vis SE, especially in the case of reflexive, demonstrative and relative pronouns; differences in the use of subject and object cases; and marked syntactic differences in the use of dependent pronouns. Indeed, the very length of the descriptions of the definite article and, say, the relative pronoun, is in itself instructive, and suggests that there is more to be said about such features than is usually said. Equally, the description of the verbs of the dialect includes many features which are highly distinctive in comparison with SE and other dialects. One might recall the use of -en plural endings, the distinctive immediate future construction with /bs:nt/ 'boun(d) to', weak verbs which are strong in SE and vice versa, the use of be as a past tense auxiliary, and so on. The modal and auxiliary verbs or anomalous finites in particular displayed many points of comparative interest, especially the highly distinctive second person singular interrogative forms. The account of affirmation and negation revealed a system of affirmative and negative particles which is richer than that of SE, and analogous to the systems of French and German. Yet other grammatically distinctive features were described in the discussion of adverbs, prepositions and conjunctions.

When preparing the description of the morphology and syntax of the dialect, it was of course impossible for me not to be constantly aware of the need for more material, more time, and subtler analytical tools. Thus, some areas of grammar warrant much more attention than they

have received here. For instance, it would be necessary to have a much larger corpus to deal adequately with all forms of the anomalous finites.

Nonetheless, it seems impossible to escape the conclusion that the dialect of Farnworth and district is grammatically distinctive when compared to SE, and many other dialects of English. It is furthermore extremely unlikely that this particular dialect should be unique in this respect, and it therefore seems reasonable to suggest that grammatical variation in English dialects is underestimated, and that further detailed descriptive work might profitably be carried out on English dialects at the morphological and syntactic levels. This is a major conclusion of the study, and has some bearing on the theory of English dialectology in general. 1

^{1.} Cf. section 1.1.1.4.

7. BIBLIOGRAPHY

- ABERCROMBIE, D. (1965): Studies in Phonetics and Linguistics, London.
- ALBROW, K. (1974): Review of O'CONNOR (1973) in <u>Journal of</u>
 the <u>International Phonetic Association 4/1, 49-51.</u>
- ALSTON, R.C. (1971): A Bibliography of the English Language from the Invention of Printing to the Year 1800.

 Vol. IX, English Dialects, Cant and Vulgar English, Leeds.
- ATKINSON, J. (1971): A Handbook for Interviewers, London.
- Atlas Linguarum Europae, Premier Questionnaire

 Responses for Harwood. Copy deposited in the Archives
 of the Centre for English Cultural Tradition and
 Language, University of Sheffield. Fieldworker:
 P. Wright. Cf. WEIJNEN et al. (1974).
- BACH, A. (1950): <u>Deutsche Mundartforschung</u>. <u>Ihre Wege</u>, <u>Ergebnisse und Aufgaben</u>, Heidelberg.
- BAGLEY, J.J. (1956): A History of Lancashire, London.
- BAMFORD, S. (1850): The Dialect of South Lancashire,
 Manchester.
- BARTON, B.T. (1875): <u>Historical Notes of Farnworth and Kersley [sic]</u>, Bolton and Farnworth.
- BARTON, B.T. (1887): <u>History of Farnworth and Kersley</u> [sic],
 Bolton.
- BLOOMFIELD, L. (1955): <u>Language</u>, London. First published 1933.
- Bolton Evening News, Friday, 17 November, 1972.

- BOWYER, R. (1973): A Study of Social Accents in a South

 London Suburb. Unpublished M.Phil. diss., University
 of Leeds.
- BOYD, A.W. (1954): "The Dialect of Great Budworth in Cheshire" in: <u>Journal of the Lancashire Dialect Society</u>, 4, 16-20.
- BROKER, H. (1930): Zu den Lautverhältnissen der Lancashire-<u>Dialekte. Auf Grund'von Lautplatten aus den englischen</u> <u>Gefangenlagern</u>, Berlin.
- BROOK, G.L. (1968): "The Future of English Dialect Studies", in: <u>Leeds Studies in English</u>, N.S. ii, 15-22.
- CAMPBELL, L. (1972): "Is a Generative Dialectology Possible?", in: Orbis, 21, 289-98.
- CAMPROUX, C. (1960): "Ma méthode d'enquête pratiquée pour 'l'Étude syntaxique des parlers gévaudanais'", in:

 Orbis, 9, 26-33.
- CARNEY, E. (1969): "Push-Button Dialectology?", in: <u>Journal</u> of the <u>Lancashire Dialect Society</u>, 18, 9-15.
- <u>Centenary Special:</u> <u>Farnworth and Worsley Journal Centenary</u>
 <u>Supplement</u>, 30 August, 1973.
- CHAO, Y-R. (1934): "The Non-Uniqueness of Phonemic Solutions of Phonetic Systems", in: M. JOOS (ed.), Readings in Linguistics, New York (1958), 38-54.
- CHAO, Y-R. (1970): "Some Aspects of the Relation between Theory and Method", in: GARVIN (1970), 15-26.
- Charter: Borough of Farnworth. Charter of Incorporation, 1939
 Bolton (1939).
- CHAURAND, J. (1972): <u>Introduction à la Dialectologie Française</u>
 Paris, Brussels, Montreal. With a Preface by J. Batany

- CHOMSKY, N. (1965): Aspects of the Theory of Syntax, Cambridge (Mass.) and London.
- CHOMSKY, N. (1966a): <u>Cartesian Linguistics</u>. A Chapter in the History of Rationalist Thought, New York and London.
- CHOMSKY, N. (1966b): Topics in the Theory of Generative Grammar, The Hague, Paris.
- CHOMSKY, N. (1968): Radio Interview with Stuart Hampshire. Printed in <u>The Listener</u>, 30 May, 1968.
- CHOMSKY, N. (1972): <u>Language and Mind</u>. Enlarged edition, New York.
- CLARKE, C.A. (no date): <u>Tum Fowt Sketches</u>, Pendlebury and Sons, Bolton, and John Heywood, Manchester.
- C.O.D.: The Concise Oxford Dictionary of Current English (eds. H.W. FOWLER and F.G. FOWLER), Fifth edition (rev. E. McINTOSH and G.W.S. FRIEDRICHSEN), Oxford (1972).
- CRYSTAL, D. (1971): Linguistics, Harmondsworth.
- CRYSTAL, D. (1975): The English Tone of Voice. Essays in intonation, prosody and paralanguage, London.
- <u>Daily Telegraph</u>, 8 November, 1977: "'Put wud i' th' ole' baffles the Asians".
- DIETH, E. (1946): "A New Survey of English Dialects", in: Essays and Studies, xxxii (1946), Oxford (1947), 74-104.
- DIJK, T.A. VAN (1972): Some Aspects of Text Grammars. A Study in Theoretical Linguistics and Poetics, The Hague, Paris.

- DUNCAN, P. (1972): "Forms of the Feminine Pronoun in Modern English Dialects", in: WAKELIN (1972b), 164-81.
- DYSON, J. (1881): Rural Congregationalism; or, Farnworth

 as it was fifty to seventy years ago. With humorous
 sketches and anecdotes, illustrating Lancashire
 manners and customs, Manchester.
- EKWALL, E. (1922): The Place-Names of Lancashire, Publications of the University of Manchester, English Series XI, Manchester, London, etc.; and Remains of the Chetham Society, New Series, Vol. 81.
- ELLIS, A.J. (1889): On Early English Pronunciation. Part V:

 Existing Dialectal as Compared with West Saxon

 Pronunciation, London.
- Enclosure Act (1796): An Act for Dividing, Allotting, and
 Inclosing, certain Commons and Waste Lands, within
 the Manors or Lordships of Farnworth and Kersley, in
 the Parish of Dean, in the County Palatime of Lancaster.
 Bound photostat copy, Bolton Library. Original at
 Lancashire Record Office.
- Farnworth, Lancashire. Official Guide and Industrial Handbook.

 Issued by Authority of the Borough of Farnworth

 (no date).
- FASHOLA, J.B. (1971): "Structural and Non-structural Factors in Linguistic Interference. (A study of the influence of RP on a speaker of English provincial standard from Workington, Cumberland)", in: Zeitschrift für Dialektologie und Linguistik, 38, 296-312.
- FORD, E. (1974): "Westhoughton Terms and the Phonological Puzzles They Bring", in: <u>Journal of the Lancashire</u>

 <u>Dialect Society</u>, 24, 27-30.

- FRANCESCATO, G. (1964): "Dialect Borders and Linguistic Systems", in: <u>Proceedings of the Ninth International Congress of Linguists, Cambridge, Mass. 1962</u>, London, The Hague, Paris, 109-14.
- FREEMAN, T.W., H.B. ROGERS and R.H. KINVIG (1966): <u>Lancashire</u>, <u>Cheshire and the Isle of Man</u>. Series: <u>Regions of the British Isles</u>, Nelson.
- FREUDENBERG, R. (1965): "Zur Entwicklungsgeschichte der dialektgeographischen Methode", in: Zeitschrift für Mundartforschung, xxxii, 170-82.
- FRINGS, T. (1948): Grundlegung einer Geschichte der deutschen Sprache, Halle (Saale).
- FUDGE, E.C. (1972): "Phonology", in: J. LYONS (ed.): New Horizons in Linguistics, Harmondsworth, 76-95.
- FUDGE, E.C. (ed.) (1973): <u>Phonology</u>. <u>Selected Readings</u>, Harmondsworth.
- GARVIN, P.L. (ed.) (1970): <u>Method and Theory in Linguistics</u>. Series: <u>Janua Linguarum</u>, <u>Series Maior</u>, 40. The Hague, Paris.
- GASKELL, A. (1964): The History and Traditions of Clifton, Pendlebury.
- GEDDIE, W. (ed.) (1970): Chambers Twentieth Century Dictionary
 Revised edition with supplement, Edinburgh.
- GIMSON, A.C. (1974): An Introduction to the Pronunciation of English, 2nd edition, repr., Bristol.
- GLEASON, H.A. (1961): An Introduction to Descriptive

 Linguistics. Revised edition, New York, Chicago,
 San Francisco, Toronto and London.
- GLINZ, H. (1965): Deutsche Syntax, Stuttgart.

- GOOSSENS, J. (1971): "Die Begrenzung dialektologischer Problemgebiete", in: Zeitschrift für Dialektologie und Linguistik, 38, 129-44.
 - GORDON, E.V. (1966): An Introduction to Old Norse, 2nd edition rev. A.R. TAYLOR, Oxford.
 - GÖSCHEL, J. (1973): Strukturelle und instrumentalphonetische

 <u>Untersuchungen zur gesprochenen Sprache</u>. Series:

 <u>Studia Linguistica Germanica</u>, 9, Berlin, New York.
 - GREGG, R.J. (1972): "The Scotch-Irish Dialect Boundaries in Ulster", in: WAKELIN (1972b), 109-39.
 - GUMPERZ, J.J. (1971): <u>Language in Social Groups</u>. <u>Essays by</u>

 <u>John J. Gumperz</u>. <u>Selected and Introduced by Anwar</u>

 <u>S. Dil</u>, Stanford.
- HAMEYER, K. (1975): "Variation in Dialect Proficiency and Restructuring", in: Phonetica, 32/1, 24-37.
- HAMMARSTRÖM, G. (1967): "Zur soziolektalen und dialektalen Funktion der Sprache", in: Zeitschrift für Mundartforschung, 34, 205-16.
- HARGREAVES, A. (1904): A Grammar of the Dialect of Adlington (Lancashire), Heidelberg.
- HEDEVIND, B. (1967): The Dialect of Dentdale. Series: Studia Anglistica Upsaliensis 5, Uppsala.
- HEYWOOD, T. (1862): Essays on the South Lancashire Dialect.

 Publications of the Chetham Society, 57, Manchester.

 (Chetham Society's Miscellanies, vol. III.)
- HIGGINBOTTOM, E.M. (1962): "A Specimen of East Lancashire Speech Modified towards RP", in: <u>Le Maître</u>

 Phonétique, cxviii, 24-7.
- HJELMSLEV, L. (1970): <u>Language</u>. <u>An Introduction</u>. Translated by F.J. WHITFIELD. Madison, Milwaukee and London.

- HOCKETT, C.F. (1970): <u>A Course in Modern Linguistics</u>, New York, Toronto, Ontario.
- HORGAN, D.M. (1963): "The Use of Dialect", in: <u>Journal of</u> the <u>Lancashire Dialect Society</u>, 12, 4-9.
- HORNBY, A.S. (ed.) (1975): Oxford Advanced Learner's Dictionary of Current English, London.
- HOUCK, C.L. (1967): Methodology of an Urban Speech Survey.

 Typescript, Institute of Dialect and Folk Life Studies,

 School of English, University of Leeds.
- HÜNERT-HOFMANN, E. (1968): "Soziologie und Mundartforschung", in: W. MITZKA (ed.), Wortgeographie und Gesellschaft (= Festgabe für L.E. Schmitt), Berlin, 3-9.
- HURFORD, J.R. (1968): "The Range of Contoidal Articulations in a Dialect", in: Orbis, 17, 389-95.
- JACOBS, R.A. (1973): Studies in Language. Introductory

 Readings in Transformational Linguistics, Lexington.
- JONES, D.: <u>The Cardinal Vowels</u>. Two Records (Eng. 252-3 and 254-5), Linguaphone Institute.
- JONES, D. (1957): An Outline of English Phonetics, 8th edition, repr., Cambridge.
- JOOS, M. (1962): The Five Clocks, The Hague, Paris.
- Journal of the International Phonetic Association, 3/2, (December, 1973).
- KELLER, R.E. (1961): German Dialects, Manchester.
- KNOWLES, G. (1976): "Scouse: A Lancashire Dialect with an Irish Accent", in: Regional Bulletin (Centre for North-West Regional Studies, University of Lancaster), 5/16, 3-5.

- KOHLER, K.J. (1967): "Structural Dialectology", in: Zeitschrift für Mundartforschung, 34, 40-4.
- KOKERITZ, H. (1932): Phonology of the Suffolk Dialect, Uppsala.
- KURATH, H. (1939): <u>Handbook of the Linguistic Geography of</u>
 New England, Providence.
- LABOV, W. (1966): The Social Stratification of English in New York City, Washington.
- LADEFOGED, P. (1971): <u>Preliminaries to Linguistic Phonetics</u>, Chicago and London.
- LAKOFF, G. (1973): <u>Interview with Herman Parrett</u>. Reproduced by Linguistic Agency University of Trier, D-55, Trier.
- LESTER, M. (1971): <u>Introductory Transformational Grammar of</u>
 <u>English</u>, New York.
- LODGE, K.R. (1966): "The Stockport Dialect", in: <u>Le Maître</u>

 <u>Phonétique</u>, cxxvi, 26-30.
- LODGE, K.R. (1973): "Stockport Revisited", in: <u>Journal of</u>
 the International Phonetic Association, 3/2, 81-7.
- LOFTHOUSE, J. (1967): Portrait of Lancashire, London.
- LYONS, J. (1969): <u>Introduction to Theoretical Linguistics</u>, Cambridge.
- MARSHALL, J.D. (1974): <u>Lancashire</u>. Series: <u>City and County</u>

 <u>Histories</u>, Newton Abbot, London, North Pomfret,

 Vancouver.
- MARTINET, A. (1962): A Functional View of Language, Oxford.
- MARTINET, A. (1964): <u>Elements of General Linguistics</u>.
 Translated E. PALMER, London.

- MATTHEWS, P.H. (1972): Inflectional Morphology, Cambridge.
- McDAVID, R.I. (1971): "What Happens in Tennessee?", in: L.H. BURGHARDT (ed.), <u>Dialectology</u>. <u>Problems and</u> <u>Perspectives</u>, Tennessee, 119-29.
- McINTOSH, A. (1961): An Introduction to a Survey of Scottish Dialects, Edinburgh.
- MELCHERS, G. (1972): Studies in Yorkshire Dialects, 2 vols.

 Stockholm Theses in English, 9, Stockholm.
- MITZKA, W. (1943): Deutsche Mundarten, Heidelberg.
- MITZKA, W. (1952): Handbuch zum deutschen Sprachatlas, Marburg
- MOULTON, W.G. (1972): "Geographical Linguistics", in:

 T.A. SEBEOK (ed.), <u>Current Trends in Linguistics</u>,

 Vol. 9: <u>Linguistics in Western Europe</u>, The Hague,

 Paris, 196-222.
- MULDER, J.W.F. (1975): "Linguistic Theory, Linguistic Descriptions and the Speech-Phenomena", in: <u>La Linguistique</u>, 11/1, 87-104.
- MULLINEUX, C.E. (1964): <u>Mast and Pannage: A History of</u>
 <u>Swinton to 1765</u>, Pendlebury.
- NODAL, J.H. and G. MILNER (1875, 1972): A Glossary of the

 Lancashire Dialect. Publications of the Manchester

 Literary Club, Manchester and London, 1875.

 Reprinted: Bath, 1972.
- O'CONNOR, J.D. (1973): Phonetics, Harmondsworth.
- Official Guide = Borough of Farnworth. Official Guide to

 Farnworth, Lancashire. Issued by Authority of the
 Borough of Farnworth, 2nd edition, not dated, but
 published in 1956.

- ORTON, H. (1952a): "A New Survey of Dialectal English", in:

 Journal of the Lancashire Dialect Society, 2, 5-13.
- ORTON, H. (1952b): "The Isolative Treatment in Living North-Midland Dialects of the O.E. <u>e</u> Lengthened in Open Syllables in Middle English", in: <u>Leeds Studies in English</u>, vii-viii, 97-128.
- ORTON, H. and E. DIETH (1962): <u>Survey of English Dialects</u>.

 <u>Introduction</u>, by H. ORTON, Leeds.
- ORTON, H. and W.J. HALLIDAY (eds.) (1962-3): Survey of English

 Dialects. (B) The Basic Material. Vol. 1: The Six

 Northern Counties and the Isle of Man, 3 Parts, Leeds.
- OXLEY, J.E. (1940): The Lindsey Dialect. Leeds School of English Language Texts and Monographs, VIII, Kendal.
- PÅHLSSON, C. (1971): "A Sociolinguistic Questionnaire for English Dialectology", in: Zeitschrift für Dialektolog und Linguistik, 38, 257-71.
- PARTINGTON, S. (1920): Romance of the Dialect, Middleton.
- PAUL, H. (1960): <u>Prinzipien der Sprachgeschichte</u>, Darmstadt. First published 1880.
- PELLOWE, J., G. NIXEN, B. STRANG and V. McNEANY (1972):

 "A Dynamic Modelling of Linguistic Variation: The
 Urban (Tyneside) Linguistic Survey", in: Lingua,
 30, 1-30.
- PETERS, R.A. (1968): A Linguistic History of English, New York
- PIKE, K.L. (1943): <u>Phonetics</u>. <u>A Critical Analysis of Phonetic</u>

 Theory and a Technic for the Practical Description of

 Sounds. <u>University of Michigan Publications</u>, <u>Language</u>

 and <u>Literature</u>, Vol. XXI, Ann Arbor and London.
- PIKE, K.L. (1959): <u>Phonemics</u>. <u>A Technique for Reducing</u>
 Languages to Writing, Ann Arbor. First published 1947.

- PIKE, K.L. (1973): "Grammatical Prerequisites to Phonemic Analysis", in FUDGE (1973), 115-35.
- PLATT, J.T. and H.K. PLATT (1975): The Social Significance of Speech. An Introduction to and Workbook in Sociolinguistics. Series: North-Holland Linguistics, Amsterdam and Oxford.
- POP, S. (1950): <u>La Dialectologie: Aperçu historique et</u> méthodes d'enquêtes linguistiques, Louvain.
- QUIRK, R. (1974): The Linguist and the English Language, London.
- QUIRK, R., S. GREENBAUM, G. LEECH and J. SVARTVIK (1976):

 A Grammar of Contemporary English, 6th impression (corrected), London.
- REANEY, P.H. (1927): A Grammar of the Dialect of Penrith

 (Cumberland), Descriptive and Historical with Specimens

 and a Glossary, Manchester, London, New York, etc.
- Record, The: Publication of the Lancashire Authors' Association, 2/11 (January, 1914).
- RIEGEL, K.F. and G.C. ROSENWALD (eds.) (1975): Structure
 and Transformation, Developmental and Historical
 Aspects. The Origins of Behaviour, Vol. 3, New York.
- RINGGAARD, K. (1965): "The Phonemes of a Dialectal Area,

 Perceived by Phoneticians and by the Speakers

 Themselves", in: E. ZWIRNER (ed.): Proceedings of

 the Fifth International Congress of Phonetic Sciences,

 Basle and New York, 495-501.
- ROGERS, S. (ed.) (1976): They Don't Speak Our Language.
 Series: Explorations in Language Study, London.

- ROMMETVEIT, R. (1974): On Message Structure: A Framework for the Study of Language and Communication, London, New York, Sydney and Toronto.
- ROSETTI, A. (1973): Etudes Linguistiques, The Hague, Paris.
- RUOFF, A. (1973): Grundlagen und Methoden der Untersuchung gesprochener Sprache. Series: Idiomatica, Vol. 1, Tübingen.
- SAMARIN, W.J. (1967): Field Linguistics. A Guide to Linguistic Field Work, New York.
- SAXELBY, C.H. (ed.) (1971): <u>Bolton Survey</u>, Bolton, 1953.

 Reprinted, with an Introduction, by S.R. Publishers

 Ltd., 1971.
- SCHILLING, K.G. (1906): A Grammar of the Dialect of Oldham (Lancashire), Darmstadt.
- SCHUBEL, F. (1939): "Zur neueren englischen Dialektforschung", in: Englische Studien, 73 (1938/9), 344-80.
- SHAW, D. (1972): Statistical Analysis of Dialectal Distribution Preliminary Report, University of Kent at Canterbury, unpublished photocopied report.
- SHORROCKS, G. (1976): "A Linguistic Bibliography of the Lancashire Dialects, Part I", in: <u>Journal of the Lancashire Dialect Society</u>, 25, 6-14.
- SHORROCKS, G. (1977a): "A Linguistic Bibliography of the Lancashire Dialects, Part II", in: <u>Journal of the Lancashire Dialect Society</u>, 26, 6-9.
- SHORROCKS, G. (1977b): "Aspects of Two Lancashire Phonemes: /ɛi/ and /o/" in: <u>Journal of the Lancashire Dialect Society</u>, 26, 20-6.

- SHORROCKS, G. (1977c): "Long and Short <u>a</u> in a Lancashire Dialect", in: <u>Regional Bulletin</u> (Centre for North-West Regional Studies, University of Lancaster), 6/20, 5-6.
- SHORROCKS, G. (1977d): "The Use of h in Lancashire Dialect Writing", in: The Record (Publication of the Lancashire Authors' Association), no. 253 (December, 1977), 9-10.
- SHORROCKS, G. (1978a): "A Linguistic Bibliography of the Lancashire Dialects, First Supplement", in: <u>Journal of the Lancashire Dialect Society</u>, 27, 8-12.
- SHORROCKS, G. (1978b): "A Phonemic and Phonetic Key to the Orthography of the Lancashire Dialect Writer, Teddy Ashton", in: <u>Journal of the Lancashire Dialect</u>
 Society, 27, 45-59.
- SHORROCKS, G. (1978c): "'Yes' and 'No' in a Lancashire
 Dialect", in: Regional Bulletin (Centre for NorthWest Regional Studies, University of Lancaster),
 7/22, 12-13.
- SHORROCKS, G. (1978d): "The Phonetic Realisation of the Definite Article in Lancashire Dialect, and its Representation in Dialect Writing", in: The Record (Publication of the Lancashire Authors' Association), no. 256 (September, 1978), 13-14.
- SIESS, M. (1929): Untersuchungen zum Dialekt von Lancashire.

 Nach den Dialekteinlagen in den Romanen: Mrs. Gaskell:

 "Mary Barton", W.H. Ainsworth: "The Lancashire Witches

 Unpublished diss., University of Vienna.
- SIVERTSEN, E. (1960): Cockney Phonology, Oslo.
- SMITH, D.M. (1969): <u>The North West</u>. Series: <u>Industrial</u>

 <u>Britain</u>, Newton Abbot.

- SPEITEL, H-H. and J.Y. MATHER (1968): "Schottische
 Dialektologie", in: L.E. SCHMITT (ed.): Germanische

 Dialektologie II, 520-41. = Festschrift für Walther

 Mitzka zum 80. Geburtstag. = Zeitschrift für

 Mundartforschung, Beihefte, Neue Folge 6, Wiesbaden.
- STRANG, B.M.H. (1968): "The Tyneside Linguistic Survey", in:

 L.E. SCHMITT (ed.): <u>Verhandlungen des Zweiten</u>

 <u>Internationalen Dialektologenkongresses</u>, II, 788-94

 = <u>Zeitschrift für Mundartforschung</u>, Beihefte, Neue
 Folge 4, Wiesbaden.
- STRANG, B.M.H. (1974): A History of English, London.
- TAYLOR, F.E. (1901): The Folk-Speech of South Lancashire, Manchester.
- TRUDGILL, P. (1972): "Sex, covert prestige and linguistic change in the urban British English of Norwich", in:

 Language in Society, I/2 (October, 1972), 179-95.
- TRUDGILL, P. (1974a): The Social Differentiation of English in Norwich, Cambridge.
- TRUDGILL, P. (1974b): Sociolinguistics: An Introduction, Harmondsworth.
- TRUDGILL, P. (1975): Accent, Dialect and the School, London.
- VERHAAR, J.W.M.: "Method, Theory and Phenomenology", in: GARVIN (1970), 42-91.
- VIERECK, W. (1964): "Der English Dialect Survey und der Linguistic Survey of Scotland Arbeitsmethoden und bisherige Ergebnisse", in: Zeitschrift für Mundartforschung, 31, 333-55.
- VIERECK, W. (1966): <u>Phonematische Analyse des Dialekts von</u>
 <u>Gateshead-upon-Tyne/Co. Durham</u>, Hamburg.

- VIERECK, W. (1968): "Englische Dialektologie", in: L.E.

 SCHMITT (ed.): Germanische Dialektologie, II, 542-64

 = Festschrift für Walther Mitzka zum 80. Geburtstag

 = Zeitschrift für Mundartforschung, Beihefte, Neue
 Folge 6, Wiesbaden.
- VIERECK, W. (1971): "Britische und Amerikanische Sprachatlante in: Zeitschrift für Dialektologie und Linguistik, 38, 167-205.
- VIERECK, W. (1973): "The Growth of Dialectology", in:

 <u>Journal of English Linguistics</u>, 7, 69-86.
- VIERECK, W. (1974): "The Growth and Present State of Dialectology", in: <u>Transactions of the Yorkshire</u>

 <u>Dialect Society</u>, Vol. XIII, Part LXXIV, 19-45.
- VIGEON, E. (1977): Clogs or Wooden Soled Shoes. Reprinted from The Journal of the Costume Society, 1977, for City of Salford.
- WACKERNAGEL-JOLLES, B. (1971): <u>Untersuchungen zur Gesprochener</u>

 <u>Sprache: Beobachtungen zur Verknüpfung spontanen</u>

 <u>Sprechens. Göppinger Arbeiten zur Germanistik</u>, No. 33.

 Göppingen.
- WAKELIN, M.F. (1972a): English Dialects. An Introduction, London.
- WAKELIN, M.F. (ed.) (1972b): <u>Patterns in the Folk Speech of</u>
 the British Isles, London.
- WAUGH, E. (1857): Sketches of Lancashire Life and Localities, 2nd edition, Manchester.
- WAX, R.H. (1971): <u>Doing Fieldwork.</u> Warnings and Advice, Chicago and London.
- WEIJNEN, A. et al. (eds.) (1974): Atlas Linguarum Europae.

 Premier Questionnaire. I Texte (version anglaise);

 II Livre de Dessins, Nijmegen.

- WELLS, J.C. (1970): "Local Accents in England and Wales", in: <u>Journal of Linguistics</u>, 6/2, 231-52.
- WIDDOWSON, J.D.A. (1970): "The Problems of Transcription".

 Unpublished paper read to the British Association for the Advancement of Science at their annual meeting, 9 September, 1970, University of Durham.
- WILLIAMS, F. (1972): <u>Language and Speech</u>, Englewood Cliffs (N.J.).
- WINDELBAND, W. (1976): <u>Lehrbuch der Geschichte der Philosophie</u>
 16th edition edited by H. HEIMSOETH, Tübingen.
- WÖLCK, W. (1965): <u>Phonematische Analyse der Sprache von Buchar</u>

 <u>Frankfurter Arbeiten aus dem Gebiete der Anglistik und der Amerika-Studien</u>, 10, Heidelberg.
- WRIGHT, J. (1892): A Grammar of the Dialect of Windhill in the West Riding of Yorkshire, London.
- WRIGHT, J. (1905): The English Dialect Grammar, Oxford.
- WRIGHT, J.T. (1966): "Urban Dialects: A Consideration of Method", in: Zeitschrift für Mundartforschung, 33, 232-47.
- WRIGHT, P. (1954): A Grammar of the Dialect of Fleetwood

 (Lancashire), Descriptive and Historical. 2 vols.,

 Ph.D. thesis, University of Leeds, microfiche and
 microcard.
- WRIGHT, P. (1976): Lancashire Dialect, Clapham via Lancaster.
- ZIMMERMANN, R. (1973): "Structural Change in the English
 Auxiliary System: On the Replacement of <u>Be</u> by <u>Have</u>",
 in: <u>Folia Linguistica</u>, VI, 107-17.

APPENDIX

Alphabet of the IPA

THE INTERNATIONAL PHONETIC ALPHABET.

-											
		Bi-labial Labio- dental	Denial and Alveolar	Retroftez	Palato- alveolar	Aiveolo- palatal	Palatal	Velar	Uvular	Pharyngal	Glottal
CONSONANTS	Plosive	p b	t d	td			c 1	kg	q G		3
	Nasal	m n	n	η			р	ŋ	N		
	Lateral Fricative		4 h								
	Lateral Non-fricative .		1	l			Δ				
	Rolled		r						R		
	Flapped		t	τ					ĸ		
	Fricative	φβfv	0 g 2 z z	8 7	J 3	9 2	çj	xγ	Хх	2 1	h fi
	Frictionless Continuants and Semi-vowels	wyv					(p) į	(w)	¥		
VOWELS	Close	(учи)					Front Centri		Ţ		
	Half-close	(\$ 0)		ļ			0 \$	T O			
	Half-open	(ce o)					£ 00 20	O A			
	Open	(0)					a	αρ			

(Secondary articulations are shown by symbols in brackets.)

OTHER SOUNDS.—Palatalized consonants: t, d, etc.; palatalized f, g: f, g. Velarized or pharyngalized consonants: f, f, g, etc. Ejective consonants (with simultaneous glottal step): g', g', etc. Implosive voiced consonants: g, g, etc. g fricative trill. g, g (labialized g, g, or g, g). g, g (labialized g, g). g, g, g, g). g (labialized g). g, g, g). g0 (labialized g). g0 (labialized g). g0 (labialized g0, g0). g0 (labialized g0). g0) g0 (labialized g0) g0)

Affricates are normally represented by groups of two consonants (ts, tf, dz, etc.), but, when necessary, ligatures are used (ts, tf, dz, etc.), or the marks or (is or tz, etc.). also denote synchronic articulation (m) = simultaneous m and n). c, I may occasionally be used in place of tf, dz, and t, 2 for tz, dz. Aspirated plosives: ph, th, etc. r-coloured vowels: ez, az, oz, etc., or e², a², o², etc., or e², a², o, etc.; r-coloured o: ez or e² or z or e or z or e.

LENGTH, STRESS, Pitch.—: (full length). '(half length). '(stress, placed at beginning of the stressed syllable). (secondary stress). '(high level pitch); (low level); '(high rising); (low rising); '(high falling); (low falling); '(rise-fall); '(fall-rise).

Modifiers.—' nasality. breath (1 = breathed 1). voice (\$ = z\$). 'slight aspiration following p, t, etc. labialization (\$ = a very close e).

specially open vowel (q = a rather open e). \star tongue raised (e or q = e). \star tongue lowered (e or q = e). \star tongue lowered (e or q = e). \star tongue advanced (u or u = an advanced u, t = t). \to or u tongue retracted (t- or t = t-, t = alveolar t). \to lips more rounded. \to lips more spread. Central vowels: Y(u), U(u), U(u),

APPENDIX

Collecting Slip of the Centre for English Cultural Traditions and Language at The University of Sheffield

THE SURVEY OF LANGUAGE AND FOLKLOP	THE	SURVEY	OF	LANGUAGE	AND	FOLKL	ORI
------------------------------------	-----	---------------	----	----------	-----	-------	-----

COLLECTING SLI

Please write item exactly as heard/remembered/practised, and full details of how, when and why it is said or done, together with all relevant background information. Continue overleaf if necessary.

PLEASE QUOTE EXACT WORDS IF POSSIBLE.

Collected by Age	Address	
From whom did you learn this?	Relationsh	ip to you
Occupation Age	Born at	Now living at
From whom, where and when did they lear		
Where and when collected?		Today's date
Please return to: The Centre for English Cu The University, Sheffield S CS2/77		

APPENDIX

Sample Tape Deposit Option Form

:	P\$4
	Distr Door Sign
Q	
Ų	COCHOSCENI

The University of Sheffield

The Archives of Cultural Tradition

TAPE ARCHIV	/E DEPOSITOR'S OPTION	11	Form TA/2
DEPOSITOR	DATE	TAPE ACC. NO	
INFORMANT	LOCALITY R	ECORDED	

PART I

As depositor of this tape recording in The Archives of Cultural Tradition (hereinafter known as the Archives), Tape Archive, upon the observation of the conditions outlined in Part III of the Option by the Archives, I agree to :-

- A. Make this original tape recording, Tape Archive copies or transcriptions thereof, at the discretion of the Director of the Archives, available for transcribing and indexing for the Archives and listening to within the Archives building for research purposes, by the following designated groups of individuals, upon their signing an agreement for access to tape recordings/transcriptions held in the Tape Archive (Form TA/5):
 - 1. Members of the designated Survey Team of the Archives of Cultural Tradition (hereinafter known as the Survey Team) and registered students of the University of Sheffield conducting research.
 - Individuals or institutions conducting scholarly research, external to the Archives and not outlined in Al, above.
- B. However, as depositor, I retain the right to restrict the making available of Tape Archive copies and transcriptions of the original recording for loan by the Archives for limited periods for research purposes. Such restrictions apply to loans to the groups of individuals listed below, until such individuals have obtained written permission from me, through the Director of the Archives and upon their signing the agreement to borrow tape recordings/transcriptions from the Tape Archive (Form TA/6):
 - 1. Members of the Survey Team for use in illustrating exhibitions, lectures and seminars.
 - 2. Members of the Survey Team and registered students of the University of Sheffield conducting research.
 - 3. Individuals or institutions conducting scholarly research, external to the Archives and not outlined in Bl and 2, above.

PART II

As depositor, I retain all rights in the event of this original tape recording, Tape Archive copies or transcriptions thereof, being used in any commercial or other performance, recording, broadcast, telecast or film.

In addition, I retain all rights to publication of analyses, descriptive reports an verbatim transcriptions of sound, speech or music in this original tape recording.

However, I agree that the Archives may make this tape recording, Tape Archive copie or transcriptions thereof, available for scholarly research, within the terms of the above Option (Part I) and at the discretion of the Director of the Archives, provided the individual(s) or institution(s) outlined in the above Option, agree not to use this material, without my consent, in any published form or for commercial or other performance, recording, broadcast, telecast or film.

Individuals or institutions wishing to make use of this original recording, Tape Archive copies or transcriptions thereof, for purposes other than scholarly research or uses as outlined in the above Option, must secure prior written premission from myself, through the Director of the Archives, and append to any resulting publication acknowledgements as designated by myself and the Director of the Archives.

In the event of my death, this Option shall be converted to Option I, and all rights held by me as depositor, outlined in the above Option, Parts I and II, shall, unless I specify otherwise, be transferred to the Director of the Archives of Cultural Tradition, to be administered by the Archives.

and the second s	
DEPOSITOR	WITNESS
SIGNED	SIGNED
ADDRESS	ADDRESS

PART III

As Director of the Archives of Cultural Tradition, I acknowledge receipt of this tape recording for our Tape Archive, and undertake to observe the conditions outlined in the above Option, and to ensure that this tape recording, Tape Archive copies or transcriptions thereof, are used only for scholarly research and the uses outlined in the above Options.

Under no circumstances will this tape recording, Tape Archive copies or transcriptions thereof, be made available to any individual(s), institution(s) or organisation(s) for commercial use without prior written permission from the depositor or his trustee.

SIGNED		DATE	
Dinastan	Amelian of Cultural Tandikia		

Special conditions/instructions - if any :-